UNCLA	SSIFIED								_	NL		
	1 oF 2 ADA 065175				11 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	The second secon			Tanana an a	A serie of a serie of the series of the ser	Contraction - Contra	 Antonio de la construcción de la const
	A the definition of the defini			 The second second			A second					
				A series of the								
							101 102 102 102 102 102 102 102 102 102	A DATE OF THE PARTY OF THE PART				
		A set of a first set of a set	A state of the sta		-h- = = =							
									 Marine and Arriver and Arrive	 μ Stating τ Andread Stating		A - A - A - A - A - A - A - A - A - A -
		- [A state of the sta				N.					



Parametrically Excited Nonlinear Multi-Degree-Of-Freedom Systems

> MAJ Edward G. Tezak HQDA, MILPERCEN (DAPC-OPP-E) 200 Stovall Street Alexandria, VA 22332

Final Report March 79

Approved for public release; distribution unlimited.



A thesis submitted to Virginia Polytechnic Institute & State University, Blacksburg, VA in partial fulfillment of the requirements for the degree of Doctor of Philosophy.

79 02 15 029

FILE COPY AD AD AD 65175

39

\$

REPORT DOCUMENTATI	ON PAGE	READ INSTRUCTIONS
REPORT NUMBER	2. GOVT ACCESSION NO.	3. RECIPIENT'S CATALOG NUMBER
(lo)	ue, to used to few	evices, a neutritiación cachos
MITLE (and Subtitie)		5. TYPE OF REPORT & PERIOD COVERED
Parametrically Excited Nonlinea	r Multi-Degree-of-	Final Report March 79
Freedom Systems		6. PERFORMING ORG. REPORT NUMBER
PROFESSION OF THE PROPERTY OF		diname . mond and emotions
AUTHOR(.)	1	8. CONTRACT OR GRANT NUMBER(.)
Edward G. Tezak	Hd GITOZAK	9 Final reptis
PERFORMING ORGANIZATION NAME AND ADD	RESS	10. PROGRAM ELEMENT, PROJECT, TASK
Student, HQDA, MILPERCEN (DAPC-	OPP-E).	AREA & WORK ONTE NUMBERS
200 Stovall Street, Alexandria,	VA 22332 391 /1	Petrometrike excitation for etali ter the licel time.
1. CONTROLLING OFFICE NAME AND ADDRESS	11-	12. REPORT DATE
HQDA, MILPERCEN, ATTN: DAPC-OP	P-E, UL	13. NUMBER OF PAGES
200 Stovall Street, Alexandria,	VA 22332	164 (w/abstract 166)
4. MONITORING AGENCY NAME & ADDRESS(If di	tierent from Controlling Office)	15. SECURITY CLASS. (of this report)
(1) 41.2 - 7		Unclassified
(2) 100 pi		15a. DECLASSIFICATION/DOWNGRADING SCHEDULE
7. DISTRIBUTION STATEMENT (of the abetract on	tered in Block 20, 11 different fra	m Report)
7. DISTRIBUTION STATEMENT (of the ebetract on	tered in Block 20, if different fro	m Report)
7. DISTRIBUTION STATEMENT (of the ebetrect on 18. SUPPLEMENTARY NOTES	stered in Block 20, 11 different fro	m Report)
7. DISTRIBUTION STATEMENT (of the ebetrect on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU	stered in Block 20, if different fra	m Report)
7. DISTRIBUTION STATEMENT (of the abetract on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU	itered in Block 20, if different fro	m Report)
7. DISTRIBUTION STATEMENT (of the abetract on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU	stered in Block 20, 11 different fro	m Report)
7. DISTRIBUTION STATEMENT (of the ebetract on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse elde if necese	ntered in Block 20, if different fro	na Report)
7. DISTRIBUTION STATEMENT (of the ebetrect on 18. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse elde if necese Parametric excitation. multi-de	ery and identify by block number, gree-of-freedom syst	m Report)
7. DISTRIBUTION STATEMENT (of the abetract on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse aids if necess Parametric excitation, multi-de; flutter, method of multiple sca	ary and identify by block number, gree-of-freedom syst les, vibrations of b	m Report) cems, nonlinear vibrations, beams and simply supported
7. DISTRIBUTION STATEMENT (of the ebetrect on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse elde if necess Parametric excitation, multi-de flutter, method of multiple sca plates, internal resonance, mod	ery and identify by block number, gree-of-freedom syst les, vibrations of b al damping, jump pho	Tems, nonlinear vibrations, beams and simply supported enomena, limit cycles and
7. DISTRIBUTION STATEMENT (of the ebetrect en 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse elde 11 necess Parametric excitation, multi-de; flutter, method of multiple scal plates, internal resonance, mod stability.	ery and identify by block number, gree-of-freedom syst les, vibrations of t cal damping, jump pho	m Report) cems, nonlinear vibrations, peams and simply supported enomena, limit cycles and
7. DISTRIBUTION STATEMENT (of the ebetrect on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse elde if necess Parametric excitation, multi-de flutter, method of multiple scal plates, internal resonance, mod stability.	ary and identify by block number, gree-of-freedom syst les, vibrations of t al damping, jump pho	m Report) cems, nonlinear vibrations, beams and simply supported enomena, limit cycles and
 DISTRIBUTION STATEMENT (of the abetract on SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU KEY WORDS (Continue on reverse alde if necesse Parametric excitation, multi-de; Plutter, method of multiple scal plates, internal resonance, mod stability. STRACT (Continue on reverse alde if necesse An analysis of parametrical 	ary and identify by block number, gree-of-freedom syst les, vibrations of t al damping, jump pho sy and identify by block number) lly excited nonlines	r multi-degree-of-freedom
 7. DISTRIBUTION STATEMENT (of the obstract on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse olde if necesse Parametric excitation, multi-de flutter, method of multiple scalplates, internal resonance, mod stability. 9. METRACT (Continue on reverse oth H necessary An analysis of parametrical systems is presented. The non 	ary and identify by block number, gree-of-freedom syst les, vibrations of t al damping, jump pho ary and identify by block number) ly excited nonlines linearity considered	Tems, nonlinear vibrations, beams and simply supported enomena, limit cycles and ar multi-degree-of-freedom l is cubic and small so that
7. DISTRIBUTION STATEMENT (of the obstract on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse elde II necess Parametric excitation, multi-de flutter, method of multiple scal plates, internal resonance, mod stability. An analysis of parametrical systems is presented. The non the system of equations is weal	ary and identify by block number, gree-of-freedom syst les, vibrations of t al damping, jump pho inv and identify by block number) ly excited nonlines linearity considered kly nonlinear. Mode	The Report) The Report) The Report of the American State of the Am
 7. DISTRIBUTION STATEMENT (of the observed on a state of the observed of the observed	ary and identify by block number, gree-of-freedom syst les, vibrations of t al damping, jump pho by and identify by block number) ly excited nonlinea linearity considered kly nonlinear. Moda armonic. The system	m Report) tems, nonlinear vibrations, beams and simply supported enomena, limit cycles and ar multi-degree-of-freedom l is cubic and small so that al damping is included and as examined include those wit
 7. DISTRIBUTION STATEMENT (of the abstract on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse side if necess Parametric excitation, multi-deg flutter, method of multiple scal plates, internal resonance, mod stability. 9. METRACT (Continue on reverse side H mecess An analysis of parametrical systems is presented. The non the system of equations is weal the parametric excitation is had distinct natural frequencies a The significant role played by 	ary and identify by block number, gree-of-freedom syst es, vibrations of b al damping, jump pho sy and identify by block number) ly excited nonlines linearity considered kly nonlinear. Moda armonic. The system s well as those with the existence of a	The Report) The Report of the second state of the second
 7. DISTRIBUTION STATEMENT (of the obstract on S. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse elde II necess Parametric excitation, multi-de flutter, method of multiple scal plates, internal resonance, mod stability. 9. METRACT (Continue on reverse elde II necess An analysis of parametrical systems is presented. The non the system of equations is weal the parametric excitation is had distinct natural frequencies a The significant role played by explored in depth. The deriva 	ary and identify by block number, gree-of-freedom syst les, vibrations of t al damping, jump pho ary and identify by block number) ly excited nonlines linearity considered kly nonlinear. Moda armonic. The system s well as those with the existence of an tive-expansion versi	Tems, nonlinear vibrations, beams and simply supported enomena, limit cycles and ar multi-degree-of-freedom l is cubic and small so that a damping is included and as examined include those wit a single repeated frequency internal resonance is on of the method of multiple
7. DISTRIBUTION STATEMENT (of the obstract on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse elde II necess Parametric excitation, multi-de; flutter, method of multiple scal plates, internal resonance, mod stability. 9. METRACT (Continue on reverse etch H necess An analysis of parametrical systems is presented. The non the system of equations is weal the parametric excitation is had distinct natural frequencies a The significant role played by explored in depth. The derival	ary and identify by block number, gree-of-freedom syst les, vibrations of t al damping, jump pho inearity considered kly nonlinear. Moda armonic. The system s well as those with the existence of an tive-expansion versi	The Report) The Report) The Report of the
7. DISTRIBUTION STATEMENT (of the ebetrect on 8. SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU 9. KEY WORDS (Continue on reverse elde if necess Parametric excitation, multi-de; Flutter, method of multiple scal plates, internal resonance, mod stability. 9. MINTRACT (Continue on reverse elde H mecess An analysis of parametrical systems is presented. The non the system of equations is weal the parametric excitation is had distinct natural frequencies a The significant role played by explored in depth. The deriva	ary and identify by block number, gree-of-freedom syst les, vibrations of t al damping, jump pho al damping, jump pho by and identify by block number) ly excited nonlinea linearity considered kly nonlinear. Moda armonic. The system s well as those with the existence of an tive-expansion version 79	The report) The report of the repeated frequency the repated frequency the repeated frequency the
 DISTRIBUTION STATEMENT (of the obstract on a supplementary notes SUPPLEMENTARY NOTES Ph.D. Dissertation VPI&SU KEY WORDS (Continue on reverse olde 11 necesses Parametric excitation, multi-de, flutter, method of multiple scales, internal resonance, mod stability. Matter (Continue on reverse other H mecesses An analysis of parametrical systems is presented. The non the system of equations is weal the parametric excitation is had istinct natural frequencies a The significant role played by explored in depth. The derivation of 1 Mov 65 is a solution of 1 Mov 65 is a solution. 	ary and identify by block number, gree-of-freedom syst les, vibrations of b al damping, jump pho ary and identify by block number) ly excited nonlines linearity considered kly nonlinear. Moda armonic. The system s well as those with the existence of an tive-expansion version sector of the system the existence of an tive-expansion version tive-expansion version	The Report) The Report of the

A

UNCLASSIFIED

SECURITY CLASSIFICATION OF THIS PAGE(When Data Entered)

Block 20 cont'd.

scales, a perturbation technique, is used to develop solvability conditions for the various combinations of internal and parametric resonances considered.

Regions where trivial and nontrivial solutions exist are defined and the stability of the solutions within each region is discussed. Non trivial, unstable solutions have been shown to exist in regions where nontrivial stable solutions are known. Numerical solutions do not hint at the existence of these solutions.

The role of internal resonance in parametrically excited systems is explored Strong modal interaction is demonstrated as a consequence of the presence of the cubic nonlinearity and the internal resonance. Because of this modal coupling, modes other than the one excited can dominate the response. A multiplicity of jumps is shown to exist. Parametric excitation in the nonlinear flutter problem has been examined in

detail for the first time. The effect of the parametric excitation can raise the flutter speed if the amplitude is small and lower it if the amplitude is large.

Limit-cycle behavior in the flutter problem is developed in a unique analytic way. The condition which predicts the onset of flutter is developed by using a linear analysis. The solution grows without bound. Interestingly, the same condition in the nonlinear analysis predicts the existence of a real nontrivial solution with a finite amplitude, the so-called limit-cycle.

ENDY AS MALL

USANAV northtressell . G. d

UNCLASSIFIED SECURITY CLASSIFICATION OF THIS PAGE(When Date Entered)

PARAMETRICALLY EXCITED NONLINEAR MULTIDEGREE-OF-FREEDOM SYSTEMS

and the state Lun

Edward G. Tezak

Dissertation submitted to the Graduate Faculty of the Virginia Polytechnic Institute and State University

..., in partial fulfillment of the requirements for the degree of a con b

DOCTOR OF PHILOSOPHY

in

Engineering Mechanics

APPROVED:

Nean Y. Moo

D. T. Mook, Chairman

A. H. Nayself A. H. Nayself Raymond Hellent

8174	White Section
818	Bett Section
ES MONTAS	0
IS OF CATIO.	
67	8 /59 PT
67 61318138710 8144	R/AVAILANLITY BORES
87 81318188710 0115	R/AVANADUNTY BORFS AVAIL BATTON SPECIAL
61018184710 0164	RZAVALLALILITY BORER AVAIL, LUIZER BALDIAL

March, 1979

Blacksburg, Virginia

MAR 2 1979 D

Frederick

R. A. Heller

D. Frederic

to:

ACKNOWLEDGEMENTS

I would like to express my gratitude to:

Dr. D. T. Mook for his continuing support, guidance, time and patience in helping me see this research effort through to completion under less than optimum conditions;

Dr. A. H. Nayfeh for his willingness to share his wisdom, expertise and advice throughout my research effort;

Dr. R. Heller for the interest he has shown in my graduate work;

Dr. R. H. Plaut for accepting on eleventh-hour request to serve on my committee and for his helpful suggestions generated by his careful reading of my thesis;

Dr. D. Frederick for his support and encouragement throughout my graduate work;

BG. F. A. Smith, Jr. for having faith in me and for affording me the opportunity to further my education and serve the U.S. Military Academy;

Mrs. J. Huebner, Mrs. J. Cooke, Ms. J. Bryant, and Mrs. P. Belcher for the tremendous job they did in typing the various drafts and final copies;

Mr. and Mrs. Roy E. Hylton, Jr. for their friendship, encouragement, hospitality and support;

My wife, Marty, and children, Chris and Scott, for the countless hours and irreplaceable time I spent away from them and the silent support and sacrifice they endured in my behalf.

TABLE OF CONTENTS

	PAGE				
ACKNOWLEDGMENTS					
TABLE OF CONTENTS					
LIST OF SYMBOLS					
1. INTRODUCTION	. 1				
1.1 Literature Review	. 2				
1.2 Assessment of Previous Work	. 5				
1.3 Contributions of the Current Work	. 7				
PART I					
2. PROBLEM FORMULATION AND METHOD OF SOLUTION	. 10				
2.1 Problem Formulation	. 10				
2.2 Method of Solution	. 11				
2.3 Summary	. 13				
3. PARAMETRIC RESONANCE IN THE ABSENCE OF					
INTERNAL RESONANCE	. 15				
3.1 No Internal Resonance	. 15				
3.2 The Case of Modes Not Involved in a					
Parametric Resonance	. 15				
3.3 The Case Where $\lambda \stackrel{\sim}{\sim} 2\omega_k$, n \neq k	. 16				
3.4 The Case Where $\lambda \stackrel{\sim}{\sim} \omega_m + \omega_k$, $m \neq k$. 20				
3.5 The Case Where $\lambda \stackrel{\sim}{\sim} \omega_k - \omega_m$, $m \neq k$. 23				
3.6 Summary	. 25				
4. THE EFFECT OF INTERNAL RESONANCE	. 26				
4.1 The Case Where $\lambda \stackrel{\sim}{\sim} 2\omega_{\rm m}$ and $\omega_{\rm k} \stackrel{\sim}{\sim} 3\omega_{\rm m}$. 28				
4.2 The Case Where $\lambda \stackrel{\sim}{\sim} 2\omega_k$ and $\omega_k \stackrel{\sim}{\sim} 3\omega_m$. 30				

TABLE OF CONTENTS - CONTINUED

		PAGE
	4.3 The Case Where $\lambda \stackrel{\sim}{\sim} 4\omega_k \stackrel{\sim}{\sim} \omega_m + \omega_k$ and $\omega_k \stackrel{\sim}{\sim} 3\omega_m$.	33
	4.4 Summary	35
	PART II	
5.	THE ANALYSIS OF SYSTEMS HAVING REPEATED FREQUENCIES IN	THE
	ABSENCE OF INTERNAL AND PARAMETRIC RESONANCES	36
	5.1 The Case of Modes Having Repeated Frequencies	41
	A. The Linear Response - The Flutter Condition	41
	B. The Nonlinear Response - Limit-Cycle	
	Behavior	43
	5.2 Summary	46
5.	PARAMETRIC RESONANCE IN THE ABSENCE OF	
	INTERNAL RESONANCE	49
	6.1 The Case Where $\lambda \stackrel{\sim}{\sim} 2\omega_j$	49
	6.2 The Case Where $\lambda \stackrel{\sim}{\sim} \omega_r + \omega_j$	52
	6.3 The Case Where $\lambda \stackrel{\sim}{\sim} \omega_r - \omega_m$	59
	6.4 Summary	60
7.	THE EFFECT OF INTERNAL RESONANCE	61
	7.1 The Case Where $\lambda \stackrel{\sim}{\sim} 2\omega_j$ and $\omega_r \stackrel{\sim}{\sim} 3\omega_j$	61
	7.2 The Case Where $\lambda \stackrel{\sim}{\sim} \omega_r + \omega_j$ and $\omega_r \stackrel{\sim}{\sim} 3\omega_j$	64
	7.3 The Case of No Parametric Resonance with	
	$\omega_r \approx 3\omega_j$	65
	7.4 Summary	67

.

TABLE OF CONTENTS - CONTINUED

PAGE

PART III

8.	LATE	RAL VIBRATIONS OF COLUMNS
	8.1	General 68
	8.2	Hinged-Hinged Beam 70
	8.3	Hinged-Clamped Beam 79
	8.4	Observations - Internal and Parametric
		Resonances
	8.5	Summary 102
9.	PANE	L FLUTTER
	9.1	General 104
	9.2	Simply Supported Plate 110
	9.3	Parametric Resonance 112
	9.4	Internal Resonance $\omega_3 \stackrel{\sim}{\sim} 3\omega$ 123
	9.5	Summary 133
REFE	RENCE	S 136
APPE	NDIX	A 148
VITA		

V

LIST OF SYMBOLS

A = cross-section, also coefficient matrix, equation (9.9).

 $A_n = \text{complex functions of } T_1 \text{ used in the perturbation scheme, equation}$ (2.6).

a = panel length.

- a_n = twice the amplitude of A_n , equation (3.1).
- Δa_n = perturbation of a_{ns} used to study the stability of the steadystate motion, equation (3.8).

B = coefficient matrix, equation (9.9).

C = damping coefficient.

 $C_n = modal damping coefficient.$

D = flexural rigidity.

 D_n = partial differentiation with respect to T_n , equation (2.3d).

E = elastic modulus.

E_n = terms in the solvability condition due to the parametric resonance, equation (2.9).

F = amplitude of the parametric excitation, and coefficient matrix of parametric terms, equation (9.9) - (9.10).

 f_{nm} = cross-modal amplitude of the excitation, equation (2.1).

 G_{nm} = constants used to compute Γ_{nmpq} .

h = panel thickness.

I = moment of inertia of the cross-section area, identity matrix.

In = terms in the solvability condition due to the internal resonance, equation (2.9).

Imag = imaginary part of a complex quantity.

LIST OF SYMBOLS - Continued

	J	= matrix of Jordan Form, equation (9.11).
	ĸ'n	= terms in the solvability condition associated with the
		repeated frequency and the aerodynamic pressure, equation
		(5.36d).
	L	= characteristic length used in the nondimensionalization
		scheme, zL = l.
	٤	= dimensional length of the beam.
	м	= Mach number.
	P(t)	= parametric excitation.
	Q,	= coefficients in the solvability conditions for the clamped-
		hinged beam, equations (8.19) - (8.23).
•	r	= radius of gyration of the cross-section area.
	R	= real part of a complex quantity.
	T ₀ ,T ₁ ,T ₂	= fast and slow scales used in the perturbation scheme,
		equation (2.3a).
	Un(t;e)	= time-dependent coefficients appearing in the expansion of
		w, equation (2.1) - (2.2).
	U _n , U _n ,	= functions of T_0 and T_1 appearing in the expansion of U_n ,
		equation (2.2).
	U_	= free stream velocity.
	v	<pre>= time-dependent coefficients in similarity transformation,</pre>
		equation (9.13).
	w	= the deflection.
	z	= matrix used for similarity transformation, equation (9.13).

vii

SYMBOLS BASED ON THE GREEK ALPHABET

- α_n = phase of A_n , equation (3.1).
- α_{nm} = elements of aerodynamic pressure matrix, equation (9.8b).
- $\beta_{\beta_i} = \beta_{\beta_i} = \beta_{\beta_i}$ = phases introduced to put the solvability conditions in autonomous form, equations (4.18) - (4.21) and (8.27) - (8.35).
- β_{nm} = coefficients of dynamic pressure terms, equation (5.1).
- Γ_{nmpq} = coefficients appearing in the governing equations for U_n , equation (2.1).

 Δ_{kn} = Kronecker delta.

 δ_n = scaling exponents of ϵ used in nondistinct frequency analysis, equations (5.1) - (5.4).

$$\varepsilon$$
 = small parameter used in the perturbation scheme, the square of the ratio of r to L.

- Λ = aerodynamic detuning parameter, used as a measure of dynamic pressure equation (5.1).
- λ = frequency of the parametric excitation, equation (2.1).
- λ = stability parameter, used as a measure of dynamic pressure, equation (9.3).
- λ_c = critical value of stability at onset of flutter.
- µ = convenient phase angle introduced to put the solvability conditions in autonomous form, equation (3.6c).
- μ_n = phases introduced to put the solvability conditions in autonomous form, equations (8.19) - (8.23).

viii

SYMBOLS BASED ON THE GREEK ALPHABET - Continued

- μ_s = steady-state value of μ , equation (8.13).
- $\Delta \mu$ = perturbation of μ used to study the stability of the steady-state motion, equation (3.8d).
- π = 3.141...
- ρ = detuning parameters for the external (parametric) resonances, equation (3.5a).

 $\boldsymbol{\rho}_{m},\boldsymbol{\rho}_{\infty}\text{=}$ panel and air density.

 ϕ_m = linear free-oscillation modes used to describe w.

 ψ_n = normalizing factor for the free-oscillation modes.

 Ω = eigenvalues in the stability analysis.

 ω_m = natural frequencies (i.e., the eigenvalues corresponding to $\phi_m).$

CHAPTER ONE

INTRODUCTION

Parametrically excited multi-degree-of-freedom systems have received considerable attention. Linear models of such systems predict regions of instability in the parameter space (in this case the parameters are essentially the frequency and the amplitude of the excitation) where the amplitude of the response grows without bound. The inclusion of damping in the model does not change the basic character of the results.

Of course, amplitudes of real systems do not grow without bound; the amplitudes are limited by nonlinear effects. Moreover, nonlinear effects make it possible for a bounded resonant motion to exist in regions where the linear models predict such motions are impossible. The interaction of the phase and amplitude, which is a characteristic of nonlinear systems, is responsible for this.

The system of governing equations is a set of second-order nonlinear ordinary differential equations having variable coefficients. The nonlinearity considered in this work is cubic in nature and considered small. Thus the equations may be referred to as weakly nonlinear. The excitation is considered to be harmonic. With some types of parametric resonance several modes can be strongly excited by a single harmonic frequency. Internal resonances can be responsible for strong modal coupling and, as a consequence, for a significant transfer of energy from one mode to another.

In what follows a general method of analyzing parametrically excited nonlinear multi-degree-of-freedom systems will be described in detail and applied to several physical examples. The parametric excitation is assumed to be harmonic with constant amplitude and frequency. Both internal and parametric resonances will be considered. The resonances to be examined include:

a) parametric

 $\lambda \stackrel{\sim}{\sim} \sum n_i \omega_i$

b) internal

 $\omega_n = \sum a_i \omega_i$

where the n, n_i, and a_i are integers, λ represents the frequency associated with the excitation, and the ω_i , i = 1, 2, 3..., represent the natural frequencies of the system. Modal damping is also included.

The response of the system is examined for the case where the natural frequencies, ω_i , are distinct as well as for the case where there is one repeated frequency, i.e. the so-called flutter case. Regions describing the existence of trivial and nontrivial solutions are discussed along with the stability of these solutions within these regions. Before proceeding further a survey of the literature concerning parametrically excited nonlinear systems is in order.

1.1 Literature Review

A survey of the literature reveals that there is an abundance of material written on parametric, nonlinear and resonance phenomenon. A totally comprehensive review would be prohibitive here both in time and space. The most complete effort in this direction has been accomplished by Nayfeh and Mook (1979).

Faraday (1831) appears to be the first to have observed the phenomenon of parametric resonance in conjunction with surface waves in fluid-filled cylinders. He was followed by Melde (1859), Strutt (1887), Stephenson (1906) and Raman (1912) who worked with vibrating strings. Stephenson (1908) showed that periodic loading of columns can have a stabilizing effect while Beliaev (1924) showed that a lateral motion can occur even though the axial loading of a column is below the static buckling load. Other early works on columns include Andronov and Leontovich (1927), Krylov and Bogoliubov (1935) and Chelomei (1939). Numerous books on parametric excitations include Bondarenko (1936), McLachlan (1947,1950), Den Hartog (1947), Minorsky (1947,1962), Stoker (1950), Hayashi, (1953a, 1964), Coddington and Levinson (1955), Malkin (1956), Cunningham (1958), Kauderer (1958), Bogoliubov and Mitropolsky (1961), Bolotin (1964), Andronov, Vitt, and Khaikin (1966), Meirovitch (1970), Cesari (1971), Nayfeh (1973) and Evan-Iwanowski (1976).

Early studies of nonlinear vibrations of bars were conducted by Woinowsky-Krieger (1950) and Burgreen (1951) and involved the use of elliptic functions in conjunction with an assumed single mode spatial function. Others who investigated the free oscillations of beams with hinged ends using various techniques include Wagner (1965), Srinivasan (1965,1967), Woodall (1966), Evensen (1968), Rehfield (1973, 1975) and Lou and Sikarskie (1925). Multiple theoretical and experimental investigations were conducted by Eisley (1964b), Morris (1965), Srinivasan

(1966a), Bennett and Rinkel (1972), Bert and Fisher (1972) and Rehfield (1974a), while multi-mode forced responses were investigated by Eisley and Bennett (1970), Bennett and Eisley (1970) and Busby and Weingarten (1972). Straight and buckled beams were investigated by Tseng and Dugundji (1970, 1971) using harmonic balance. Others who analyzed buckled beams include Eisley (1964a), and Min and Eisley (1972).

Atluri (1973) examined beams with large curvatures and Nayfeh (1973a) studied a beam with slowly varying properties along its length. Both used the method of multiple scales. Nayfeh, Mook and Lobitz (1974) and Verma and Murthy (1974) analyzed nonuniform beams while Raju, Venkateswara and Kanaka Raju (1976) studied tapered beams.

A comprehensive treatment of the vibrations of linear flat plates is given in Leissa (1969). Nonlinear vibrations of plates has received extensive treatment. Yamaki (1961), Smith, Malme and Gogos (1961), Eisley (1964b), Murthy and Sherbourne (1972), Bayles, Lowery and Boyd (1973), Vendham (1975a,b), Crawford and Atluri (1975), Venkateswara, Raju and Kanaka Raju (1976a) all analyzed nonlinear vibrations of rectangular plates. Anisotropic rectangular plates were treated by Sathyamoorthy and Pandalai (1970), Bennett (1971) Chandra and Basava Raju (1975a,b) and Chandra (1976). Plates of a variety of shapes were investigated by numerous authors: circular by Yamaki (1961), Nowinski (1962), Bulkeley (1963) Huang and Sandman (1971), Huang (1972a,b, 1973, 1974) and Sridhar, Mook and Nayfeh (1975, 1978); annular by Sandman and Huang (1971), Huang and Woo (1973) and Huang, Woo and Walker (1976); tringular by Vendhan and Dhoopar (1973), Vendhan and Das (1975) and Vendhan, (1975b); and elliptic by Lobitz, Nayfeh and Mook (1977).

The phenomenon of flutter in plates and shells has received wide attention. The book by Dowell (1975) discusses the problem very succinctly. Other work in this area includes the study of flat plates by Kobayashi (1962a,b), Bolotin (1963), Dugundji (1966), Dowell (1966, 1967a, 1973), Morino (1969), Vetres and Dowell (1970), Eastep and McIntosh (1971), Kuo, Morino and Dugundji (1972) and Smith and Morino (1976). Numerous authors have studied curved plates and shells. Interaction of panel flutter with parametric excitation was analyzed by Dowell (1970a), Dzygadlo (1970) and Kuo, Morino and Dugundji (1973).

1.2 Assessment of Previous Work

As demonstrated in the last section there is a tremendous volume of material which has been written on free and forced vibration; parametric excitations; linear and nonlinear systems; resonance; and single, twodegree and multi-degree-of-freedom systems. An attempt to comment on the value and limitations of each of these works and how each relates to the current discussion would be of marginal utility. Instead, the discussion will be limited to these few recent works which are the most relavent and which are significant contributions in themselves. Most authors cited here have discussed the various aspects of the analysis described in the introduction. However, no one has attempted to unify the analysis and make it comprehensive until now.

For instance, Tseng and Dugundji (1970, 1971) discussed straight and buckled beams using harmonic balance, but their analysis did not involve modal coupling. Atluri (1973) studied nonlinear vibrations of hinged beams for large curvatures but did not consider axial excitation.

Nayfeh (1973a) considered a beam with slowly varying properties along its length but did not consider modal coupling. Yamomoto and Saito (1970) analyzed parametrically excited multi-degree-of-freedom systems using the method of averaging. However, the system considered was linear and did not include repeated frequencies. Hsu (1963, 1965) also studied parametric excitation in multi-degree-of-freedom systems and developed stability criteria. He used the method of averaging and considered distinct frequencies. Tso and Asmis (1974) studied multiple parametric resonances in nonlinear systems. They discussed parametric resonances and combination resonances but confined their analysis to two-degreeof-freedom systems. In addition, they did not include an all important cubic term in the analysis. Fu and Nemat-Nasser (1972,1975) studied the stability of multi-degree-of-freedom systems and included the effect of repeated frequencies. However, their analysis was confined to linear systems. Another excellent work on parametrically excited linear systems having many degrees of freedom was conducted by Nayfeh and Mook (1977). Additional studies were conducted by Sugiyama, Fujiwara, and Sekiya (1968) using analog simulation, and they found combination resonances. Iwatsubo, Saigo and Sugiyama (1973) and Iwatsubo, Sugiyama and Ogino (1974) performed theoretical and experimental investigations on parametrically excited columns and also demonstrated the existence of combination resonances. They included the effects of internal and external damping in their analysis but not internal resonance. The effects of internal resonance have been virtually ignored in most works until recently. The importance of internal resonance in the analysis of

ship motions has been demonstrated by Nayfeh, Mook and Marshall (1973, 1974) and Mook, Marshall and Nayfeh (1974). Its importance in the analysis of structural vibrations has been shown by Nayfeh, Mook and Sridhar (1973), Nayfeh, Mook and Lobitz (1974), Sridhar, Nayfeh and Mook (1975) and Mook, Sridhar and Nayfeh (1978). These latter works have been a major advancement toward unifying nonlinear, resonance studies. However, they do not include parametric excitations.

The majority of work discussed so far deals with distinct frequencies. When the natural frequencies are no longer distinct the response enters the so-called flutter mode or condition. The book by Dowell (1975) provides an insight into the mechanism that can cause flutter. Morino (1969) and Smith and Morino (1976) have presented an excellent discussion demonstrating limit cycles and stability. However, neither included the effect of an in-plane excitation. The role of a parametric excitation has thus been neglected. The numerical works of Dowell (1966, 1970) suffer from the same deficiency as those of Kuo, Morino and Dugundji (1972). Constant in-plane loads were used in these analyses. A harmonic excitation and its influence on flutter were considered by Dowell (1974), Dzygadlo (1970) and Kuo, Morino and Dugundji (1973). However, the excitation was not parametric in nature.

1.3 Contributions of the Current Work

Though a vast amount of work has been cited here, there are some significant facets of analysis of parametrically excited multi-degreeof-freedom systems that have not been examined. The role that internal resonance plays has been virtually ignored. Parametric excitations

in the nonlinear flutter problem has been similarly neglected. In addition, there has been no single method of analysis that is general in nature and applicable to a wide variety of problems. This work investigates these fascinating aspects of multi-degree-of-freedom systems and thus fills a longstanding void. In the process a unified versatile approach is developed and some interesting behavior is discussed. The following contributions are unique and significant:

a. A general method for analyzing parametrically excited nonlinear multi-degree-of-freedom systems is developed. To demonstrate its versatility, we apply the method to the following physical systems:

- 1) Systems with distinct natural frequencies
 - a) . without internal resonance
 - b) with internal resonance
- 2) Systems with repeated natural frequencies
 - a) without internal resonance
 - b) with internal resonance

b. Regions where trivial and nontrivial solutions exist are defined and the stability of the solutions within each region is discussed. Nontrivial, unstable solutions have been shown to exist in regions where nontrivial stable solutions are known. Numerical solutions do not hint at the existence of these solutions.

c. The role of internal resonance in parametrically excited, nonlinear systems is explored. Strong modal interaction is demonstrated as a consequence of the presence of the cubic nonlinearity and the internal resonance. Because of this modal coupling, modes other than

the one excited can dominate the response. A multiplicity of jumps is shown to exist.

d. Parametric excitation in the nonlinear flutter problem has been examined in detail for the first time, and includes the effects of internal resonance.

e. Limit-cycle behavior in the flutter problem is developed in a unique analytic way. The condition which predicts the onset of flutter is developed by using a linear analysis. The solution grows without bound. Interestingly, the same condition in the nonlinear analysis predicts the existence of a real nontrivial solution with a finite amplitude, the so-called limit cycle.

f. The current work is divided into several parts. In Part I, the analysis deals with distinct natural frequencies. Chapter 2 is devoted to the problem formulation and generation of the solvability conditions. Chapter 3 deals with the analysis of parametric resonance in the absense of internal resonance. Chapter 4 is the heart of the analysis and includes the internal as well as the parametric resonance.

In Part II, the same general approach is followed for the analysis of non-distinct natural frequencies. Chapter 5 develops the limit-cycle behavior and the general solvability conditions for the situation involving repeated frequencies. Chapter 6 considers parametric resonance in the absence of internal resonance and Chapter 7 discusses internal and parametric resonance.

Part III is devoted to numerical examples. Chapter 8 deals with the case of distinct frequencies and uses the beam as an example. Chapter 9 develops the flutter problem for a simply supported plate.

CHAPTER TWO

PROBLEM FORMULATION AND METHOD OF SOLUTION

In this chapter the nonlinear system studied is defined. Modification of the basic system is made in a subsequent chapter when the case of non-distinct natural frequencies is discussed. Solvability conditions are developed by using the method of multiple scales.

2.1 Problem Formulation

The system to be analyzed is governed by the following set of ordinary-differential equations:

$$U_{n} + \omega_{n}^{2}U_{n} = \varepsilon \left[2\cos\lambda t \int_{m=1}^{\infty} f_{nm}U_{m} - 2C_{n}U_{n} + \int_{m=1}^{\infty} \sum_{p=1}^{\infty} \sum_{q=1}^{p} r_{nmpq}U_{m}U_{p}U_{q} \right]$$

where the ω_n are the distinct natural frequencies corresponding to the linear free-oscillation modes; ε is a small dimensionless parameter; the f_{nm} are the amplitudes associated with the harmonic parametric excitation; λ is the constant frequency of excitation; the C_n are the modal viscous damping coefficients; and the Γ_{nmpq} are constant coefficients.

Experience has shown that a straightforward perturbation expansion for U_n , for small ε , results in the emergence of secular terms, which make this expansion not uniformly valid for large t. Hence, in order to obtain a uniformly valid expansion, we employ the method of multiple scales. Specifically, the derivative-expansion version of the method of multiple scales is utilized to study the steady-state solutions and their stability.

2.2 Method of Solution

Following the method of multiple scales (Nayfeh, 1973), we assume expansions for the U_n in the form

$$U_{n}(t;\varepsilon) = U_{n^{0}}(T_{0},T_{1}) + \varepsilon U_{n^{1}}(T_{0},T_{1}) + \dots \qquad (2.2)$$

where

$$T_0 = t \text{ and } T_1 = \varepsilon t$$
 (2.3a)

The derivatives become

$$\frac{d}{dt} = D_0 + \varepsilon D_1 + O(\varepsilon^2)$$
(2.3b)

and

$$\frac{d^2}{dt^2} = D_0^2 + \epsilon 2 D_0 D_1 + O(\epsilon^2)$$
 (2.3c)

where

$$D_0 = \frac{\partial}{\partial T_0}$$
 and $D_1 = \frac{\partial}{\partial T_1}$ (2.3d)

Substituting equations (2.2) and (2.3) into equation (2.1) and equating the coefficients of ε^0 and ε , we obtain

$$D_0^2 U_{n^0} + \omega_n^2 U_{n^0} = 0 \tag{2.4}$$

$$D_{0}^{2}U_{n^{1}} + \omega_{n}^{2}U_{n^{1}} = -2D_{0}D_{1}U_{n^{0}} - 2\cos\lambda T_{0}\sum_{m=1}^{\infty}f_{nm}U_{m} - 2C_{n}D_{0}U_{n^{0}} + \sum_{m=1}^{\infty}\sum_{p=1}^{\infty}\sum_{q=1}^{\infty}\Gamma_{nmpq}U_{m^{0}}U_{p^{0}}U_{q^{0}} \text{ for } n = 1,2,... \quad (2.5)$$

We can write the solution of equations (2.4) in the form

$$U_{n^0} = A_n(T_1) \exp(i\omega_n T_0) + cc$$
 (2.6)

for n = 1, 2, ..., where cc stands for the complex conjugate of the preceding term and the A_n are, at this point, unspecified complex function of T_1 . They are determined by eliminating "troublesome" terms from the U_{n^1} .

Substituting equation (2.6) into equation (2.5) leads to

$$D_{0}^{2}U_{n^{1}} + \omega_{n}^{2}U_{n^{1}} = -2i\omega_{n}(D_{1}A_{n} + C_{n}A_{n})\exp(i\omega_{n}T_{0}) - \sum_{m=1}^{\infty} f_{nm}A_{m}^{4}\exp[i(\omega_{m} + \lambda)T_{0}] + \exp[i(\omega_{m} - \lambda)T_{0}] + \sum_{m=1}^{\infty} \sum_{p=1}^{\infty} \sum_{q=1}^{\infty} \Gamma_{nmpq}^{4}A_{m}A_{p}A_{q} \times \exp[i(\omega_{m} + \omega_{p} + \omega_{q})T_{0}] + A_{m}A_{p}\overline{A}_{q}\exp[i(\omega_{m} + \omega_{p} - \omega_{q})T_{0}] + A_{m}\overline{A}_{p}A_{q}\exp[i(\omega_{m} - \omega_{p} - \omega_{q})T_{0}] + cc \qquad (2.7)$$

where the overbars denote the complex conjugates.

The terms containing the factor $\exp(\pm i\omega_n T_0)$ in equation (2.7) lead to so-called secular terms in U_{n^1} , while terms containing factors such as $\exp(i\Omega T_0)$, where $\Omega - \omega_n = O(\varepsilon)$, lead to so-called small-divisor terms in U_{n^1} . The former renders the expansion nonuniform (U_{n^1}/U_{n^0}) is not bounded) as t increases, while the latter are inconsistent with the assumed expansion (2.2) because they raise εU_{n^1} to the same order as U_{n^0} . Consequently, both kinds of terms are "troublesome" and must be eliminated for a uniform expansion.

Secular terms, for example, are associated with the combination $\omega_m + \omega_p - \omega_q$

when m = n and p = q. In addition to all these possible combinations, there may be small-divisor terms associated with combinations having the form

$$\omega_{n} = a\omega_{m} + b\omega_{p} + c\omega_{q} + \varepsilon\sigma_{n} \qquad (2.8a)$$

where a, b, and c are integers such that |a| + |b| + |c| = 3, and σ_n is a detuning parameter. If equation (2.8a) is satisfied, an internal resonance is said to exist. Finally, if there are combinations having the form

$$\lambda = a\omega_m + b\omega_n + \varepsilon\rho \tag{2.8b}$$

where a and b are integers such that |a| + |b| = 2, and ρ is another detuning parameter, additional small-divisor terms arise. If equation (2.8b) is satisfied, a parametric resonance is said to exist.

It follows that the general form of the conditions that eliminate the troublesome terms from the U_{n^1} is

$$-2i\omega_{n}(D_{1}A_{n} + C_{n}A_{n}) + A_{n} \sum_{p=1}^{\infty} \gamma_{np}A_{p}\overline{A}_{p} + I_{n} + E_{n} = 0 \text{ for } n = 1,2,\dots$$
(2.9)

where

$$\gamma_{np} = \begin{bmatrix} 3\Gamma_{nnnn} & \text{if } p = n \\ \\ 2(\Gamma_{nnpp} + \Gamma_{npnp} + \Gamma_{nppn}) & \text{if } p \neq n \end{bmatrix}$$

The terms $I_n(A_1, A_2, ...)$ and $E_n(A_1, A_2, ...)$ are complex functions, resulting from the internal and parametric resonances, respectively, if any exist.

2.3 Summary

A general approach to the solution of equations (2.1) has been developed using the method of multiple scales to obtain a uniformly valid solution. The solvability conditions yielded equations (2.9), which are general and can be applied to a large variety of problems, depending on the form of the internal and parametric resonances in-volved.

CHAPTER THREE

PARAMETRIC RESONANCE IN THE ABSENCE OF INTERNAL RESONANCE

In this chapter parametric resonance is examined in detail. The effect of an internal resonance is discussed in the next chapter. Various cases are considered and some general observations are given. The emphasis is on the steady-state response and its stability.

3.1 No Internal Resonance

In this case $I_n \equiv 0$ for all modes and $E_n = 0$ for all modes that are not part of any resonance with the excitation. The E_n for the resonating modes depend on the resonant combinations of frequencies. In the first case considered, there is neither internal nor parametric resonance.

3.2 The Case of Modes Not Involved in a Parametric Resonance

For modes not associated with a parametric or internal resonance $I_n = E_n \equiv 0$ in (2.9). Now we let

$$A_{n} = \frac{1}{2} a_{n}(T_{1}) \exp[i\alpha_{n}(T_{1})]$$
 (3.1)

where the amplitude a_n and the phase α_n are real functions of T_1 . Upon substituting (3.1) into the solvability condition (2.9) and separating the resulting equation into its real and imaginary parts, we obtain

$$\omega_n(a'_n + C_n a_n) = 0 \tag{3.2a}$$

$$\omega_{n}a_{n}\alpha_{n}' + \frac{1}{8}a_{n}\sum_{p}\gamma_{np}a_{p}^{2} = 0 \qquad (3.2b)$$

where the primes designate differentiation with respect to T_1 . The solution of (3.2a) is

$$a_n = a_0 \exp[-C_n T_1] \tag{3.3}$$

and it follows that $a_n \rightarrow 0$ as $T_1 \rightarrow \infty$; so that the steady-state amplitude $a_n = 0$. Therefore, the steady-state solution has the form

$$U_{n} = O(\varepsilon) \tag{3.4}$$

Consequently, only the directly excited modes and the modes involved in an internal resonance can be part of the first approximation of the steady-state response.

3.3 The Case Where $\lambda \stackrel{\sim}{\sim} 2\omega_k$, $n \neq k$

The first case to be considered is the case of a parametric resonance where $\lambda \stackrel{\sim}{\sim} 2\omega_k$. A detuning parameter, ρ , is introduced and used to express the nearness of λ to $2\omega_k$ quantitatively according to

$$\lambda = 2\omega_{\nu} + \varepsilon \rho \tag{3.5a}$$

It follows that

$$(\lambda - \omega_k)T_0 = \omega_k T_0 + \varepsilon T_0 \rho = \omega_k T_0 + \rho T_1 \qquad (3.5b)$$

Then for $n \neq k$ the resonant terms become

$$E_n = 0 \quad \text{for } n \neq k \tag{3.5c}$$

and

$$E_{k} = -f_{kk}\bar{A}_{k}exp(i\rho T_{1}) = -\frac{f_{kk}}{2}a_{k}exp[i(\rho T_{1} - \alpha_{k})] \qquad (3.5d)$$

Substituting (3.5) into the solvability conditions (2.9) and separating into real and imaginary parts, we obtain

$$\omega_k(a_k' + C_k a_k) + \frac{f_{kk}}{2} a_k \sin\mu = 0$$
 (3.6a)

and

$$\omega_{k}a_{k}\alpha_{k}^{\prime} + \frac{1}{8}a_{k}\sum_{p=1}^{\infty}\gamma_{kp}a_{p}^{2} - \frac{f_{kk}}{2}a_{k}\cos\mu = 0$$
 (3.6b)

where

$$\mu = \rho T_1 - 2\alpha_k \tag{3.6c}$$

For the steady-state solution all the $a'_n = 0$ and $\mu' = 0$. From equations (3.3) and (3.4) it follows that $a_n \neq 0$ as $T_1 \neq \infty$ so that the steadystate amplitude $a_n = 0$ for $n \neq k$. The trivial solution $a_k = 0$ is a solution of equations (3.6). For nontrivial solutions $a_k \neq 0$, equations (3.6) reduce to

$$\omega_{k}C_{k} + \frac{t_{kk}}{2} \sin\mu = 0 \qquad (3.7a)$$

and

$$\frac{\rho\omega_k}{2} + \frac{1}{8}\gamma_{kk}a_k^2 - \frac{f_{kk}}{2}\cos\mu = 0 \qquad (3.7b)$$

where

or

$$\mu' = 0 = \rho - 2\alpha_k' \tag{3.7c}$$

Equations (3.7) may now be solved for a_k and μ . Squaring and adding (3.7a) and (3.7b) leads to

$$\omega_{k}^{2}C_{k}^{2} + \left(\frac{\rho\omega_{k}}{2} + \frac{1}{8}\gamma_{kk}a_{k}^{2}\right)^{2} = \frac{f^{2}kk}{4}$$

$$a_{k} = \sqrt{\frac{8\omega_{k}}{\gamma_{kk}} \left[-\frac{\rho}{2} \pm \sqrt{\frac{f^{2}kk}{4\omega_{k}^{2}} - C_{k}^{2}}\right]} \qquad (3.7d)$$

We note that for nontrivial solutions to exist

$$|f_{kk}| \ge 2\omega_k C_k \tag{3.7e}$$

when $\rho > 0$ and $\gamma_{kk} < 0$. When $\rho < 0$ and $\gamma_{kk} < 0$

$$\rho^{2} < \frac{f^{2}_{kk}}{\omega_{k}^{2}} - 4C_{k}^{2} \qquad (3.7f)$$

is required for nontrivial solutions. A more complete discussion of the regions where solutions exist is contained in the numerical examples in Chapter 8. We note here that $\gamma_{kk} < 0$ in those examples.

When equation (3.7d) is substituted into equations (2.2) and (2.6), the resulting steady-state solution becomes

$$U_n = O(\varepsilon) \quad n \neq k \tag{3.7g}$$

and

$$U_{k} = a_{k} \cos[(\omega_{k} + \varepsilon \frac{\rho}{2})t + \tau_{k}] + O(\varepsilon)$$
(3.7h)

where the phase $\boldsymbol{\tau}_{\boldsymbol{k}}$ is a constant that depends on the initial conditions.

The stability of the steady-state solution can now be studied by determining the behavior of the system when it is perturbed slightly. Thus, we let

$$a_{k} = a_{k} + \Delta a_{k}$$
(3.8a)
$$\mu = \hat{\mu} + \Delta \mu$$
(3.8b)

where \hat{a}_k and $\hat{\mu}$ represent steady-state values. Substituting (3.8) into (3.6) yields

$$2\omega_{k}(\hat{a}_{k}^{\prime} + \Delta a_{k}^{\prime} + C_{k}\hat{a}_{k}^{\prime} + C_{k}\Delta a_{k}^{\prime}) + f_{kk}(\hat{a}_{k}^{\prime} + \Delta a_{k}^{\prime})\sin(\hat{\mu} + \Delta \mu) = 0$$
(3.9a)

$$\omega_{k}(\hat{a}_{k} + \Delta a_{k})\alpha_{k}' + \frac{1}{8}\gamma_{kk}(a_{k} + \Delta a_{k})^{3} - \frac{f_{kk}}{2}(\hat{a}_{k} + \Delta a_{k})\cos(\hat{\mu} + \Delta \mu) = 0$$
(3.9b)

After expanding, neglecting products of small terms, and noting that $\hat{a}'_k = 0$, we obtain

$$2\omega_{k}(\Delta a_{k}^{\dagger} + C_{k}\Delta a_{k}) + 2\omega_{k}\hat{a}_{k}C_{k} + f_{kk}\Delta a_{k}\sin\hat{\mu} + f_{kk}\hat{a}_{k}\sin\hat{\mu}$$

$$+ f_{kk}\hat{a}_{k}\cos\hat{\mu}\Delta\mu = 0 \qquad (3.9c)$$

$$\omega_{k}a_{k} \left(\frac{\rho - \Delta\mu^{\dagger}}{2}\right) + \omega_{k}\Delta a_{k}\left(\frac{\rho}{2}\right) + \frac{1}{8}\gamma_{kk}\hat{a}_{k}^{3} + \frac{3}{8}\gamma_{kk}\hat{a}_{k}^{2}\Delta a_{k} - \frac{f_{kk}}{2}\hat{a}_{k}\cos\hat{\mu}$$

$$+ \frac{f_{kk}}{2}\hat{a}_{k}\sin\hat{\mu}\Delta\mu - \frac{f_{kk}}{2}\Delta a_{k}\cos\hat{\mu} = 0 \qquad (3.9d)$$

Recalling the steady-state equations and letting

 $(\Delta a_k, \Delta \mu) \propto \exp(\Omega T_1)$

we obtain

$$(2\omega_{\mathbf{k}}\Omega)\Delta \mathbf{a}_{\mathbf{k}} + (\mathbf{f}_{\mathbf{k}\mathbf{k}}\hat{\mathbf{a}}_{\mathbf{k}}\cos\hat{\boldsymbol{\mu}})\Delta\boldsymbol{\mu} = 0$$
(3.10a)

$$- (\frac{1}{4} \gamma_{kk} \hat{a}_{k}) \Delta a_{k} + (\frac{\omega_{k} \Omega}{2} + \omega_{k} C_{k}) \Delta \mu = 0 \qquad (3.10b)$$

from which

$$\omega_{k}^{2}\Omega^{2} + 2\omega_{k}^{2}C_{k}\Omega + \frac{1}{4}\gamma_{kk}\hat{a}_{k}^{2}f_{kk}\cos\hat{\mu} = 0 \qquad (3.10c)$$

or

$$\Omega^{2} + 2C_{k}\Omega + \frac{\gamma_{kk}f_{kk}a_{k}^{2}}{4\omega_{k}^{2}}\cos\mu = 0 \qquad (3.10d)$$

which has the solution

$$\Omega = -C_{k} \pm \sqrt{C_{k}^{2} - \frac{\gamma_{kk}f_{kl}^{2}}{4\omega_{k}}} OSu^{2}$$
(3.11)

and

For stable solutions the real part of $\boldsymbol{\Omega}$ must be negative, or zero i.e.,

$$\operatorname{Re}(\Omega) \leq 0 \tag{3.12}$$
or for stability
$$\operatorname{cos} \hat{\mu} \geq 0 \tag{3.13}$$

Hence, the ρ for stability can be determined from equation (3.7d). Additional discussion of the ramifications of the stability conditions will be presented in Chapter 8 in the numerical examples.

3.4 The Case Where $\lambda \approx \omega_m + \omega_k$, $m \neq k$

The case of parametric resonance where $\lambda \approx \omega_m + \omega_k$ is analyzed. The detuning parameter ρ is now used to measure the nearness of λ to $\omega_m + \omega_k$ according to:

.

$$\lambda = \omega_{\rm m} + \omega_{\rm k} + \varepsilon \rho \tag{3.14a}$$

It follows that

$$(\lambda - \omega_m)T_0 = \omega_k T_0 + \varepsilon T_0 \rho = \omega_k T_0 + \rho T_1 \qquad (3.14b)$$

and

$$(\lambda - \omega_k)T_0 = \omega_m T_0 + \varepsilon T_0 \rho = \omega_m T_0 + \rho T_1 \qquad (3.14c)$$

$$E_n = 0 \quad \text{for } n \neq k \text{ or } m \tag{3.15a}$$

$$E_{m} \exp(i\omega_{m} T_{0}) = -f_{mk} \overline{A}_{k} \exp[i(\lambda - \omega_{k}) T_{0}] = -f_{mk} \overline{A}_{k} \exp[i(\omega_{m} T_{0} + \rho T_{1})]$$
(3.15b)

and

$$E_{k}exp(i\omega_{k}T_{0}) = -f_{km}\overline{A}_{m}[i(\lambda-\omega_{m})T_{0}] = -f_{km}\overline{A}_{m}exp[i(\omega_{k}T_{0} + \rho T_{1})]$$
(3.15c)

Using the polar notation in equation (3.1), we can write

$$E_{m} = -\frac{f_{mk}a_{k}}{2} \exp[i(\rho T_{1} - \alpha_{k})]$$
 (3.15d)

and

Substituting equations (3.15) into equation (2.9), dividing by $exp(i\alpha_n)$, and separating the resulting equations into real and imaginary parts, we obtain

$$\omega_{\rm m}(a_{\rm m}' + C_{\rm m}a_{\rm m}) + \frac{f_{\rm mk}}{2} a_{\rm k} \sin\mu = 0$$
 (3.16a)

$$\omega_k(a_k + C_k a_k) + \frac{k_m}{2} a_m \sin \mu = 0$$
 (3.16b)

$$\omega_{\rm m} a_{\rm m} \alpha'_{\rm m} + \frac{1}{8} a_{\rm m} \sum_{\rm p=1}^{\infty} \gamma_{\rm mp} a_{\rm p}^2 - \frac{\tau_{\rm mk}}{2} a_{\rm k} \cos \mu = 0 \qquad (3.16c)$$

$$\omega_k a_k \alpha'_k + \frac{1}{8} a_k \sum_{p=1}^{\infty} \gamma_{kp} a_p^2 - \frac{f_{km}}{2} a_m \cos\mu = 0$$
 (3.16d)

where

$$\mu = \rho T_1 - \alpha_k - \alpha_m \tag{3.16e}$$

For $n \neq k$, or m, the amplitudes and phases are governed by equations (3.2) and (3.3); thus, these modes do not appear in the first approximation. For the steady-state solution all $a'_n = 0$ and $\mu' = 0$. It follows that $a_k = a_m = 0$ is a solution to equation (3.16), but nontrivial solution (i.e., $a_k \neq 0$ and $a_m \neq 0$) are desired. Hence, the steady-state version of equations (3.16a) and (3.16b) becomes

$$\omega_{m}C_{m}a_{m} + \frac{f_{mk}}{2}a_{k}\sin\mu = 0 \qquad (3.17a)$$

$$\omega_{k}C_{k}a_{k} + \frac{f_{km}}{2}a_{m}\sin\mu = 0 \qquad (3.17b)$$

Solving equations (3.16c,d) for α'_m and α'_k , respectively, and substituting the results into equation (3.16e), when differentiated, leads to

$$p + \frac{1}{8} \left[\frac{\gamma_{mm}}{\omega_m} + \frac{\gamma_{km}}{\omega_k} \right] a_m^2 + \frac{1}{8} \left[\frac{\gamma_{mk}}{\omega_m} + \frac{\gamma_{kk}}{\omega_k} \right] a_k^2 - \frac{1}{2} \left[\frac{f_{mk}a_k}{a_m\omega_m} + \frac{f_{km}a_m}{a_k\omega_k} \right] \cos\mu = 0$$
(3.18a)

Combining equations (3.17a) and (3.17b) leads to

$$a_{m} = \sqrt{\frac{\omega_{k} c_{k} f_{mk}}{\omega_{m} c_{m} f_{km}}} a_{k}$$
(3.18b)

Hence, for nontrivial solutions to exist f_{mk} and f_{km} must have the same sign. Substituting equation (3.18b) into equation (3.18a) and dividing by a_k^2 , we obtain

$$p\left(\frac{\omega_{k}C_{k}f_{mk}}{\omega_{m}C_{m}f_{km}}\right)^{1/2} - \frac{1}{2} \frac{f_{mk}}{\omega_{m}} \left(1 + \frac{C_{k}}{C_{m}}\right)\cos\mu + \frac{1}{8}\left(\frac{\omega_{k}C_{k}f_{mk}}{\omega_{m}C_{m}f_{km}}\right)^{1/2} \times \left[\left(\frac{\gamma_{mm}}{\omega_{m}} + \frac{\gamma_{km}}{\omega_{k}}\right)\left(\frac{\omega_{k}C_{k}f_{mk}}{\omega_{m}C_{m}f_{km}}\right) + \left(\frac{\gamma_{mk}}{\omega_{m}} + \frac{\gamma_{kk}}{\omega_{k}}\right)\right]a_{k}^{2} = 0 \quad (3.18c)$$

and hence

$$a_{k} = \sqrt{\frac{\frac{1}{2} \frac{f_{mk}}{\omega_{m}} (1 + \frac{C_{k}}{C_{m}}) \cos \mu}{\sqrt{\frac{1}{8} (\frac{\omega_{k}C_{k}f_{mk}}{\omega_{m}C_{m}f_{km}})^{1/2} \left[(\frac{\gamma_{mm}}{\omega_{m}} + \frac{\gamma_{km}}{\omega_{k}}) (\frac{\omega_{k}C_{k}f_{mk}}{\omega_{m}C_{m}f_{km}}) + \frac{\gamma_{mk}}{\omega_{m}} + \frac{\gamma_{kk}}{\omega_{k}}) \right]}}$$
(3.18d)

The steady state solution now has the following form:

$$U_n = O(\varepsilon)$$
 $n \neq k \text{ or } m$ (3.19a)

$$U_{m} = a_{m} \cos[(\omega_{m} + \varepsilon \alpha'_{m})t + \tau_{m}] + O(\varepsilon)$$
 (3.19b)

$$U_{k} = a_{k} \cos[(\omega_{k} + \varepsilon \alpha_{k}')t + \tau_{k}] + O(\varepsilon) \qquad (3.19c)$$

where τ_m and τ_k are constants depending on initial conditions. The nonlinearity adjusts the frequencies so that

$$\omega_{\rm m} + \varepsilon \alpha'_{\rm m} + \omega_{\rm k} + \varepsilon \alpha'_{\rm k} = \omega_{\rm m} + \omega_{\rm k} + \varepsilon \rho = \lambda \qquad (3.20)$$

If both solutions $(a_m = a_k = 0 \text{ and } a_m \neq 0, a_k \neq 0)$ are stable, then the initial conditions determine which solution represents the response. The stability of the system may be examined by the same method used in the case of λ near $2\omega_k$.

3.5 The Case Where $\lambda \approx \omega_k - \omega_m$, $m \neq k$

The case of parametric resonance where $\lambda \approx \omega_k - \omega_m$ is analyzed. The detuning parameter, ρ , is introduced and used to express the nearness of λ to $\omega_k - \omega_m$, according to

$$\lambda = \omega_{k} - \omega_{m} + \varepsilon \rho \qquad (3.21a)$$

. from which we can write

$$(\omega_{\mathbf{k}} - \lambda) \mathsf{T}_{0} = (\omega_{\mathbf{m}} - \varepsilon \rho) \mathsf{T}_{0} = \omega_{\mathbf{m}} \mathsf{T}_{0} - \rho \mathsf{T}_{1}$$
(3.21b)

and

$$(\lambda + \omega_{m})T_{0} = (\omega_{k} + \varepsilon\rho)T_{0} = \omega_{k}T_{0} + \rho T_{1} \qquad (3.21c)$$

$$E_n = 0 \text{ for } n \neq k \text{ or } m \tag{3.22a}$$

$$E_{k} \exp[i\omega_{k}T_{0}] = -f_{km}A_{m}\exp[i(\lambda + \omega_{m})T_{0}] = -f_{km}A_{m}\exp[i(\omega_{k}T_{0} + \rho T_{1})]$$
(3.22b)

and

$$E_{m} \exp[i\omega_{m}T_{0}] = -f_{mk}A_{k} \exp[i(\omega_{m} - \lambda)T_{0}] = -f_{mk}A_{k} \exp[(i\omega_{m}T_{0} - \rho T_{1})]$$
(3.22c)

Using the polar notation in equation (3.1), we can write

$$E_{k} = -\frac{f_{km}a_{m}}{2} \exp[i(\rho T_{1} + \alpha_{m})] \qquad (3.22d)$$
and

$$E_{m} = -\frac{f_{mk}a_{k}}{2} \exp[i(-\rho T_{1} + \alpha_{k})] \qquad (3.22e)$$

Substituting equation (3.22) into equation (2.9), dividing by $exp(i\alpha_k)$ and $exp(i\alpha_m)$, respectively, and separating the resulting equations into real and imaginary parts leads to

$$\omega_{\rm m}(a_{\rm m}' + C_{\rm m}a_{\rm m}) - \frac{f_{\rm m}k^{\rm a}k}{2}\sin\mu = 0$$
 (3.23a)

$$\omega_k(a_k' + C_k a_k) + \frac{f_{km} a_m}{2} \sin \mu = 0$$
 (3.23b)

$$\omega_{m}a_{m}\alpha_{m}' + \frac{a_{m}}{8}\sum_{p=1}^{\infty}\gamma_{mp}a_{p}^{2} - \frac{f_{mk}a_{k}}{2}\cos\mu = 0 \qquad (3.23c)$$

$$\omega_{k}a_{k}\alpha_{k}' + \frac{a_{k}}{8}\sum_{p=1}^{\infty}\gamma_{kp}a_{p}^{2} - \frac{f_{km}a_{m}}{2}\cos\mu = 0$$
 (3.23d)

where

 $\mu = \rho T_{1} - \alpha_{k} + \alpha_{m}$ (3.23e)

For n \neq k or m, the amplitudes and phases are governed by equations (3.2) and (3.3). For the steady-state solution all $a'_n = 0$ and $\mu' = 0$. It follows that $a_k = a_m = 0$ is a solution to equations (3.23), but nontrivial solutions (i.e., $a_k \neq 0$ and $a_m \neq 0$) are desired. Hence, the steady-state version of equations (3.23) becomes

$$\omega_{\rm m} C_{\rm m} a_{\rm m} - \frac{f_{\rm m} k^{\rm a} k}{2} \sin\mu = 0 \qquad (3.24a)$$

$$\omega_k C_k a_k + \frac{f_{km} a_m}{2} \sin\mu = 0 \qquad (3.24b)$$

and

$$h + \frac{1}{8} \left[\frac{-\gamma_{mm}}{\omega_{m}} + \frac{\gamma_{km}}{\omega_{k}} \right] a_{m}^{2} + \frac{1}{8} \left[\frac{-\gamma_{mk}}{\omega_{m}} + \frac{\gamma_{kk}}{\omega_{k}} \right] a_{k}^{2}$$
$$- \frac{1}{2} \left[\frac{-f_{mk}a_{k}}{a_{m}\omega_{m}} + \frac{f_{km}a_{m}}{a_{k}\omega_{k}} \right] \cos \mu = 0 \qquad (3.24c)$$

which leads to

$$a_{\rm m} = \sqrt{-\frac{\omega_{\rm k} C_{\rm k} f_{\rm mk}}{\omega_{\rm m} C_{\rm m} f_{\rm km}}} a_{\rm k}$$
(3.25)

Thus, a solution can exist only if f_{km} and f_{mk} have the different signs. In this case, the result for a_k can be obtained from (3.18d) by simply changing the sign of ω_m .

3.6 Summary

Various cases concerning a parametric resonance have been discussed. The following observations are pertinent in the absence of an internal resonance:

- (a) Only modes which are directly excited by a parametric resonance are part of the steady-state response.
- (b) For combination-type resonances the lower mode appears to be dominant in the response.
- (c) The steady-state response is essentially that of the forced linear-oscillation modes with the frequency being adjusted.
- (d) Stability criteria have been outlined when more than one stable solution exists, the initial conditions will determine the response.

More insight into the response in the presence of parametric resonance will be gained in Chapter 8 where specific numerical examples are presented.

CHAPTER FOUR

THE EFFECT OF INTERNAL RESONANCE

In this chapter we discuss the effect of an internal resonance. Specifically, we examine in detail the case where $\omega_k \gtrsim 3\omega_m$. Part of the rationale for this choice is that this combination occurs in physical systems such as the beam considered in the numerical example in Chapter 8.

The parametric resonances of the last chapter are examined in the presence of this internal resonance. See Chapter 2 for a review of internal resonance.

An internal resonance emerges when equation (2.8a) is satisfied. When the form of equation (2.8a) is

$$\omega_{\mathbf{k}} = \omega_{\mathbf{m}} + \omega_{\mathbf{p}} + \omega_{\mathbf{q}} + \varepsilon\sigma \tag{4.1}$$

it follows from equation (2.7) that the terms involved in the internal resonance are given by the following:

$$I_{k} = \hat{\Gamma}_{k} A_{m} A_{p} A_{q} \exp(-i\sigma T_{1})$$
(4.2a)

$$I_{m} = \hat{\Gamma}_{m} A_{k} \overline{A}_{p} \overline{A}_{q} \exp(i\sigma T_{1})$$
(4.2b)

$$I_{p} = \hat{\Gamma}_{p} A_{k} \overline{A}_{m} \overline{A}_{q} \exp(i\sigma T_{1})$$
(4.2c)

$$I_{q} = \hat{\Gamma}_{q} A_{k} \overline{A}_{m} \overline{A}_{p} \exp(i\sigma T_{1})$$
(4.2d)

$$\hat{\Gamma}_{k} = \Gamma_{k}mpq + \Gamma_{k}pmq + \Gamma_{k}pqm + \Gamma_{k}qpm + \Gamma_{k}qpm + \Gamma_{k}qpp \qquad (4.3a)$$

$$\hat{\Gamma}_{m} = \Gamma_{m}kpq + \Gamma_{m}kqp + \Gamma_{m}pkq + \Gamma_{m}qkp + \Gamma_{m}pqk + \Gamma_{m}pqk \qquad (4.3b)$$

$$\Gamma_{p} = \Gamma_{pmqk} + \Gamma_{pqmk} + \Gamma_{pmkq} + \Gamma_{pqkm} + \Gamma_{pkmq} + \Gamma_{pkqm}(4.3c)$$

$$\hat{\Gamma}_{q} = \Gamma_{qmpk} + \Gamma_{qpmk} + \Gamma_{qmkp} + \Gamma_{qpkm} + \Gamma_{qkmp} + \Gamma_{qkpm}(4.3d)$$
When the form of equation (2.8a) is

$$\omega_{\mathbf{k}} = \omega_{\mathbf{m}} + 2\omega_{\mathbf{p}} + \varepsilon\sigma \tag{4.4}$$

it follows from equation (2.7) that

$$I_{k} = \hat{\Gamma}_{k} A_{m} A_{p}^{2} \exp(-i\sigma T_{1})$$
(4.5a)

$$I_{m} = \hat{\Gamma}_{m} A_{k} \overline{A}_{p}^{2} \exp(i\sigma T_{1})$$
(4.5b)

$$I_{p} = \hat{\Gamma}_{p} A_{k} \overline{A}_{m} \overline{A}_{p} \exp(i\sigma T_{1})$$
(4.5c)

$$\Gamma_{k} = \Gamma_{kmpp} + \Gamma_{kpmp} + \Gamma_{kppm}$$
(4.6a)

$$\hat{\Gamma}_{m} = \Gamma_{mppk} + \Gamma_{mpkp} + \Gamma_{mkpp}$$
(4.6b)

$$\Gamma_{p} = \Gamma_{pmpk} + \Gamma_{ppmk} + \Gamma_{ppkm} + \Gamma_{pmkp} + \Gamma_{pkpm} + \Gamma_{pkmp}$$
(4.6c)

When the form of equation (2.8a) is

$$\omega_{\mathbf{k}} = 3\omega_{\mathbf{m}} + \varepsilon\sigma \tag{4.7}$$

it follows that

$$I_{k} = \hat{\Gamma}_{k} A_{m}^{3} \exp(-i\sigma T_{1})$$
(4.8a)

$$I_{m} = \hat{\Gamma}_{m} A_{k} \overline{A}_{m}^{2} \exp(i\sigma T_{1})$$
(4.8b)

where

$$\Gamma_{k} = \Gamma_{kmmm}$$
 (4.9a)

$$\hat{\Gamma}_{m} = \Gamma_{mmmk} + \Gamma_{mmkm} + \Gamma_{mkmm}$$
(4.9b)

When the form of equation (2.8a) is

$$\omega_{\mathbf{k}} = \omega_{\mathbf{m}} + \omega_{\mathbf{p}} - \omega_{\mathbf{q}} + \varepsilon\sigma \tag{4.10}$$

it follows that

$$I_{k} = \hat{\Gamma}_{k} A_{m} A_{p} \overline{A}_{q} \exp(-i\sigma T_{1})$$
(4.11a)

$$I_{m} = \hat{\Gamma}_{m} A_{k} \overline{A}_{p} A_{q} \exp(i\sigma T_{1})$$
(4.11b)

$$I_{p} = \hat{\Gamma}_{p} A_{k} \overline{A}_{m} A_{q} \exp(i\sigma T_{1})$$
 (4.11c)

$$I_{q} = \hat{\Gamma}_{q} A_{m} A_{p} \overline{A}_{k} exp(-i\sigma T_{1})$$
(4.11d)

where $\hat{\Gamma}_{k}$, $\hat{\Gamma}_{m}$, $\hat{\Gamma}_{p}$, and $\hat{\Gamma}_{q}$ are given by equations (4.3).

Other possible internal resonances can be treated in a similar way.

4.1 The Case Where
$$\lambda \stackrel{\sim}{\sim} 2\omega_{\rm m}$$
 and $\omega_{\rm k} \stackrel{\sim}{\sim} 3\omega_{\rm m}$.

The first case to be considered is $\lambda \stackrel{\sim}{\sim} 2\omega_m$, $\omega_k \stackrel{\sim}{\sim} 3\omega_m$. The detuning parameters are defined in the following equations.

$$\omega_{\mathbf{k}} = 3\omega_{\mathbf{m}} + \varepsilon\sigma \tag{4.12a}$$

$$\lambda = 2\omega_{\rm m} + \epsilon\rho = \omega_{\rm k} - \omega_{\rm m} + \epsilon(\rho - \sigma) \qquad (4.12b)$$

$$E_{m} = -f_{mk}A_{k}exp[-i(\rho - \sigma)T_{1}]$$

- f_mmA_mexp(i\sigmaT_{1}) (4.13a)

$$E_{k} = -f_{km}A_{m}exp[i(\rho - \sigma)T_{1}] \qquad (4.13b)$$

$$I_{m} = \hat{\Gamma}_{m} A_{k} \overline{A}_{m}^{2} \exp(i\sigma T_{1})$$
(4.14a)

$$I_{k} = \Gamma_{k} A_{m}^{3} \exp(-i\sigma T_{1})$$
(4.14b)

Using the polar form (3.1), substituting (4.13) and (4.14) into equation (2.9), neglecting all except the m-th and the k-th modes, and separating the results into real and imaginary parts, we obtain

$$\omega_{\rm m}(a_{\rm m}' + C_{\rm m}a_{\rm m}) - \frac{\Gamma_{\rm m}}{8} a_{\rm k}a_{\rm m}^2 \sin\mu + \frac{f_{\rm mm}a_{\rm m}}{2} \sin\beta$$
$$- \frac{f_{\rm mk}}{2} a_{\rm k}\sin(\beta - \mu) = 0 \qquad (4.15a)$$

$$\omega_{m}a_{m}\alpha_{m}' + \frac{1}{8}a_{m}(\gamma_{mm}a_{m}^{2} + \gamma_{mk}a_{k}^{2}) + \frac{1}{8}a_{k}a_{m}^{2}\cos\mu - \frac{f_{mm}a_{m}}{2}\cos\beta - \frac{f_{mk}}{2}a_{k}\cos(\beta - \mu) = 0 \qquad (4.15b)$$

$$\omega_{k}(a_{k}' + C_{k}a_{k}) + \frac{\Gamma_{k}}{8}a_{m}^{3}\sin\mu + \frac{f_{km}a_{m}}{2}\sin(\beta - \mu) = 0 \quad (4.16a)$$

$$\omega_{k}a_{k}\alpha_{k}' + \frac{a_{k}}{8} (\gamma_{km}a_{m}^{2} + \gamma_{kk}a_{k}^{2}) + \frac{f_{k}}{8}a_{m}^{3}\cos\mu - \frac{f_{km}a_{m}}{2}\cos(\beta - \mu) = 0$$
(4.16b)

where

$$\mu = \sigma T_1 - 3\alpha_m + \alpha_k \tag{4.17a}$$

$$\beta = \rho T_1 - 2\alpha_m \tag{4.17b}$$

For the steady-state solution it follows that $a_m = a_k = 0$ is a solution. For nontrivial solutions

$$\mu' = \sigma + \alpha_{k}' - 3\alpha_{m}' = 0 \tag{4.18a}$$

$$\beta' = \rho - 2\alpha_m' = 0$$
 (4.18b)

which leads to

$$\alpha'_{\rm m} = \frac{\rho}{2}$$
 and $\alpha'_{\rm k} = \frac{3\rho}{2} - \sigma$ (4.19)

Equations (4.16) become

$$\omega_{m}C_{m}a_{m} + \frac{f_{mm}a_{m}}{2}\sin\beta - \frac{f_{mk}}{2}a_{k}\sin(\beta - \mu) - \frac{1}{8}\hat{\Gamma}_{m}a_{k}a_{m}^{2}\sin\mu = 0$$
(4.20a)
$$\omega_{m}\frac{\rho a_{m}}{2} + \frac{\gamma_{mm}}{8}a_{m}^{3} + \frac{\gamma_{mk}a_{m}}{8}a_{k}^{2} - \frac{f_{mm}a_{m}}{2}\cos\beta - \frac{f_{mk}a_{k}}{2}\cos(\beta - \mu)$$

$$+ \frac{1}{8}\hat{\Gamma}_{m}a_{k}a_{m}^{2}\cos\mu = 0$$
(4.20b)

$$\omega_k C_k a_k + \frac{f_{km} a_m}{2} \sin(\beta - \mu) + \frac{1}{8} \hat{\Gamma}_k a_m^3 \sin\mu = 0$$
 (4.20c)

$$\omega_{k}a_{k}\left(\frac{3\rho}{2}-\sigma\right) + \frac{\gamma_{mk}}{8}a_{k}a_{m}^{2} + \frac{\gamma_{kk}}{8}a_{k}^{3} - \frac{f_{km}a_{m}}{2}\cos(\beta-\mu) + \frac{1}{8}\hat{\Gamma}_{k}a_{m}^{3}\cos\mu = 0 \qquad (4.20d)$$

A detailed discussion of the behavior of the solution of equations (4.20) is given in Chapter 8 in the numerical examples.

4.2 The Case Where $\lambda \approx 2\omega_k$ and $\omega_k \approx 3\omega_m$.

The next case to be considered is $\lambda \stackrel{\sim}{\sim} 2\omega_k$, $\omega_k \stackrel{\sim}{\sim} 3\omega_m$. The detuning parameters are defined as follows:

 $\lambda = 2\omega_{\mathbf{k}} + \varepsilon \rho \tag{4.22a}$

$$\omega_{\mathbf{k}} = 3\omega_{\mathbf{m}} + \varepsilon\sigma \tag{4.22b}$$

The parametric resonant terms are of the form

$$E_{m} = 0$$
 (4.22c)

$$E_{k} = -f_{kk}\overline{A}_{k}\exp(i\rho T_{1}) \qquad (4.22d)$$

and the internal resonant terms are the same as those in equations (4.14). Substituting (4.22cd) and (4.14) into equation (2.9) and separating the result into real and imaginary parts, we find that

$$\omega_{\rm m}(a_{\rm m}^{\prime} + C_{\rm m}a_{\rm m}) - \frac{1}{8}\hat{\Gamma}_{\rm m} a_{\rm k}a_{\rm m}^{2}\sin\mu = 0$$
 (4.23a)

$$\Delta_{m}a_{m}\alpha_{m}' + \frac{a_{m}}{8} (\gamma_{mm}a_{m}^{2} + \gamma_{mk}a_{k}^{2}) + \frac{1}{8} \hat{\Gamma}_{m}a_{k}a_{m}^{2}\cos\mu = 0 \qquad (4.23b)$$

$$\omega_k(a'_k + C_k a_k) + \frac{f_k k^a k}{2} \sin\beta + \frac{1}{8} \hat{\Gamma}_k a_m^3 \sin\mu = 0$$
 (4.23c)

$$\omega_{k}a_{k}\alpha_{k}' + \frac{a_{k}}{8}(\gamma_{km}a_{m}^{2} + \gamma_{kk}a_{k}^{2}) - \frac{f_{kk}a_{k}}{2}\cos\beta + \frac{1}{8}\hat{\Gamma}_{k}a_{m}^{3}\cos\mu = 0$$
(4.23d)

where

$$\beta = \rho T_1 - 2\alpha_k \tag{4.24a}$$

$$\mu = \sigma T_1 + \alpha_k - 3\alpha_m \tag{4.24b}$$

For steady state solutions, it is easily seen that $a_m = a_k = 0$ is a solution. For nontrivial solutions equations (4.24) become

$$\beta' = \rho - 2\alpha_k' = 0$$
 (4.25a)

$$\mu' = \sigma + \alpha_{k}' - 3\alpha_{m}' = 0$$
 (4.25b)

so that

$$\alpha_k' = \frac{\rho}{2}$$
 and $\alpha_m' = \frac{\sigma + \frac{\rho}{2}}{3}$ (4.26a)

and equations (4.23) become

$$\omega_{\rm m} C_{\rm m} - \frac{1}{8} \hat{\Gamma}_{\rm m} a_{\rm k} a_{\rm m} \sin\mu = 0$$
 (4.27a)

$$\omega_{\rm m} \frac{\sigma + \frac{\beta}{2}}{3} + \frac{\gamma_{\rm mm} a_{\rm m}^2}{8} + \frac{\gamma_{\rm mk} a_{\rm k}^2}{8} + \frac{1}{8} \hat{\Gamma}_{\rm m} a_{\rm k} a_{\rm m} \cos\mu = 0 \qquad (4.27b)$$

$$\omega_k C_k a_k + \frac{f_k k^a k}{2} \sin\beta + \frac{1}{8} \hat{\Gamma}_k a_m^3 \sin\mu = 0$$
 (4.27c)

$$\omega_{k}a_{k}\frac{\rho}{2} + \frac{\gamma_{km}}{8}a_{k}a_{m}^{2} + \frac{\gamma_{kk}}{8}a_{k}^{3} - \frac{f_{kk}a_{k}}{2}\cos\beta$$
$$+ \frac{1}{8}\hat{\Gamma}_{k}a_{m}^{3}\cos\mu = 0 \qquad (4.27d)$$

A detailed discussion of the behavior of the solutions of equations (4.27) is given in Chapter 8 in the numerical examples.

It should be noted here that it is possible for $a_m = 0$ and $a_k \neq 0$, in which case equations (4.27c,d) become

$${}^{\omega}_{k}C_{k}a_{k} + \frac{f_{k}k^{a}k}{2}\sin^{\beta} = 0$$
 (4.28a)

and

$${}^{\omega}_{k} a_{k} \frac{\rho}{2} + \frac{\gamma_{kk}}{8} a_{k}^{3} - \frac{f_{kk}a_{k}}{2} \cos\beta = 0$$
 (4.28b)

This is exactly the same type of response that occurs when there is no internal resonance, equations (3.7a,b). Thus, the only mode which responds is the one which is directly excited. This is in contrast to the case where the frequency of the excitation λ is near $2\omega_{\rm m}$ and a single-mode response is not possible.

4.3 The Case Where $\lambda \stackrel{\sim}{\sim} 4\omega_m$ and $\omega_k \stackrel{\sim}{\sim} 3\omega_m$.

The last case to be considered is for $\lambda \stackrel{\sim}{\sim} 4\omega_{\rm m}$ and $\omega_{\rm k} \stackrel{\sim}{\sim} 3\omega_{\rm m}$. In the presence of this internal resonance the parametric resonance can be expressed equally well as $\lambda \stackrel{\sim}{\sim} \omega_{\rm m} + \omega_{\rm k}$. We may write

$$\lambda = 4\omega_{\rm m} + \varepsilon\rho = \omega_{\rm m} + \omega_{\rm k} + \varepsilon(\rho - \sigma) \tag{4.29a}$$

where

$$\omega_{\rm k} = 3\omega_{\rm m} + \varepsilon\sigma \tag{4.29b}$$

The terms involved in the parametric resonance are

$$E_{m} = -f_{mk}\overline{A}_{k}\exp[i(\rho - \sigma)T_{1}] \qquad (4.30a)$$

$$E_{k} = -f_{km}\overline{A}_{m}exp[i(\rho - \sigma)T_{I}]$$
(4.30b)

Substituting equations (4.30) and (4.14) into equation (2.9), introducing the polar form (3.1), and separating the result into real and imaginary parts, we obtain

$$\omega_{m}(a_{m}' + C_{m}a_{m}) + \frac{f_{m}k^{a}k}{2} \sin\beta - \frac{1}{8}\hat{\Gamma}_{m}a_{k}a_{m}^{2}\sin\mu = 0 \qquad (4.31a)$$

$$\omega_{m}a_{m}\alpha_{m}' + \frac{1}{8}a_{m}(\gamma_{mm}a_{m}^{2} + \gamma_{mk}a_{k}^{2}) - \frac{f_{m}k^{a}k}{2}\cos\beta + \frac{1}{8}\hat{\Gamma}_{m}a_{k}a_{m}^{2}\cos\mu = 0 \qquad (4.31b)$$

$$\omega_{k}(a_{k} + C_{k}a_{k}) + \frac{f_{km}a_{m}}{2} \sin\beta + \frac{1}{8}\hat{\Gamma}_{k}a_{m}^{3}\sin\mu = 0 \qquad (4.31c)$$

$$\omega_{k}a_{k}\alpha_{k}' + \frac{a_{k}}{8}(\gamma_{km}a_{m}^{2} + \gamma_{kk}a_{k}^{2}) - \frac{f_{km}a_{m}}{2}\cos\beta + \frac{\hat{\Gamma}}{8}a_{m}^{3}\cos\mu = 0 \qquad (4.31d)$$

where

$$\beta = (\rho - \sigma) T_1 - \alpha_k - \alpha_m \qquad (4.32a)$$

$$\mu = \sigma T_1 + \alpha_k - 3\alpha_m \tag{4.32b}$$

For the steady-state solution, it follows that $a_m = a_k = 0$. For the nontrivial solution,

$$\beta' = \rho - \sigma - \alpha_{\mathbf{k}}' - \alpha_{\mathbf{m}}' = 0 \qquad (4.33a)$$

and

$$\mu' = \sigma + \alpha_k' - 3\alpha_m' = 0 \tag{4.33b}$$

which lead to

$$\alpha_{\rm m}' = \frac{\rho}{4}$$
 and $\alpha_{\rm k}' = \frac{3\rho}{4} - \sigma$ (4.33c)

The steady-state equations can be written as

$$\omega_{\rm m} C_{\rm m} a_{\rm m} + \frac{f_{\rm mk} a_{\rm k}}{2} \sin\beta - \frac{1}{8} \hat{\Gamma}_{\rm m} a_{\rm k} a_{\rm m}^2 \sin\mu = 0 \qquad (4.34a)$$

$$\omega_{m} a_{m} \frac{\rho}{4} + \frac{a_{m}^{3} \gamma_{mm}}{8} + \frac{a_{m} a_{k}^{2} \gamma_{mk}}{8} - \frac{f_{mk} a_{k}}{2} \cos\beta + \frac{1}{8} \hat{\Gamma}_{m} a_{k} a_{m}^{2} \cos\beta = 0 \qquad (4.34b)$$

$${}^{\omega_{k}}C_{k}a_{k}^{a} + \frac{f_{km}a_{m}}{2} \sin\beta + \frac{1}{8}\hat{\Gamma}_{k}a_{m}^{3}\sin\mu = 0$$
 (4.34c)

$$\omega_{k}a_{k} \frac{3\rho - 4\sigma}{4} + \frac{a_{k}^{3}\gamma_{kk}}{8} + \frac{a_{k}a_{m}^{2}\gamma_{km}}{8} - \frac{f_{km}a_{m}}{2}\cos\beta + \frac{1}{8}\hat{\Gamma}_{k}a_{m}^{3}\cos\mu = 0 \qquad (4.34d)$$

A detailed discussion of the response predicted by these equations is given in Chapter 8.

4.4 Summary

Internal-resonance terms involving $\omega_k \approx 3\omega_m$ have been developed. The response in the presence of this internal resonance and several parametric resonances has been examined. The following comments are pertinent:

a. It has been shown that, even in the presence of internal resonance, it is possible for only the mode which is excited to enter the response. The behavior is then essentially identical to that in the absence of internal resonance.

b. In all cases discussed, strong modal interactions due to the presence of internal resonance are possible. A full discussion of the dominance of one mode over the others is presented in Chapter 8, which contains the numerical examples.

CHAPTER FIVE

THE ANALYSIS OF SYSTEMS HAVING REPEATED FREQUENCIES IN THE ABSENCE OF INTERNAL AND PARAMETRIC RESONANCES

In the last three chapters we have discussed the response of a nonlinear system subject to a parametric excitation. Throughout the entire analysis we have considered the natural frequencies to be distinct. This is valid for the beam-column we discuss in Chapter 8 as well as for many other physical systems. The analysis contained in the next three chapters deals with the situation where one of the natural frequencies is repeated. As an example, this occurs in plates subject to aerodynamic loading at the onset of flutter. Two natural structural frequencies merge due to aeroelastic coupling. For a discussion of the mechanism of flutter the reader is referred to Dowell (1975).

In this chapter we follow a format similar to that of Chapter 2. The governing equations (2.1) are modified to account for the repeated frequency and the aerodynamic loading. In addition, scaling factors will be introduced in the form ε^{δ_i} to insure that terms interact at the proper order. The method of multiple scales is utilized to generate the solvability conditions. The conditions that predict the onset of flutter are developed and limit-cycle behavior is demonstrated. We begin now with the governing equations when one natural frequency, ω_j , is repeated, thus we write

$$\ddot{U}_{n} + \omega_{n}^{2}U_{n} = \varepsilon^{\delta}F[-(2\cos \lambda t)\sum_{m=1}^{\infty} f_{nm} U_{m}]$$

$$- \varepsilon^{\delta}C[2C_{n} \dot{U}_{n}] + \varepsilon^{\delta}N[\sum_{m=1}^{\infty} \sum_{p=1}^{\infty} \sum_{q=1}^{\infty} \Gamma_{nmpq}U_{m}U_{p}U_{q}]$$

$$+ \varepsilon^{\delta}A[A\sum_{m=1}^{\infty} \beta_{nm}U_{m}] \quad n = 1,2... \quad n \neq k \quad (5.1a)$$

and

$$\ddot{U}_{k} + \omega_{j}^{2}U_{k} + U_{j} = \varepsilon^{\delta}F[-(2\cos \lambda t) \sum_{m=1}^{\infty} f_{km}U_{m}]$$

$$- \varepsilon^{\delta}C[2C_{k}\dot{U}_{k}] + \varepsilon^{\delta}N[\sum_{m=1}^{\infty} \sum_{p=1}^{\infty} q_{=1}^{\infty} \Gamma_{kmpq}U_{m}U_{p}U_{q}]$$

$$+ \varepsilon^{\delta}A[A \sum_{m=1}^{\infty} \beta_{km}U_{k}] \quad \text{where} \quad k = j + 1 \quad (5.1b)$$

where k = j + 1, Λ is a detuning parameter (associated with the aerodynamic loading) to be explained later in a numerical example, and δ_F , δ_C , δ_N , and δ_Λ are constants to be chosen so as to insure the proper interaction for the various resonances to be considered.

We note that, if the nonlinear, damping and forcing terms were zero, the k-th mode would grow indefinitely, and the system would be in flutter. In the present situation, we expect the k-th mode to have a much larger amplitude than the j-th mode, and we have no indication of the other amplitudes. Thus, using the method of multiple scales, we obtain a uniformly valid approximation of the solution of equations (5.1) in the form

$$U_{n} = \varepsilon^{-\delta_{n}} [U_{n_{0}}(T_{0}, T_{1}, T_{2}) + \varepsilon^{\delta_{0}} U_{n_{1}}(T_{0}, T_{1}, T_{2}) + \varepsilon^{2\delta_{0}} U_{n_{2}}(T_{0}, T_{1}, T_{2})] + \dots$$
(5.2a)

where the $\boldsymbol{\delta}_n$ and $\boldsymbol{\delta}_0$ are more constants to be determined, and

$$T_{n} = \varepsilon^{n\delta_{0}}t$$
 (5.2b)

The derivatives become

$$\frac{d}{dt} = D_0 + \varepsilon^{\delta_0} D_1 + \varepsilon^{2\delta_0} D_2 + \dots \qquad (5.3a)$$

and

$$\frac{d^2}{dt^2} = D_0^2 + 2\varepsilon^{\delta_0} D_0 D_1 + \varepsilon^{2\delta_0} (D_1^2 + 2D_0 D_2) + \dots$$
 (5.3b)

where

$$D_n \equiv \frac{\partial}{\partial T_n}$$
(5.3c)

Substituting equations (5.2) and (5.3) into equations (5.1) yields $D_{0}^{2}U_{n_{0}} + \omega_{n}^{2}U_{n_{0}} + \varepsilon^{\delta_{0}}[D_{0}^{2}U_{n_{1}} + \omega_{n}^{2}U_{n_{1}} + 2D_{0}D_{1}U_{n_{0}}]$ $+ \varepsilon^{2\delta_{0}}[D_{0}^{2}U_{n_{2}} + \omega_{n}^{2}U_{n_{2}} + 2D_{0}D_{2}U_{n_{0}} + D_{1}^{2}U_{n_{0}} + 2D_{0}D_{1}U_{n_{1}}]$ $+ \Delta_{kn} [\varepsilon^{\delta_{k}-\delta_{j}}U_{j_{0}} + \varepsilon^{\delta_{0}+\delta_{k}-\delta_{j}}U_{j_{1}} + \varepsilon^{2\delta_{0}+\delta_{k}-\delta_{j}}U_{j_{2}}]$

$$+ \epsilon^{\delta_{c}} [2c_{n} D_{0} U_{n_{0}}] + \epsilon^{\delta_{c}^{+\delta_{0}}} [2c_{n} (D_{0} U_{n_{1}}^{+} + D_{1} U_{n_{0}}^{-})]$$

$$+ \epsilon^{\delta_{c}^{+}} [2c_{n} (D_{2} U_{n_{0}}^{-} + D_{1} U_{n_{1}}^{-}] - [\Lambda \sum_{m=1}^{\infty} \beta_{nm} (U_{m_{0}}^{-} + \epsilon^{\delta_{0}} U_{m_{1}}^{-}]$$

$$+ \epsilon^{2\delta_{0}} U_{m}^{-} \epsilon^{\delta_{n}^{-\delta_{m}}}] + [\sum_{m=1}^{\infty} f_{nm}^{-} (U_{m_{0}}^{-} + \epsilon^{\delta_{0}} U_{m_{1}}^{-}] + \epsilon^{2\delta_{0}} U_{m_{2}}^{-}] \times$$

$$+ \epsilon^{\delta_{c}^{+\delta_{n}^{-\delta_{m}}}] (2cos\lambda T_{0}) - \sum_{m=1}^{\infty} p_{m=1}^{\infty} q_{m=1}^{\infty} \Gamma_{nmpq}^{-} (U_{m_{0}}^{-} + \epsilon^{\delta_{0}} U_{m_{1}}^{-}]$$

$$+ \epsilon^{2\delta_{0}} U_{m_{2}}^{-}) \times (U_{p_{0}}^{-} + \epsilon^{\delta_{0}} U_{p_{1}}^{-}] + \epsilon^{2\delta_{0}} U_{p_{2}}^{-}) \times (U_{q_{0}}^{-} + \epsilon^{\delta_{0}} U_{q_{1}}^{-}]$$

$$+ \epsilon^{2\delta_{0}} U_{q_{2}}^{-}) \times \epsilon^{\delta_{N}^{+\delta_{n}^{-\delta_{m}^{-\delta_{p}^{-\delta_{q}^{-\delta$$

where

k = j + 1	•	(5.4b)

$$\omega_j = \omega_k \tag{5.4c}$$

and

 $\Delta_{kn} = (5.4d)$ 0 otherwise

Equation (5.4a) is the basis for the analysis of systems having a repeated frequency, and we refer to it again in subsequent chapters.

In the present chapter, we consider the case in which there is neither a parametric nor an internal resonance. Thus, here we consider the flutter condition for the linear system and obtain the limit-cycle behavior of the nonlinear system. In this chapter, we need to use only a two-term expansion, but in Chapter 6 where various parametric resonances are studied, we need a three-term expansion.

For a reference, we use the j-th mode and hence let

 $\delta_{i} = 0 \tag{5.5}$

In order to have the effect of damping included in a first approximation, we must let

 $\delta_{\rm C} = \delta_0 \tag{5.6a}$

And to account for the repeated frequency here, we must let

 $\delta_k - \delta_j = \delta_0$ or simply $\delta_k = \delta_0$ (5.6b)

We presume that the j-th and k-th modes are the ones primarily involved in the representation of the flutter system. Thus we focus on these mode. It appears that to bring the effect of the aerodynamic loading into the first approximation, we must put

$$\delta_{\Lambda} + \delta_{n} - \delta_{m} = \delta_{0}$$

We note that, when n=j and m=k, this statement leads to

 $\delta_{\Lambda} = 2\delta_0 \tag{5.7}$

and when n=k and m=j, it leads to

$$\delta_{\Lambda} = 0$$

If $\delta_{\Lambda} = 0$, the aerodynamic term in the equation governing the U_j is elevated to $O(\epsilon^{-\delta_0})$, which is unacceptable. On the other hand, if $\delta_{\Lambda} = 2\delta_0$, such unacceptable result follows; thus, we let $\delta_{\Lambda} = 2\delta_0$.

To account for the nonlinear terms in the first approximation, we must put

$$\delta_{\mathbf{N}} + \delta_{\mathbf{n}} - \delta_{\mathbf{m}} - \delta_{\mathbf{n}} - \delta_{\mathbf{q}} = \delta_{\mathbf{0}}$$

When n=j and m=p=q=k, this leads to

 $\delta_{N} = 4\delta_{0} \tag{5.8}$

For other combinations of modes, one obtains different results for $\delta_{N},$ but these all lead to inconsistencies.

Here we do not consider the effect of a parametric excitation. Thus, for the time being, we consider

 $f_{mn} = 0$ for all n and m (5.9)

Finally, when $n \neq j$ or k, we let

$$0 \leq \delta_{n} \leq \delta_{0} \tag{5.10}$$

5.1 The Case of Modes Having Repeated Frequencies

A. The Linear Response - The Flutter Condition

In the absence of both an internal and a parametric resonance the only modes remaining in the analysis are the j-th and k-th modes. For the linear response, we let

$$\Gamma_{nmpq} = 0 \quad \text{for all } n, m, p, q \tag{5.11}$$

Substituting equations (5.5) thru (5.11) into equation (5.4) and equating coefficients of $e^{n\delta_0}$, we obtain

$$D_{0}^{2}U_{i0} + \omega_{i}^{2}U_{i0} = 0$$
 (5.12a)

$$D_0^2 U_{k^0} + \omega_1^2 U_{k^0} = 0$$
 (5.12b)

$$D_{0}^{2}U_{j^{1}} + \omega_{j}^{2}U_{j^{1}} = -2D_{0}D_{1}U_{j^{0}} - 2C_{j}D_{0}U_{j^{0}} + \Lambda\beta_{jk}U_{k^{0}}$$
(5.12c)

$$D_0^2 U_{k^1} + \omega_j^2 U_{k^1} = -2D_0 D_1 U_{k^0} - U_{j^0} - 2C_k D_0 U_{k^0}$$
(5.12d)

The solution of equations (5.12a) and (5.12b) can be written in the form

$$U_{j^0} = A_j(T_1) \exp(i\omega_j T_0) + cc$$
 (5.13a)

$$U_{k^0} = A_k(T_1) \exp(i\omega_i T_0) + cc$$
 (5.13b)

Substituting equations (5.13) into equations (5.12c) and (5.12d), and separating the result into real and imaginary parts, we obtain the following conditions for the elimination of secular terms

$$-2i\omega_{j}(A_{j}^{\prime}+C_{j}A_{j}) + \Lambda\beta_{jk}A_{k} = 0$$
 (5.14a)

$$-2i\omega_{j}(A_{k}' + C_{k}A_{k}) - A_{j} = 0$$
 (5.14b)

Writing A_j and A_k as

$$A_j = B_j \exp(\lambda T_1)$$
 and $A_k = B_k \exp(\lambda T_1)$ (5.15)

and substituting into equation (5.15) into equations (5.14) leads to

$$2i\omega_{j}(\lambda + C_{j})B_{j} - \Lambda B_{jk}B_{k} = 0 \qquad (5.16a)$$

$$B_{j}' + 2i\omega_{j}(\lambda + C_{k})B_{k} = 0$$
 (5.16b)

Setting the determinant of the coefficients equal to zero, we obtain

$$2i\omega_{j}(\lambda + C_{j}) - \Lambda \beta_{jk} = 0 , (5.17a)$$

$$1 2i\omega_{j}(\lambda + C_{k})$$

or

$$\lambda^{2} + (C_{j} + C_{k})\lambda + (C_{j}C_{k} - \frac{\Lambda\beta_{jk}}{4\omega_{j}^{2}}) = 0$$
 (5.17b)

Hence

$$\lambda = -\frac{C_{j} + C_{k}}{2} \pm \frac{1}{2} \sqrt{(C_{j} - C_{k})^{2} + \frac{\Lambda\beta_{jk}}{\omega_{j}^{2}}}$$
(5.17c)

From (5.17c) we see that the solution decays if

$$(C_{j} + C_{k})^{2} > (C_{j} - C_{k})^{2} + \frac{\Lambda \beta_{jk}}{\omega_{j}^{2}}$$
 (5.18a)

$$\Lambda > 4\omega_j^2 C_j C_k / \beta_{jk}$$
(5.18b)

43

is the condition for flutter. Thus in the absence of damping, any value of Λ will cause flutter. Here we should note that β_{jk} is a positive aerodynamic coefficient that is associated with the so-called "pistontheory" approximation. Then Λ is a detuning parameter that is associated with the flutter speed; Λ is zero when the airspeed equals the flutter speed for the undamped system. When Λ is greater than zero, the airspeed is above the flutter speed. A more detailed explanation will be given in the numerical example of Chapter 9.

B. The Nonlinear Response - Limit Cycle Behavior

To examine the effect of the nonlinearity on the response, we no longer use equation (5.11). Again substituting equations (5.5) thru (5.10) into equation (5.4a) and equating coefficients of $e^{n\delta_0}$, we obtain

$$D_0^2 U_{i0} + \omega_i^2 U_{i0} = 0$$
 (5.19a)

$$D_0 U_{k^0} + \omega_j^2 U_{k^0} = 0$$
 (5.19b)

$$D_{0}^{2}U_{j^{1}} + \omega_{j}^{2}U_{j^{1}} = -2D_{0}D_{1}U_{j^{0}} - 2C_{j}D_{0}U_{j^{0}} + \Gamma_{jkkk}U_{k^{0}}^{3} + \Lambda B_{jk}U_{k^{0}}$$
(5.19c)

$$D_{0}^{2}U_{k^{1}} + \omega_{j}^{2}U_{k^{1}} = -2D_{0}D_{1}U_{k^{0}} - 2C_{k}D_{0}U_{k^{0}} + U_{j^{0}}$$
(5.19d)

The solution of equations (5.19a) and (5.19b) is equations (5.13). Substituting equations (5.13) into equations (5.19c) and (5.19d), introducing the polar form (3.1), and separating the result into real

or

and imaginary parts, we obtain the following conditions for the elimination of secular terms:

$$\omega_j(a_j + C_j a_j) + (\frac{3}{8} \Gamma_{jkkk} a_k^3 + \frac{\Lambda \beta_{jk}}{2} a_k) \sin \gamma = 0$$
 (5.20a)

$$\omega_j(a'_k + C_k a_k) + \frac{1}{2} a_j \sin \gamma = 0$$
 (5.20b)

$$\alpha_{j}^{i}a_{j}\omega_{j} + \left(\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k}\right)\cos\gamma = 0 \qquad (5.20c)$$

$$\alpha_{k}^{i}a_{k}\omega_{j} - \frac{1}{2}a_{j}\cos\gamma = 0 \qquad (5.20d)$$

where

$$\gamma = \alpha_{j} - \alpha_{k} \tag{5.21a}$$

For steady-state solutions $a'_j = a'_k = 0$. We see immediately that $a_j = a_k = 0$ are possible solutions. For nontrivial solutions, $a_j \neq a_k \neq 0$, it follows that $\gamma' = 0$ then

$$\alpha'_{j} = \alpha'_{k} \tag{5.21b}$$

The equations governing the steady-state response then become

$$\omega_{j}C_{j}a_{j} + (\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k})sin\gamma = 0$$
 (5.22a)

$$\alpha_{j}^{i}a_{j}\omega_{j} + \left(\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k}\right)\cos\gamma = 0 \qquad (5.22b)$$

$$\omega_j C_k a_k + \frac{a_j \sin\gamma}{2} = 0 \qquad (5.22c)$$

$$\alpha_{k}^{\dagger}a_{k}\omega_{j} - \frac{a_{j}\cos\gamma}{2} = 0 \qquad (5.22d)$$

We now multiply (5.22a) by $C_k a_k$ and (5.22c) by $C_j a_j$, subtract, and obtain

$$\left[\left(\frac{3}{8}\Gamma_{jkkk}a_{k}^{2}+\frac{\Lambda\beta_{jk}}{2}\right)C_{k}a_{k}^{2}-\frac{C_{j}a_{j}^{2}}{2}\right]\sin\gamma=0$$
 (5.23a)

Similarly we multiply (5.22b) by a_k and (5.22d) by a_j , subtract, and obtain

$$\left[\left(\frac{3}{8}\Gamma_{jkkk}a_{k}^{2}+\frac{\Lambda\beta_{jk}}{2}\right)a_{k}^{2}+\frac{a_{j}^{2}}{2}\right]\cos\gamma=0$$
 (5.23b)

The coefficient of siny in equation (5.23a) and the coefficient of $\cos\gamma$ in equation (5.23b) cannot be zero simultaneously. Moreover, siny and $\cos\gamma$ cannot be zero simultaneously. It follows from equation (5.22c) that

$$\sin\gamma \neq 0$$
 (5.24a)

Thus, we must have

$$\cos\gamma = 0 \tag{5.24b}$$

and

$$(\frac{3}{4}\Gamma_{jkkk}a_{k}^{2} + \Lambda\beta_{jk})C_{k}a_{k}^{2} - C_{j}a_{j}^{2} = 0$$
 (5.24c)

If cosy = 0, then from (5.22c) it follows that

 $sin\gamma = -1$ (5.24d)

Substituting (5.24d) into (5.22c) leads to

$$a_j = 2\omega_j C_k a_k \tag{5.25a}$$

Then substituting (5.25a) into (5.24c) leads to

$$a_{k} = 2 \sqrt{\frac{4\omega_{j}^{2}C_{j}C_{k} - \Lambda\beta_{jk}}{3\Gamma_{jkkk}}}$$
(5.25b)

If Γ_{jkkk} is negative (this is the case in the numerical example considered in Chapter 9), the condition for the existence of a real solution is

$$\Lambda > 4\omega_j^2 C_j C_k / \beta_{jk}$$
(5.25c)

in full agreement with equation (5.18b), which predicts the onset of flutter. The linear analysis merely predicts growth, while the nonlinear analysis predicts the development of a finite-amplitude limit cycle. The existence of limit cycles was found numerically, but not analytically, by Dowell (1966,1970). Smith and Morino (1976) developed limit cycles analytically but not in the presence of parametric excitation.

5.2 Summary

The analysis of linear and nonlinear systems having one repeated frequency has been carried out in the absence of either an internal or a parametric resonance. The amplitudes of all the modes not associated with the repeated frequency are shown to decay. For the two modes associated with the repeated frequency, the linear analysis predicts an infinitely growing solution if the condition

$$\Lambda > 4\omega_j^2 C_j C_k / \beta_{jk}$$
(5.26)

is met. The nonlinear analysis requires that the same condition be met if nontrivial solutions are to exist. However, the nonlinearity puts a bound on the response and limit-cycle behavior is found.

Now we can write a complete generalization of the solvability condition of equation (2.9) that is valid for the cases where either distinct or repeated frequencies exist. It has the form

$$-2i\omega_{n}(D_{1}A_{n} + C_{n}A_{n}) + A_{n} \sum_{p=1}^{\infty} \gamma_{np}A_{p}\overline{A}_{p} + I_{n} + E_{n} + K_{n} = 0$$
(5.27a)

where

$$K_{n} = \begin{bmatrix} \sum_{p=1}^{\infty} \Lambda \beta_{np} A_{p} & \text{if } n \neq j. \ m \neq k \\ \sum_{p=1}^{\infty} \Lambda \beta_{jp} A_{p} + A_{k} \sum_{p=1}^{\infty} \gamma_{jp} A_{p} \overline{A}_{p} & \text{if } n = j \\ \sum_{p=1}^{\infty} \Lambda \beta_{kp} A_{p} + A_{j} \sum_{p=1}^{\infty} \gamma_{kp} A_{p} \overline{A}_{p} - A_{j} & \text{if } n = k \end{bmatrix}$$
(5.27b)
and
$$\begin{bmatrix} 3\Gamma_{jkkk} & \text{if } p = k \end{bmatrix}$$

$$\gamma_{jp} = \begin{bmatrix} 3\Gamma_{jkkk} & \text{if } p = k \\ \\ 2(\Gamma_{jkpp} + \Gamma_{jpkp} + \Gamma_{jppk} & \text{if } p \neq k \end{bmatrix} (5.27c)$$

)

and

$$\gamma_{kp} = \begin{bmatrix} 3\Gamma_{kjjj} & \text{if } p = j \\ \\ 2(\Gamma_{kjpp} + \Gamma_{kpjp} + \Gamma_{kppj} & \text{if } p \neq j \end{bmatrix} (5.27d)$$

It should be noted that for the case of distinct frequencies the ${\rm K}_{\rm n}$ term is set equal to zero and equation (5.31a) becomes identically equation (2.9). When repeated frequencies are considered and appropriate scaling parameters, δ_i , are specified by equations (5.5) through (5.11), then we note that in equation (5.27a) the term

$$A_n \sum_{p=1}^{\infty} \gamma_{np} A_p \overline{A}_p$$

is a higher-order term and hence does not enter the analysis at this level of approximation. In addition equation (5.27b) simplifies to

$$\kappa_{n} = \begin{bmatrix} 0 & \text{if } n \neq j \text{ or } k \\ \Lambda \beta_{jk} A_{k} + 3\Gamma_{jkkk} A_{k}^{2} \overline{A}_{k} & \text{if } n = j \\ -A_{j} & \text{if } n = k \end{bmatrix} (5.27e)$$

CHAPTER SIX

PARAMETRIC RESONANCE IN THE ABSENCE OF INTERNAL RESONANCE

In this chapter we investigate several cases of parametric resonance in the absence of an internal resonance. We demonstrate that using a two-term expansion used in the method of multiple scales is inadequate and a three-term expansion is required for the combination resonance.

We recall from Chapter 5 that the modes not involved in an internal or a parametric resonance decay as $t \rightarrow \infty$. Here the emphasis is on steady-state solution; thus, we do not consider these modes further.

6.1 The Case Where $\lambda \stackrel{\sim}{\sim} 2\omega_j$.

The resonant condition associated with this case is

$$\lambda = 2\omega_i + \varepsilon^{\delta} \rho \tag{6.1}$$

where δ is a constant to be determined and ρ is a detuning parameter. Based on equations (5.27a) and (5.27e), we see that the solvability conditions for the elimination of secular terms have the form

$$-2i\omega_{j}(A_{j}^{\prime}+C_{j}A_{j})+I_{j}+E_{j}+3\Gamma_{j}kkk^{A^{2}}k^{\overline{A}}k+\Lambda\beta_{j}k^{A}k$$
 (6.2a)

$$-2i\omega_{j}(A'_{k} + C_{k}A_{k}) + I_{k} + E_{k} - A_{j} = 0$$
 (6.2b)

where I_n and E_n are terms resulting from internal and parametric resonances, respectively. In the last chapter, we analyzed these equations with $I_n = E_n = 0$. To examine the effect of the above resonant combination, we must include the $f_{jk}U_{k0}$ term from equation (5.4a) with n = j. We keep $E_k = 0$ and $I_j = I_k = 0$. The condition for inclusion of this parametric resonant term is

$$\delta_{\mathbf{F}} + \delta_{\mathbf{i}} - \delta_{\mathbf{k}} = \delta_{\mathbf{0}} \tag{6.3a}$$

Substituting equations (5.5) and (5.6b) into equation (6.3a) we have

$$\delta_{\mathbf{F}} = 2\delta_0 \tag{6.3b}$$

Hence the parametric resonant term becomes

$$E_{j} = -f_{jk}\overline{A}_{k} \exp[(i\varepsilon^{\delta_{0}}\rho)T_{0}]$$
 (6.4a)

Because the interaction occurs in the second terms of the expansion, we put

$$\delta = \delta_0 \tag{6.4b}$$

so that using equation (5.2b) we obtain

$$E_{j} = -f_{jk}\overline{A}_{k}exp(i\rho T_{1})$$
 (6.4c)

The solvability conditions for the elimination of secular terms become

$$-i2\omega_{j}(A'_{j} + C_{j}A_{j}) - f_{jk}\overline{A}_{k}exp(i\rho T_{1}) + 3\Gamma_{jkkk}A^{2}_{k}\overline{A}_{k} + \Lambda\beta_{jk}A_{k} = 0$$
(6.5a)

$$- i2\omega_{j}(A_{k}' + C_{k}A_{k}) - A_{j} = 0$$
 (6.5b)

Introducing the polar form (3.1) and separating the result into real and imaginary parts, we obtain

$$ω_j(a'_j + C_j a_j) + \frac{f_{jk}}{2} a_k \sin\beta + (\frac{3}{8} \Gamma_{jkkk} a^3_k + \Lambda \frac{\beta_{jk} a_k}{2}) \sin\gamma = 0$$
(6.6a)

$$\omega_{j}a_{j}\alpha_{j}^{\prime} - \frac{f_{jk}}{2}a_{k}\cos\beta + (\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \Lambda \frac{\beta_{jk}a_{k}}{2})\cos\gamma = 0 \quad (6.6b)$$

$$\omega_j(a'_k + C_k a_k) + \frac{a_j}{2} \sin \gamma = 0$$
 (6.6c)

$$\omega_j a_k \alpha'_k - \frac{a_j}{2} \cos \gamma = 0$$
 (6.6d)

where

$$\beta = \rho T_1 - \alpha_k - \alpha_j \tag{6.7a}$$

$$\gamma = \alpha_{j} - \alpha_{k} \tag{6.7b}$$

For steady-state conditions, $a'_j = a'_k = 0$. Either both a_j and a_k are zero, or neither is zero. For nontrivial solutions,

$$\beta' = \rho - \alpha'_j - \alpha'_k = 0 \qquad (6.8a)$$

and '

$$\gamma' = \alpha'_j - \alpha'_k = 0 \tag{6.8b}$$

which yield

$$\alpha'_{j} = \alpha'_{k} = \frac{\rho}{2} \tag{6.9}$$

The steady-state equations become

$$ω_j C_{ja_j} + \frac{f_{jk}}{2} a_k sin_\beta + (\frac{3}{8} \Gamma_{jkkk} a_k^3 + \Lambda \frac{\beta_{jk} a_k}{2}) sin_\gamma = 0$$
 (6.10a)
a.

$$\omega_j C_k a_k + \frac{a_j}{2} \sin\gamma = 0 \tag{6.10b}$$

$$\omega_{j}a_{j}\frac{\rho}{2} - \frac{f_{jk}}{2}a_{k}\cos\beta + (\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \Lambda \frac{\beta_{jk}a_{k}}{2})\cos\gamma = 0$$
(6.10c)

$$\omega_j a_k \rho - a_j \cos \gamma = 0 \tag{6.10d}$$

It is possible to combine equations (6.10b) and (6.10d) to obtain

$$a_j^2 = \omega_j^2 a_k^2 (\rho^2 + 4C_k^2)$$
 (6.11a)

Equations (6.10b) and (6.10d) may be substituted into (6.10a) and (6.10c), respectively, so that, upon solving for sin β and cos β and squaring the results, we can use (6.11a) to obtain a closed-form solution:

$$a_{k}^{2} = \frac{4}{3\Gamma_{jkkk}} \left[-\Lambda\beta_{jk} + \omega_{j}^{2} (4C_{j}C_{k} - \rho^{2}) \pm \sqrt{f_{jk}^{2} - 4\omega_{j}^{4}\rho^{2}(C_{j} + C_{k})^{2}} \right]$$
(6.11b)

We note that, if a nontrivial solution exists as $f_{jk} \neq 0$, then $\rho \neq 0$ and hence equation (6.11b) reduces to

$$a_{k} = 2 \sqrt{\frac{4\omega_{j}^{2}C_{j}C_{k} - \Lambda\beta_{jk}}{3\Gamma_{jkkk}}}$$
(6.11c)

and (6.11a) reduces to

$$a_j = 2\omega_j C_k a_k \tag{6.11d}$$

in full agreement with equation (5.25), which gives the amplitudes of the limit cycle for the unforced response.

More discussion of the results can be found in Chapter 9 where a numerical example is presented.

6.2 The Case Where $\lambda \approx \omega_r + \omega_j$.

The resonant combination associated with this case is

$$\lambda = \omega_{\rm p} + \omega_{\rm j} + \varepsilon^{\delta} \rho \tag{6.12}$$

To determine the effect of the resonant combination, we must include the following terms in (5.4a): $f_{jr}U_{r^0}$ when n = j and $f_{rk}U_{k^0}$ when n = r. We now have three equations to analyze since the r^{th} -mode has been drawn into the response by the resonant combination of eouation (6.12). If the scaling conditions specified in equations (5.5) thru (5.10) are met along with equation (6.3b), we see that the resonant term of the form $f_{jn}U_{r^0}$ in equation (5.4a), for n = j, will be of higher order and will not enter the two-term expansion. Satisfying equation (5.10) for $\delta_r = \delta_0$ allows for inclusion of the $f_{jr}U_{r^0}$ term; however the term $f_{rk}U_{r^0}$ in equation (5.4a), for n = r, will be of higher order and will not enter the two-term expansion. In this case the solution predicts that A_r decays as $t + \infty$. Thus it appears that the original two-term expansion is an inadequate approximation, and recourse is made to a three-term expansion.

We re-examine equation (5.4a) and proceed with an analysis which parallels that of Chapter 5.

We arbitrarily choose the scaling coefficients, ε^{δ_0} , so that the effects of damping, the repeated frequency, the aerodynamic loading, the nonlinearity, and the parametric resonance all interact at the same order of approximation. We have already determined that this will not occur at $O(\varepsilon^{\delta_0})$, so we expect that a uniformly valid solution will exist at $O(\varepsilon^{2\delta_0})$. Thus the δ in equation (6.12) is set at

 $\delta = 2\delta_0 \tag{6.13a}$

For a reference, we again use the j-th mode and hence let

$$\delta_j = 0 \tag{6.13b}$$

In order to include the effect of damping, now we must let

$$\delta_c = 2\delta_0 \tag{6.13c}$$

Accounting for the effect of the repeated frequency, we must let

$$\delta_k - \delta_i = 2\delta_0$$

Using (6.13a) we obtain

$$\delta_{\mathbf{k}} = 2\delta_0 \tag{6.13d}$$

To bring the aerodynamic loading into the approximation, we set

 $\delta_{\Lambda} + \delta_n - \delta_m = 2\delta_0$,

where we note that for n = j and m = k

$$\delta_{\Lambda} = 4\delta_0 \tag{6.13e}$$

and for n = k and m = j

$$\delta_{\Lambda} = 0$$

Following the reasoning of Chapter 5, we conclude that equation (6.13e) must be satisfied to produce acceptable results.

To account for the nonlinearity, we set

 $\delta_{\rm N} + \delta_{\rm n} - \delta_{\rm m} - \delta_{\rm p} - \delta_{\rm q} = 2\delta_0$

When n = j and m = p = q = k, this leads to

$$\delta_{N} = 8\delta_{0} \tag{6.13f}$$

Other combinations of n, m, p, and q lead to different values for δ_N , but all these cases lead to inconsistencies in the expansion.

Finally, to include the effect of the parametric resonance, we set

$$\delta_F + \delta_n - \delta_m = 2\delta_0$$

Setting

$$\delta_{\mathbf{r}} = 3\delta_0 \tag{6.13g}$$

allows for acceptable solutions. If n = j and m = r, then equation (6.13g) yields

$$\delta_{\mu} = \delta_0 \tag{6.13h}$$

and one of the parametric resonance terms $f_{jr}U_{r^0}$ can be accomodated. When n = r and m = k the other resonant terms $f_{rk}U_{k^0}$ is also included. Substituting equations (6.13) into equation (5.4a) and equating coefficients of equal powers of $\varepsilon^{n\delta_0}$, we obtain

$$D_0^2 U_{i^0} + \omega_i^2 U_{i^0} = 0 \tag{6.14a}$$

 $D_{0}^{2}U_{k^{0}} + \omega_{j}^{2}U_{k^{0}} = 0$ (6.14b)

$$D_0^2 U_{r^0} + \omega_r^2 U_{r^0} = 0 \tag{6.14c}$$

$$D_{0}^{2}U_{j^{1}} + \omega_{j}^{2}U_{j^{1}} + 2D_{0}D_{1}U_{j^{0}} + 2f_{jk}U_{k^{0}}\cos\lambda T_{0} = 0 \qquad (6.15a)$$

$$D_{0}^{2}U_{k^{1}} + \omega_{j}^{2}U_{k^{1}} + 2D_{0}D_{1}U_{k^{0}} = 0$$
 (6.15b)

$$D_{\theta}^{2}U_{r1} + \omega_{r}^{2}U_{r1} + 2D_{0}D_{1}U_{r0} = 0$$
 (6.15c)

$$D_{0}^{2}U_{j^{2}} + \omega_{j}^{2}U_{j^{2}} + 2D_{0}D_{2}U_{j^{0}} + D_{1}^{2}U_{j^{0}} + 2D_{0}D_{1}U_{j^{1}} + 2C_{j}D_{0}U_{j^{0}}$$

$$+ 2f_{jr}U_{r^{0}}\cos\lambda T_{0} + 2f_{jk}U_{k^{1}}\cos\lambda T_{0} - \Gamma_{jkkk}U_{k^{0}}^{3}$$

$$- \Lambda \beta_{jk}U_{k^{0}} = 0 \qquad (6.16a)$$

$$D_{0}^{2}U_{k^{2}} + \omega_{j}^{2}U_{k^{2}} + 2D_{0}D_{2}U_{k^{0}} + D_{1}^{2}U_{k^{0}} + 2D_{0}D_{1}U_{k^{1}} + 2C_{k}D_{0}U_{k^{0}}$$

$$+ U_{j^{0}} = 0 \qquad (6.16b)$$

$$D_{0}^{2}U_{r^{2}} + \omega_{r}^{2}U_{r^{2}} + 2D_{0}D_{2}U_{r^{0}} + D_{1}^{2}U_{r^{0}} + 2D_{0}D_{1}U_{r^{1}} + 2C_{r}D_{0}U_{r^{0}}$$

+ $2f_{rk}U_{k^0}\cos\lambda T_0 = 0$ (6.16c)

The solutions of equations (6.14) are

$$U_{j^0} = A_j(T_1, T_2) \exp(i\omega_j T_0) + cc$$
 (6.17a)

$$U_{k^0} = A_k(T_1, T_2) \exp(i\omega_i T_0) + cc$$
 (6.17b)

$$U_{r^{0}} = A_{r}(T_{1}, T_{2}) \exp(i\omega_{r}T_{0}) + cc \qquad (6.17c)$$

Substituting equations (6.17) into equations (6.15), we find that secular terms are eliminated provided

$$D_1A_i = D_1A_k = D_1A_r = 0$$
 (6.18)

and that the \boldsymbol{A}_n are independent of \boldsymbol{T}_1 .

It follows that

$$U_{j^{1}} = f_{jk} \left[\frac{A_{k} \exp[i(\lambda + \omega_{j})T_{0}]}{\lambda(\lambda + 2\omega_{j})} + \frac{\overline{A}_{k} \exp[i(\lambda - \omega_{j})T_{0}]}{\lambda(\lambda - 2\omega_{j})} \right] + cc$$
(6.19a)

$$U_{k^{1}} = U_{r^{1}} = 0 \tag{6.19b}$$

Substituting (6.17) through (6.19) into (6.16) yields the solvability conditions

$$-2i\omega_{j}(A'_{j} + C_{j}A_{j}) - f_{jr}\overline{A}_{r}exp(i\rho T_{2}) + 3\Gamma_{jkkk}A^{2}_{k}\overline{A}_{k} + \Lambda\beta_{jk}A_{k} = 0$$
(6.20a)

$$-2i\omega_{j}(A_{k}^{\prime}+C_{k}A_{k})-A_{j}=0$$
 (6.20b)

$$-2i\omega_{r}(A'_{r} + C_{r}A_{r}) - f_{rk}\overline{A}_{k}exp(i\rho T_{2}) = 0$$
 (6.20c)

where primes denote differentiation with respect to T_2 . Letting

$$A_j = \frac{1}{2} a_n(T_2) \exp[i\alpha_n(T_2)]$$
 $n = j, k, r$ (6.21)

substituting equation (6.21) into equations (6.20) and separating the result into real and imaginary parts, we obtain

$$\omega_{j}(a_{j}^{\prime} + C_{j}a_{j}) + \frac{f_{jr}}{2} a_{r}^{sin\beta_{1}} + (\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k}^{\prime})sin\gamma = 0$$
(6.22a)

$$\omega_{j}a_{j}\alpha_{j}^{\prime} - \frac{f_{jr}}{2}a_{r}\cos\beta_{1} + (\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k})\cos\gamma = 0 \quad (6.22b)$$

$$\omega_j(a_k' + C_k a_k) + \frac{a_j}{2} \sin\gamma = 0$$
 (6.22c)

$$\omega_{j}a_{k}\alpha_{k}' - \frac{a_{j}}{2}\cos\gamma = 0 \qquad (6.22d)$$

$$\omega_r(a'_r + C_r a_r) + \frac{f_{rk}}{2} a_k \sin \beta_2 = 0$$
 (6.22e)

$$\omega_{r}a_{r}\alpha_{r}' - \frac{f_{rk}}{2}a_{k}\cos\beta_{2} = 0 \qquad (6.22f)$$

where

$$\beta_1 = \rho T_2 - \alpha_r - \alpha_j \qquad (6.23a)$$

$$\gamma = \alpha_j - \alpha_k \tag{6.23b}$$

$$\beta_2 = \rho T_2 - \alpha_r - \alpha_k \qquad (6.23c)$$

For the steady-state response, $a'_j = a'_k = a'_r = 0$ and $\beta'_1 = \gamma' = \beta'_2 = 0$. Thus

$$\alpha'_j = \alpha'_k \tag{6.24a}$$

$$\alpha'_{r} + \alpha'_{j} = \alpha'_{r} + \alpha'_{k} = \rho \qquad (6.24b)$$

$$\omega_{j}C_{j}a_{j} + \frac{f_{jr}}{2}a_{r}\sin\beta_{1} + \left[\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k}\right]\sin\gamma = 0 \quad (6.25a)$$

$$\omega_j C_k a_k + \frac{a_j}{2} \sin\gamma = 0 \qquad (6.25b)$$

$$\omega_{r} C_{r} a_{r} + \frac{\tau_{rk}}{2} a_{k} \sin\beta_{2} = 0 \qquad (6.25c)$$

$$\rho = \frac{a_j}{2\omega_j a_k} \cos\gamma + \frac{f_r k^a k}{2\omega_r a_r} \cos\beta_2 \qquad (6.25d)$$

$$\rho = \frac{1}{\omega_{j}a_{j}} \left\{ \frac{f_{jr}}{2} a_{r} \cos\beta_{1} - \left[\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2} a_{k}\right]\cos\gamma \right\}$$

$$+ \frac{f_{rk}a_{k}}{2\omega_{r}a_{r}}\cos\beta_{2} \qquad (6.25e)$$

$$a_{j}^{2}\cos\gamma = 2a_{k} \left\{ \frac{f_{jr}}{2} a_{r}\cos\beta_{1} - \left[\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2} a_{k}\right]\cos\gamma \right\}$$

J_(6.25f)

From (6.22) we see that in the steady-state if any one of the $a_n = 0$ the other two are also zero.

r

These equations do not readily admit a closed-form solution. They are discussed further in Chapter 9, where a numerical example is presented. We note that only five equations are independent, equation (6.25f) being a combination of equations (6.25e) and (6.25d). A sixth equation can be obtained from equations (6.23):

$$\beta_2 = \beta_1 + \gamma \tag{6.25g}$$

6.3 The Case Where $\lambda \stackrel{\sim}{\sim} \omega_r - \omega_j$.

In this case the ordering scheme turns out to be the same as that of the previous section. Hence, we write the resonant combination as

$$\lambda = \omega_{\rm r} - \omega_{\rm j} + \varepsilon^{2\delta_0} \rho \tag{6.26}$$

The resonant terms become

$$E_{j}exp(i\omega_{j}T_{0}) = f_{jr}A_{r}exp[i(\omega_{j} - \varepsilon^{2\delta_{0}}\rho)T_{0}] \qquad (6.27a)$$

and

$$E_{r} \exp(i\omega_{r} T_{0}) = f_{rk} A_{k} \exp[i(\omega_{r} + \varepsilon^{2\delta_{0}}\rho)T_{0}]$$
(6.27b)

with

$$\beta_1 = -\rho T_2 + \alpha_p - \alpha_j \qquad (6.28a)$$

$$\gamma = \alpha_i - \alpha_k \tag{6.28b}$$

$$\beta_2 = \rho T_2 - \alpha_r + \alpha_k \tag{6.28c}$$

The governing equations have the same form as those of the last section. Hence the steady-state equations become

.
$$\omega_{j}C_{j}a_{j} + \frac{f_{j}r}{2}a_{r}\sin\beta_{1} + \left[\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k}\right]\sin\gamma = 0 \quad (6.29a)$$

$$\omega_j C_j a_j + \frac{1}{2^k} \sin \gamma = 0$$
 (6.29b)

$$\omega_r C_r a_r + \frac{rk}{2} a_k \sin\beta_2 = 0 \qquad (6.29c)$$

$$\rho = \frac{f_{rk}a_{k}}{2\omega_{r}a_{r}}\cos\beta_{2} + \frac{a_{j}}{2\omega_{j}a_{k}}\cos\gamma \qquad (6.29d)$$

$$\rho = \frac{f_{rk}a_{k}}{2\omega_{r}a_{r}} \cos\beta_{2} + \frac{1}{\omega_{j}a_{j}} \left\{ \frac{f_{jr}}{2} a_{r}\cos\beta_{1} - \left[\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k}\right]\cos\gamma \right\}$$

$$(6.29e)$$

$$\frac{a_{j}\cos\gamma}{2\omega_{j}a_{k}} = \frac{1}{\omega_{j}a_{j}} \left\{ \frac{f_{jr}}{2} a_{r}\cos\beta_{1} - \left[\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k}\right]\cos\gamma \right\}$$

$$(6.29e)$$

6.4 Summary

Several cases in which parametric resonances are involved have been investigated. The resonance case $\lambda \stackrel{\sim}{\sim} 2\omega_j$ appears at first order, while the resonance cases $\lambda \stackrel{\sim}{\sim} \omega_r \pm \omega_j$ appear at second order. Nothing conclusive is self-evident from a cursury examination of the steady-state equations. Analysis and discussion await the coverage of numerical examples in Chapter 9.

CHAPTER SEVEN

THE EFFECT OF INTERNAL RESONANCE

Again we choose to investigate an internal resonance of the form $\omega_r \approx 3\omega_j$. In this chapter we examine parametric resonances in the presence of this internal resonance. We also discuss the case where the internal resonance is present but there is no parametric resonance, the only excitation coming from the aerodynamic loading.

7.1 The Case Where $\lambda \stackrel{\sim}{\sim} 2\omega_j$ and $\omega_r \stackrel{\sim}{\sim} 3\omega_j$.

The resonances associated with this case are

$$\lambda = 2\omega_{i} + \varepsilon^{\delta}\rho \qquad (7.1a)$$

and

$$\omega_{r} = 3\omega_{j} + \varepsilon^{\delta}\sigma \tag{7.1b}$$

In this case, the general form of the solvability conditions is as follows:

$$- 2i\omega_{j}(A'_{j} + C_{j}A_{j}) + I_{j} + E_{j} + 3\Gamma_{jkkk}A_{k}\overline{A}_{k} + \Lambda B_{jk}A_{k} = 0 \quad (7.2a)$$
$$- 2i\omega_{j}(A'_{k} + C_{k}A_{k}) + I_{k} + E_{k} - A_{j} = 0 \quad (7.2b)$$

$$-2i\omega_{r}(A'_{r} + C_{r}A_{r}) + I_{r} + E_{r} = 0$$
 (7.2c)

To obtain the best approximation, we need to develop an ordering scheme which includes the effect of the repeated frequency, the parametric resonance, the internal resonance, and as many other terms as it is consistent to retain. Proceeding as in Section 6.1, we find that equations (5.5) thru (5.10) and (6.3b) still hold and that

$$\delta_r = 0 \tag{7.3a}$$

Thus, the nonzero resonant terms in the solvability conditions become

$$E_{j} = -f_{jk}\overline{A}_{k}exp(i\rho T_{1})$$
(7.3b)

$$\mathbf{E}_{\mathbf{r}} = -\mathbf{f}_{\mathbf{r}\mathbf{k}}\mathbf{A}_{\mathbf{k}}\exp[\mathbf{i}(\rho - \sigma)\mathbf{T}_{1}]$$
(7.3c)

and

$$I_r = \Gamma_{rkkk} A_k^{3} \exp(-i\sigma T_1)$$
(7.3d)

Substituting equations (7.3b) - (7.3d) into equations (7.2), introducing the polar form (3.1), and separating the result into real and imaginary parts, we obtain

$$\omega_{j}(a_{j}' + C_{j}a_{j}) + \frac{f_{jk}a_{k}}{2} \sin\beta_{1} + (\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k})\sin\gamma = 0$$
(7.4a)
$$\omega_{j}a_{j}\alpha_{j}' - \frac{f_{jk}a_{k}}{2}\cos\beta_{1} + (\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{jk}}{2}a_{k})\cos\gamma = 0$$
(7.4b)
$$a_{j}$$

$$\omega_j(a_k' + C_k a_k) + \frac{\alpha_j}{2} \sin \gamma = 0$$
 (7.4c)

$$\omega_{j}a_{k}\alpha_{k}' - \frac{a_{j}}{2}\cos\gamma = 0 \qquad (7.4d)$$

$$\omega_{r}(a_{r}^{\prime} + C_{r}a_{r}) - \frac{1}{8}\Gamma_{rkkk}a_{k}^{3}\sin\mu + \frac{1}{2}f_{rk}a_{k}\sin\beta_{2} = 0 \qquad (7.4e)$$

$$\omega_{r}a_{r}\alpha_{r}' + \frac{1}{8}\Gamma_{rkkk}a_{k}^{3}\cos\beta_{2} - \frac{1}{2}f_{rk}a_{k}\cos\beta_{2} = 0$$
 (7.4f)

where

$$\beta_1 = \rho T_1 - \alpha_k - \alpha_j \tag{7.4g}$$

$$\gamma = \alpha_j - \alpha_k \tag{7.4h}$$

$$\mu = -\sigma T_1 + 3\alpha_k - \alpha_r \tag{7.41}$$

$$\beta_2 = (\rho - \sigma)T_1 + \alpha_k - \alpha_r = \beta_1 + \gamma + \mu \qquad (7.4j)$$

For the steady-state solutions $a'_j = a'_k = a'_r = 0$, and it follows that $a_j = a_k = a_r = 0$ are possible solutions. For non-trivial solutions

$$\beta_1' = \beta_2' = \mu' = \gamma' = 0$$
 (7.5a)

so that

$$\rho = \alpha'_k + \alpha'_i \tag{7.5b}$$

$$\alpha'_{j} = \alpha'_{k} \tag{7.5c}$$

and

 $\sigma = 3\alpha'_k - \alpha'_r \tag{7.5d}$

from which

$$\alpha'_{j} = \alpha'_{k} = \frac{\rho}{2} \tag{7.5e}$$

and

$$\alpha'_{r} = \frac{3\rho}{2} - \sigma \tag{7.5f}$$

The steady-state equations become

$$\omega_{j}C_{j}a_{j} + \frac{f_{j}k^{a}k}{2} \sin\beta_{1} + (\frac{3}{8}\Gamma_{jkkk}a_{k}^{3} + \frac{\Lambda\beta_{j}k^{a}k}{2})\sin\gamma = 0 (7.6a)$$

$$a_{j}$$

$$\omega_j C_k a_k + \frac{a_j}{2} \sin\gamma = 0 \tag{7.6b}$$

$$\omega_r c_r^a - \frac{1}{8} \Gamma_{rkkk} a_k^3 \sin \mu + \frac{f_{rk} a_k \sin \beta_2}{2} = 0 \qquad (7.6c)$$

$$\omega_j a_j \frac{\rho}{2} - \frac{f_j k^a k}{2} \cos \beta_1 + (\frac{3}{8} \Gamma_{jkkk} a_k^3 + \frac{\Lambda \beta_j k^a k}{2}) \cos \gamma = 0$$
 (7.6d)

$$\omega_{j}a_{k}\rho - a_{j}\cos\gamma = 0 \qquad (7.6e)$$

$$\omega_{r}a_{r}(\frac{3\rho}{2} - \sigma) + \frac{1}{8}\Gamma_{rkkk}a_{k}^{3}cos\mu - \frac{rk^{2}k^{2}cos\mu}{2} = 0 \qquad (7.6f)$$

Equations (7.6a, b, d, e) are identical to equations (6.10) and hence can be reduced to

$$a_{k}^{2} = \frac{4}{3\Gamma_{jkkk}} \left[-\Lambda\beta_{jk} + \omega_{j}^{2} (4C_{k}C_{j} - \rho^{2}) \pm \sqrt{f_{jk}^{2} - 4\omega_{j}^{4}\rho^{2}(C_{j} + C_{k})^{2}} \right]$$
(7.6g)

$$a_j = \omega_j a_k \sqrt{\rho^2 + 4C_k^2}$$
 (7.6h)

Moreover, β_1 and γ can be obtained from these equations. To find a_r , β_2 and μ , we solve equations (7.6c,f) and (7.4j) numerically. The response is the same as that in Chapter 6 for the two modes associated with the repeated frequency. Due to the presence of an internal resonance, a third mode is drawn into the response, but it does not influence the j-th and k-th modes.

7.2 The Case Where $\lambda \approx \omega_r + \omega_j$ and $\omega_r \approx 3\omega_j$.

The resonances associated with this case are

$$\lambda = \omega_r + \omega_j + \varepsilon^{\delta} \rho \tag{7.7a}$$

and

$$\omega_{r} = 3\omega_{j} + \varepsilon^{\delta}\sigma \qquad (7.7b)$$

The solvability conditions of equations (7.2) remain valid. Starting with the formulation of the three-term expansion of Section 6.2, we note

that the parametric resonances have been accounted for and that we must attempt to include the internal resonance terms

$$I_{j} = (\Gamma_{jrkk} + \Gamma_{jkrk} + \Gamma_{jkkr})A_{r}\overline{A_{k}^{2}} \exp(i\varepsilon^{\delta}\sigma T_{0})$$
(7.8a)

and

$$I_{r} = \Gamma_{rkkk} A_{k}^{3} \exp(-i\epsilon^{\delta} \sigma T_{0})$$
 (7.8b)

Unfortunately, both of these terms are order $e^{3\delta_0}$ and do not enter at this level of approximation.

We thus conclude that the effects of the parametric resonance and the repeated frequency dominate the response and that the effect of the internal resonance is of higher order. To this order of approximation the response in the presence of a parametric resonance is the same regardless of whether an internal resonance is present or not.

7.3 The Case of No Parametric Resonance with $\omega_r \approx 3\omega_j$.

Only an internal resonance is associated with this case and it is associated with the combination

$$\omega_{r} = 3\omega_{j} + \varepsilon^{\delta}\sigma \tag{7.9}$$

The solvability conditions of equations (7.2) remain valid. There is no parametric resonance so the $E_n = 0$. Using the scaling conditions of equations (5.5) thru (5.10), (6.3b) and (7.3a), we find that the only internal resonance term that appears to first order is

$$I_{r} = \Gamma_{rkkk} A_{k}^{3} \exp(-i\varepsilon^{\delta} \sigma T_{0})$$
 (7.10)

Substituting equation (7.10) into equation (7.2), introducing the polar form (3.1), and separating the result into real and imaginary parts, we obtain equations identical to equations (7.4) when

$$f_{jk} = f_{rk} = 0$$
 (7.11)

The resulting steady-state equations lead to the following:

$$a_{k} = 2 \sqrt{\frac{4\omega_{j}^{2}C_{j}C_{k} - \Lambda\beta_{jk}}{3\Gamma_{jkkk}}}$$
(7.12a)

$$a_j = 2\omega_j c_k a_k \tag{7.12b}$$

and

$$a_{r} = \frac{\frac{1}{8} \Gamma_{rkkk} a_{k}^{3}}{\sqrt{\omega_{r}^{2} (C_{r}^{2} + \sigma^{2})}}$$
(7.12c)

Hence we note that equations (7.12a) and (7.12b) are the limit-cycle equations developed in Chapter 5 when no resonance was present. The modes associated with the repeated frequency are independent of the internal resonance, which only serves to draw the r-th mode into the response. Moreover, the steady-state amplitudes a_j and a_k predicted by equations (7.6g) and (7.6h) approach those given by equations (7.12a) and (7.12b) as f_{jk} vanishes. We note that, if a steady-state solution exists, $|f_{jk}| > |\rho| 2\omega_j (C_j + C_k)$ and hence ρ must vanish as f_{jk} vanishes. Finally, we note that equations (7.6c) and (7.6f) yield equation (7.12c) as f_{jk} and f_{rk} vanish.

7.4 Summary

We have examined two cases of parametric resonance in the presence of the internal resonance $\omega_r \approx 3\omega_i$.

a. When λ is near twice the repeated frequency the response for the repeated modes is the same as for the case of no internal resonance. The third mode is merely drawn into the response.

b. When λ is away from twice the repeated frequency the response tends to be the same regardless of the presence of an internal resonance. The effect of the parametric resonance and the repeated frequency dominate the response.

c. In the case of no parametric resonance or a parametric resonance away from twice the repeated frequency, the form of the response for the modes associated with the repeated frequency is the same as that discussed in Chapter 5 where no resonance was present. Hence the only mode affected is the mode drawn into the response by the internal resonance.

CHAPTER EIGHT LATERAL VIBRATIONS OF COLUMNS

The lateral vibrations of columns are used as an example where the analysis contained in Chapters, 2, 3, and 4 can be applied. The natural frequencies of oscillations are distinct. The governing equation will be nondimensionalized and converted to the form discussed in Chapters 2, 3, and 4.

8.1 General

Relatively large-amplitude lateral deflections cause significant stretching of the neutral axis when the longitudinal displacements of the ends are restrained. For small, but finite deflections, one needs to use nonlinear geometrical relationships to account for this stretching. The equations governing the deflection can be written in the form

$$EI \frac{\partial^{4} w}{\partial x^{4}} + \rho A \frac{\partial^{2} w}{\partial t^{2}} = [p(t) + \frac{EA}{2\ell} \int_{0}^{\infty} (\frac{\partial w}{\partial \xi})^{2} d\xi] \frac{\partial^{2} w}{\partial x^{2}} - C \frac{\partial w}{\partial t}$$
(8.1)

The cubic term accounts for the stretching. The terms which account for shear deformation and rotary inertia are formally the same order as the term which accounts for stretching. However, these terms are linear, and consequently, they are responsible for only small perturbations of the solution of the linear problem. Here the focus is on nonlinear effects.

It is convenient to introduce nondimensional variables (denoted by primes); thus, we let

$$x = Lx', \quad w = \frac{r^2}{L} w', \quad t = \frac{L^2}{r} \sqrt{\frac{\rho}{E}} t'$$
 (8.2a)

$$p(t) = EA(\frac{r}{L})^{4} p'(t'), \quad C = \frac{2Ar^{3}}{L^{2}} \sqrt{\rho E} \quad C', \ \varepsilon = \frac{r^{4}}{L^{4}} EA \quad (8.2b)$$

Substituting equation (8.2) into equation (8.1) and dropping the primes, we obtain

$$\frac{\partial^4 w}{\partial x^4} + \frac{\partial^2 w}{\partial t^2} = \varepsilon \left\{ \left[p(t) + \frac{1}{4} \int_0^2 \left(\frac{\partial w(\xi, t)}{\partial \xi} \right)^2 d\xi \right] \frac{\partial^2 w}{\partial x^2} - 2C \frac{\partial w}{\partial t} \right\}$$
(8.3)

Here the characteristic length is one-half the actual length of the beam. We seek approximate solutions of equation (8.3) which are valid for small, but finite, values of ε .

We express the deflection as an expansion in terms of the linear free-oscillation modes as

$$w(x,t) = \sum_{m=1}^{\infty} U_m(t;\varepsilon)\phi_m(x)$$
(8.4)

In equation (8.4), the ϕ_m are the orthonormal solutions of the following:

$$\phi_{\rm m} \mathbf{i} \mathbf{v} - \omega_{\rm m}^2 \phi_{\rm m} = 0 \tag{8.5a}$$

where

$$\phi_{\rm m} = 0 \text{ and } \phi_{\rm m}' = 0 \tag{8.5b}$$

at clamped ends and

$$\phi_{\rm m} = 0 \text{ and } \phi_{\rm m}^{\rm m} = 0$$
 (8.5c)

at hinged ends. The ω_m are the natural frequencies and the ϕ_m are the linear free-oscillation modes.

Substituting equation (8.4) into equation (8.3), multiplying by ϕ_n , and integrating the result from x = 0 to x = 2, we obtain

$$\overset{\cdots}{U}_{n} + \omega_{n}^{2}U_{n} = \varepsilon \left[-2\cos\lambda t \prod_{m=1}^{\infty} f_{nm}U_{m} - 2C_{n}\dot{U}_{n} + \sum_{m=1}^{\infty} \sum_{p=1}^{\infty} \sum_{q=1}^{\infty} \Gamma_{nmpq}U_{m}U_{p}U_{q}\right]$$
for n = 1, 2, ... (8.6a)

where

$$P(t) = F \cos \lambda t$$

$$2f_{nm} = -F \int_{0}^{2} \phi_{n} \phi_{m}^{"} dx = F \int_{0}^{2} \phi_{n}^{'} \phi_{m}^{'} dx \qquad (8.6b)$$

and

$$\Gamma_{nmpq} = -\frac{1}{4} \int_{0}^{1} \phi'_{n} \phi'_{m} dx \int_{0}^{1} \phi'_{p} \phi'_{q} dx \qquad (8.6c)$$

Equations (8.6a) and (2.1) are the same.

8.2 Hinged-Hinged Beam

In this case,

$$\phi_n = \sin \sqrt{\omega_n} x \tag{8.7a}$$

and

$$\omega_{\rm n} = \frac{1}{4} \, {\rm n}^2 \pi^2 \tag{8.7b}$$

It follows that resonant combinations of natural frequencies abound; however, for this beam

$$\Gamma_{nmpq} = -\frac{1}{4} \omega_n \omega_p \delta_{nm} \delta_{pq}$$
(8.8a)

Thus all the ${\rm I}_{\rm n}$ are zero. Essentially, there is no internal resonance for this beam.

From equation (8.6b) it follows that

$$f_{nm} = \frac{1}{2} \omega_n F \delta_{nm}$$
(8.8b)

Therefore, resonances of the combination type, λ near $\omega_n \pm \omega_m$, are not possible for this beam.

Using equations (8.7) and considering λ to be near $2\omega_n$, we obtain the following from equations (3.7):

$$a'_{n} + C_{n}a_{n} + \frac{1}{4}Fa_{n}\sin\mu = 0$$
 (8.9a)

$$a_n \alpha'_n - 3a_n^3 \omega_n / 32 - \frac{1}{4} Fa_n \cos \mu = 0$$
 (8.9b)

where

$$\mu = \rho T_1 - 2\alpha_n \tag{8.9c}$$

For the non-trivial steady-state solution, we find that

$$\sin\mu = \frac{4C_n}{F}, \quad \cos\mu = \frac{16\rho - 3\omega_n a_n^2}{8F}$$
 (8.10a)

and

$$a_n = \sqrt{\frac{16\rho}{3\omega_n}} \pm \frac{8}{3\omega_n} \sqrt{F^2 - 16C_n^2}$$
 (8.10b)

It follows from (8.10b) that

$$F \ge 4C_n \tag{8.11}$$

if a real solution for a_n can exist. If $F \ge 4C_n$, then

(1) when

$$p < -\frac{1}{2}\sqrt{F^2 - 16C_n^2}$$
, (8.12a)

no real solution exists;

(2) when

$$\frac{1}{2} \mathbf{f}^2 - 16C_n^2 \leq \rho \leq \frac{1}{2} \sqrt{F^2 - 16C_n^2}$$
(8.12b)

one real solution exists;

(3) when

$$\rho > \frac{1}{2}\sqrt{F^2 - 16C_n^2}$$
 (8.12c)

two real solutions exist.

Now we consider the stability of the various possibilities . Using the analysis in Chapter 3, equations (3.8) through (3.13), we find that for this case equation (3.11) becomes

$$\Omega = -C_{n} \pm \sqrt{C_{n}^{2} \pm \frac{3\omega_{n}}{32}} a_{ns}^{2} \sqrt{F^{2} - 4C_{n}^{2}}$$
(8.13)

When the negative sign inside the radical is used, the real parts of both values of Ω are negative and hence the coresponding motion is stable. The negative sign corresponds to the larger a_n given by equation (8.10b). Hence, when two real solutions exist, only the one with the large amplitude is stable, and when only one real solution exists, it is stable.

The stability of the trivial solutions can be examined in a similar fashion. It turns out that, when ρ is in the region where there is no real solution of equation (8.10b), the trivial solution is stable; when ρ is in the region where there is one real solution of equation (8.10b), the trivial solution is unstable; and when ρ is in the region where there are two real solutions of equation (8.10b), the trivial solution is stable.

The results are illustrated in Figures 1-3 when λ is near ω_1 . For combinations of F and ρ in Region I, no nontrivial steady-state solution



Figure 1. Parameter space, F is the amplitude and $\rho(\epsilon \rho = \lambda - 2\omega_1)$ is the frequency of the excitation.

exists, and the trivial solution is stable. In Region II, the trivial solution is unstable, while the only nontrivial solution is stable. In Region III, two nontrivial solutions are possible. The trivial solution and the larger-amplitude nontrivial solution are stable, while the smaller-amplitude nontrivial solution is unstable.

In Region I, it appears that regardless of how large the initial amplitude is, the phasing never becomes such that the rate at which work is being done is as large as the rate at which energy is being dissipated. Indeed, if the damping term were deleted from equation (8.9a), the predicted response would still decay. Thus, we conclude that, for the linear as well as the nonlinear system, the phasing is such that the force actually does negative work and contributes to the decay.

In Region II, for a linear system the phasing, which does not change with amplitude, is such that work is being done at a faster rate than energy is being dissipated and thus the response to any initial disturbance grows without bound. For the nonlinear system, the phasing for large amplitudes differs from the phasing in the linear system as a consequence of the nonlinear term in equation (8.9a). The effect is to limit the rate at which work is being done to the rate at which energy is being dissipated and thereby to produce a bounded harmonic response. If the initial disturbance is very large, the response will decay until the steady-state solution is reached, while if the initial disturbance is very small the response will grow (the system being governed by the linear equations when the amplitude is small) until the nonlinear term in equation (8.9b) becomes large enough to cause the phase to shift

significantly. Thus, in Region II all initial disturbances produce the same steady-state response.

In Region III, the response of the linear system to an initial disturbance always decays. The mechanism causing the decay is the same as in Region I. The results for the nonlinear system show that only the larger of the two possible steady-state responses is stable. Thus, it appears that, if the initial disturbance is small, the nonlinear term in equation (8.9b) does not have a strong influence on the resulting motion and the system behaves essentially as a linear system; the motion decays. On the other hand, when the initial disturbance is large enough to produce the nontrivial steady-state response, the nonlinear term has a strong influence, and changes in the phase occur. Thus, in Region III there is the possibility of producing motions which have characteristics that are similar to those of the motions in Region I as well as those in Region II. The type of motion is determined by the amplitude and phase of the initial disturbance. This is a rare example in which a nontrivial, steady-state response of the nonlinear system exists in a region where the response of the linear system decays.

Let us suppose that the frequency of the excitation is varied while the amplitude is held fixed. This process is represented by the line through points A_1 , A_2 and A_3 . Between points A_1 and A_2 , the trivial solution is stable; moreover, it is the only steady-state solution possible. Between points A_2 and A_3 , the trivial solution is unstable, and the only realizable solution is the one given by equation (8.10b). Beyond point A_3 , the trivial solution is again stable and two solutions are realizable -- the trivial solution and the larger of the two solutions given by equation (8.10b).

Finally, let us suppose the amplitude of the excitation is varied while the frequency is held fixed. This process is represented by the line through points B_1 , B_2 and B_3 . Between points B_1 and B_2 , only the trivial solution is possible because the requirement of equation (8.11) is not satisfied. Between points B_2 and B_3 , two solutions are realizable-the trivial solution, which is stable, and the larger of the two solutions given by equation (8.10b). Beyond point B_3 , the trivial solution is unstable and the only realizable solution is the one given by equation (8.10b).

For the first process, the amplitude of the solution is plotted as a function of ρ in Figure 2. If the frequency of the excitation decreases from a large value, we note that upon reaching point A₃, where the trivial solution becomes unstable, there is a jump in the value of a₁ up from point A₃ to point A'₃. This process is indicated by the arrows.

For the second process, the amplitude of the solution is plotted as a function of F in Figure 3. We note the presence of a jump phenomenon. If F is increased slowly from Point 0 toward point B_1 , a_1 remains zero. However, the trivial solution becomes unstable at B_3 and a slight increase in F at this point causes a_1 to jump up to point B_3° . Then further increases cause a_1 to follow curve $CB_3^{\circ}D$ toward point D. When F is decreased a_1 follows curve $CB_3^{\circ}D$ away from point D pasts point B_3° , without jumping down to point B_3 , until it reaches point C. At this point a slight decrease in F causes a_1 to jump back to zero at point B_2 .







amplitude of the response, a₁, as a function of the amplitude of the excitation, F, for constant frequency of the excitation, p. —— stable, ---- unstable.

The arrows indicate this path. Figure 3 corresponds to ρ greater than zero. As ρ decreases, point C approaches point B₃, and the multi-valued region and hence the jump vanish.

The results for λ near $2\omega_n,$ where n = 2, 3,..., are similar to the above results.

Next we consider an example for which the internal resonance plays an important role.

8.3 Hinged-Clamped Beam

For a beam which is hinged at the left end and clamped at the right $\phi_n = \psi_n \left[\sin \sqrt{\omega_n} x - \frac{\sin 2\sqrt{\omega_n}}{\sinh 2\sqrt{\omega_n}} \sinh \sqrt{\omega_n} x \right] \quad (8.16a)$

where

$$\psi_{n} = \frac{\sinh 2\sqrt{\omega_{n}}}{\left[\sinh^{2} 2\sqrt{\omega_{n}} - \sin^{2} 2\sqrt{\omega_{n}}\right]^{1/2}}$$
(8.16b)

and the $\boldsymbol{\omega}_n$ are the roots of

 $tanh2\sqrt{\omega_n} - tan2\sqrt{\omega_n} = 0$ (8.16c)

The first five roots of equation (8.16c) are

 $\omega_1 = 3.855, \omega_2 = 12.491, \omega_3 = 26.062, \omega_4 = 44.567, and$

 $\omega_5 = 68.008$

Referring to equation (2.8a), we find that for the first five modes the combinations for which $\varepsilon\sigma$ is below eight percent of ω_n are

$$3\omega_1 = \omega_2 - \varepsilon \sigma_1 \tag{8.18a}$$

$$\omega_1 + \omega_2 + \omega_3 = \omega_4 + \varepsilon \sigma_2 \tag{8.18b}$$

$$2\omega_2 + \omega_4 = \omega_5 + \varepsilon \sigma_3 \tag{8.18c}$$

$$\omega_2 + 2\omega_3 = \omega_5 + \varepsilon \sigma_4 \tag{8.18d}$$

$$\omega_4 + \omega_3 - \omega_1 = \omega_5 + \varepsilon \sigma_5 \tag{8.18e}$$

$$2\omega_4 - \omega_3 = \omega_5 + \varepsilon \sigma_6 \tag{8.18f}$$

Referring to equations (2.9), we find that the conditions for the elimination of secular terms are

$$\omega_1(a_1^1 + C_1a_1) - 3Q_1a_1^2a_2\sin\mu_1 + Q_2a_2a_3a_4\sin\mu_2 - Q_5a_3a_4a_5\sin\mu_5 - Imag[E_1exp(-i\alpha_1)] = 0$$
(8.19)

$$\omega_1 a_1 \alpha'_1 + \frac{1}{8} a_1 \sum_{m=1}^{n} \gamma_1 m^2_m + 3Q_1 a_1^2 a_2 \cos \mu_1 + Q_2 a_2 a_3 a_4 \cos \mu_2$$

+
$$Q_{5}a_{3}a_{4}a_{5}\cos\mu_{5}$$
 + $Re[E_{1}exp(-i\alpha_{1})] = 0$ (8.19b)

 $\omega_2(a_2^{\dagger} + C_2a_2) + Q_1a_1^3\sin\mu_1 + Q_2a_1a_3a_4\sin\mu_2 + 2Q_3a_2a_4a_5\sin\mu_3$

+
$$Q_4a_3^2a_5sin\mu_4$$
 - Imag[E₂exp(-i α_2)] = 0 (8.20a)

$$\omega_{2}a_{2}\alpha_{2}^{1} + \frac{1}{8}a_{2}\sum_{m=1}^{5}\gamma_{2m}a_{m}^{2} + Q_{1}a_{1}^{3}\cos\mu_{1} + Q_{2}a_{1}a_{3}a_{4}\cos\mu_{2}$$

+ 2Q_{3}a_{2}a_{4}a_{5}\sin\mu_{3} + Q_{4}a_{3}^{2}a_{5}\sin\mu_{4} + RE[E_{2}exp(-i\alpha_{2})] = 0
(8.20b)
(8.20b)

$$- Q_{6}a_{4}^{2}a_{5}\sin\mu_{6} - Imag[E_{3}exp(-i\alpha_{3})] = 0 \qquad (8.21a)$$

$$\omega_{3}a_{3}\alpha_{3}^{4} + \frac{1}{8}a_{3}\sum_{m=1}^{5}\gamma_{3m}a_{m}^{2} + Q_{2}a_{1}a_{2}a_{4}\cos\mu_{2} + 2Q_{4}a_{2}a_{3}a_{5}\cos\mu_{4}$$

$$+ Q_{5}a_{1}a_{4}a_{5}\cos\mu_{5} + Q_{6}a_{4}^{2}a_{5}\cos\mu_{6} + Re[E_{3}exp(-i\alpha_{3})] = 0$$

$$(8.21b)$$

$$\omega_{4}(a_{4}^{4} + C_{4}a_{4}) - Q_{2}a_{1}a_{2}a_{3}\sin\mu_{2} + Q_{3}a_{2}^{2}a_{5}\sin\mu_{3} + Q_{5}a_{1}a_{3}a_{5}\sin\mu_{5}$$

+
$$2Q_{6}a_{3}a_{4}a_{5}\sin\mu_{6} - Imag[E_{4}exp(-i\alpha_{4})] = 0$$
 (8.22a)
 $\omega_{4}a_{4}\alpha_{4}^{4} + \frac{1}{8}a_{4}\sum_{m=1}^{5}\gamma_{4}m_{m}^{2} + Q_{2}a_{1}a_{2}a_{3}\cos\mu_{2} + Q_{3}a_{2}^{2}a_{5}\cos\mu_{3}$
+ $Q_{5}a_{1}a_{3}a_{5}\cos\mu_{5} + 2Q_{6}a_{3}a_{4}a_{5}\cos\mu_{6} + Re[E_{4}exp(-i\alpha_{4})] = 0$

$$\omega_{5}(a_{5}^{4} + C_{5}a_{5}) - Q_{3}a_{2}^{2}a_{4}\sin\mu_{3} - Q_{4}a_{2}a_{3}^{2}\sin\mu_{4} - Q_{5}a_{1}a_{3}a_{4}\sin\mu_{5}$$

$$- Q_{6}a_{3}a_{4}^{2}\sin\mu_{6} - Imag[E_{5}exp(-i\alpha_{5})] = 0 \qquad (8.23a)$$

$$\omega_{5}a_{5}\alpha_{5}^{4} + \frac{1}{8}a_{5}\sum_{m=1}^{5}\gamma_{m}a_{m}^{2} + Q_{3}a_{2}^{2}a_{n}\cos\mu_{3} + Q_{6}a_{2}a_{3}^{2}\sin\mu_{4}$$

+ $Q_5a_1a_3a_4\cos\mu_5$ + $Q_6a_3a_4^2\cos\mu_6$ + $Re[E_5exp(-i\alpha_5)] = 0$

(8.23b)

where

$$Q_{1} = \Gamma_{1112}/8$$

$$Q_{2} = (\Gamma_{1423} + \Gamma_{1234} + \Gamma_{1324})/4$$

$$Q_{3} = (\Gamma_{2254} + 2\Gamma_{2425})/8$$

$$Q_{4} = (\Gamma_{2533} + 2\Gamma_{2335})/8$$

$$Q_{5} = (\Gamma_{1435} + \Gamma_{1345} + \Gamma_{1534})/4$$

$$Q_{6} = (\Gamma_{3544} + 2\Gamma_{3445})/8$$

$$\mu_{1} = \alpha_{2} - 3\alpha_{1} + \sigma_{1}T_{1}$$

$$\mu_{2} = \alpha_{1} - \alpha_{2} - \alpha_{3} - \alpha_{4} - \sigma_{2}T_{1}$$

$$\mu_{3} = 2\alpha_{2} + \alpha_{4} - \alpha_{5} + \sigma_{3}T_{1}$$

$$\mu_{4} = \alpha_{2} + 2\alpha_{3} - \alpha_{5} + \sigma_{4}T_{1}$$

$$\mu_{5} = \alpha_{3} + \alpha_{4} - \alpha_{1} - \alpha_{5} - \sigma_{5}T_{1}$$

$$\mu_{6} = 2\alpha_{4} - \alpha_{3} - \alpha_{5} + \sigma_{6}T_{1}$$

and the ${\rm E}_{\rm n}$ depend on the excitation.

In this section, the following three cases are considered:

- (1) λ near $2\omega_1$
- (2) λ near $2\omega_2$
- (3) λ near $\omega_1 + \omega_2$

It follows that the E_n , for n = 3, 4, and 5, are zero for all three cases being considered; thus, from equations (8.19) - (8.23), it follows that a possible solution is

 $a_3 = a_4 = a_5 \equiv 0$

and a_1 and a_2 are given by

 $\omega_1(a_1^{\prime} + C_1 a_1) - 3Q_1 a_1^2 a_2 \sin \mu_1 - Imag[E_1 exp(-i\alpha_1)] = 0 \quad (8.24a)$

$$\omega_1 a_1 \alpha_1^{i} + \frac{1}{8} a_1 (\gamma_{11} a_1^2 + \gamma_{12} a_2^2) + 3Q_1 a_1^2 a_2 \cos \mu_1 + \text{Re}[E_1 \exp(-i\alpha_1)] = 0$$
(8.24b)

and

$$\omega_{2}(a_{2}^{\prime} + C_{2}a_{2}) + Q_{1}a_{1}^{3}\sin\mu_{1} - Imag[E_{2}exp(-i\alpha_{2})] = 0 \qquad (8.25a)$$

$$\omega_{2}a_{2}\alpha_{2}^{\prime} + \frac{1}{8}a_{2}(\gamma_{2}a_{1}^{2} + \gamma_{2}a_{2}^{2}) + Q_{1}a_{1}^{3}\cos\mu_{1} + Re[E_{2}exp(-i\alpha_{2})] = 0 \qquad (8.25b)$$

which are either equations (4.30) or (4.45) depending on the form of the parametric resonance. Tso and Asmis considered the system above, but took Q_1 to be zero.

Case (1), λ near $2\omega_1$

To express the nearness of λ to $2\omega_1,$ we introduce another detuning factor ρ defined by

$$\lambda = 2\omega_1 + \varepsilon \rho = \omega_2 - \omega_1 + \varepsilon (\rho - \sigma_1)$$
(8.26)

where equation (8.18a) was used. Then equations (8.24) and (8.25) can be rewritten exactly as (4.30) with m=1 and k=2.

$$\omega_{1}(a_{1} + C_{1}a_{1}) + \frac{1}{2}f_{11}a_{1}\sin\beta - \frac{1}{2}f_{12}a_{2}\sin(\beta - \mu_{1})$$

$$- 3Q_{1}a_{1}^{2}a_{2}\sin\mu_{1} = 0 \qquad (8.27a)$$

$$\omega_{1}a_{1}\alpha_{1}^{1} + \frac{1}{8}a_{1}(\gamma_{11}a_{1}^{2} + \gamma_{12}a_{2}^{2}) - \frac{1}{2}f_{11}a_{1}\cos\beta - \frac{1}{2}f_{12}a_{2}\cos(\beta - \mu_{1})$$

$$+ 3Q_{1}a_{1}^{2}a_{2}\cos\mu_{1} = 0 \qquad (8.27b)$$

$$\omega_{2}(a_{2}^{1} + C_{2}a_{2}) + \frac{1}{2}f_{21}a_{1}\sin(\beta - \mu_{1}) + Q_{1}a_{1}^{3}\sin\mu_{1} = 0 \qquad (8.27c)$$

$$\omega_{2}a_{2}\alpha_{2}^{1} + \frac{1}{8}a_{2}(\gamma_{21}a_{1}^{2} + \gamma_{22}a_{2}^{2}) - \frac{1}{2}f_{21}a_{1}\cos(\beta - \mu_{1}) + Q_{1}a_{1}^{3}\cos\mu_{1} = 0$$

where

$$\beta = \rho T_1 - 2\alpha_1 \tag{8.28a}$$

(8.27d)

 $\mu_1 = \sigma_1 T_1 + \alpha_2 - 3\alpha_1 \tag{8.28b}$

For the steady-state response,

 $a_1^i = a_2^i = 0$ and $\mu_1^i = \beta^i = 0$ (8.29)

It follows that the trivial solution is possible:

$$a_1 = a_2 = 0$$
 (8.30)

If neither a_1 nor a_2 is zero, one can use equations (8.28) and (8.29) to rewrite equations (8.27) as

$$\omega_1 C_1 a_1 + \frac{1}{2} f_{11} a_1 \sin\beta - \frac{1}{2} f_{12} a_2 \sin(\beta - \mu_1) - 3Q_1 a_1^2 a_2 \sin\mu_1 = 0$$
(8.31a)

$$\frac{1}{2}\omega_{1}a_{1}\rho + \frac{1}{8}(\gamma_{11}a_{1}^{2} + \gamma_{12}a_{2}^{2})a_{1} - \frac{1}{2}f_{11}a_{1}\cos\beta - \frac{1}{2}f_{12}a_{2}\cos(\beta - \mu_{1}) + 3Q_{1}a_{1}^{2}a_{2}\cos\mu_{1} = 0$$
(8.31b)

$$\omega_{2}C_{2}a_{2} + \frac{1}{2}f_{21}a_{1}\sin(\beta - \mu_{1}) + Q_{1}a_{1}^{3}\sin\mu_{1} = 0 \qquad (8.31c)$$

$$\omega_{2}a_{2}(\frac{3}{2}\rho - \sigma_{1}) + \frac{1}{8}(\gamma_{21}a_{1}^{2} + \gamma_{22}a_{2}^{2})a_{2} - \frac{1}{2}f_{21}a_{1}\cos(\beta - \mu_{1}) + Q_{1}a_{1}^{3}\cos\mu_{1} = 0 \qquad (8.31d)$$

A Newton-Raphson technique is used to find the solutions of equations (8.31).

In Figure 4, some typical results show the amplitudes of the first and second modes $(a_1 \text{ and } a_2)$ as functions of the amplitude of the excitation F, (refer to equation 8.6b). We note the multiplicity of possible jumps; jumps are indicated by the arrows. We can trace the histories of a_1 and a_2 as F increases very slowly from zero to a large value and then slowly decreases back to zero. Initially, both a_1 and a_2 are zero, and they remain zero until point A is reached. At this point the trivial solution becomes unstable. Then the amplitudes jump either to curve BCDE or FGH. If the jump is to G, then as F continues to increase, the

AD-A065 175 UNCLASSIFIED		ARMY MILITARY PERSONNEL CENTER ALEXANDRIA VA F/G 20/11 PARAMETRICALLY EXCITED NONLINEAR MULTI-DEGREE-OF-FREEDOM SYSTEMETC(U) MAR 79 E G TEZAK NL											
	2 OF 2 ADA 065175		- A second se					A A A				-	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
			R AND			A second							
ultimus (m. 4) in minimus (m. 4)													" a rear the
		$(1,1,2,\dots,2) \in \mathbb{R}^{d_{1}}$								A manufacture and the second s	The second secon		
							 The second second	Manager -	The second secon	<text><text><text><text><text></text></text></text></text></text>			
			The second secon	And a state of the	END DATE FILMED 4 79								



response follows curve FGH. If the jump is to D, then as F continues to increase the response follows BCDE until another jump occurs from E to H. An additional increase in F causes the response to follow FGH again. Decreasing F at this time causes the amplitudes to remain on FGH until F is reached. Then a jump to either C or the trivial solution occurs. If the jump is to C, then an increase or decrease in F causes the response to follow BCDE until the next jump is encountered. The jump at E has already been discussed and the one at B returns the amplitudes to the trivial solution. In Figure 5 the amplitudes are shown as a function of the detuning ρ . Possible jumps are again indicated by the arrows. The response for a_1 is initially similar to that of Figure 2 beginning with the trivial solution when ρ is very negative. As ρ increases the trivia! solution becomes unstable and the amplitudes follow curve ABC as in Figure 2. As ρ increases beyond C a transfer of energy occurs between the modes, and a_2 grows at the expense of a_1 until E is reached. The dominance of the second mode is quite clear. An additional increase in ρ produces jumps. The amplitudes jump either to G or they jump to the trivial solution. A further increase in ρ will cause the response to continue along FG or remain trivial, depending on which way the jump occured.

At this point, if the trivial response was attained, the solution would remain trivial for a decrease in ρ until H is reached. Here another jump occurs from H to B since the trivial solution is unstable. below H. Once at B, an increase or decrease in ρ would cause the response to follow ABCDE.





 $\rho(\epsilon\rho$ = λ - $2\omega_1$), for constant amplitude of the excitation, F.

---- stable, ---- unstable.

If the jump from E had been to G and then ρ decreased, the response would follow GF to F where jumps again occur. Both amplitudes could become trivial or jump to D. An increase or decrease in ρ at this point would cause the response to follow ABCDE again or remain trivial until H is reached as discussed above.

Case (2), λ near $2\omega_2$

In this case, we introduce the detuning parameter ρ according to $\lambda = 2\omega_2 + \epsilon\rho$ (8.32) For the steady-state, equations (8.34) and (8.35) reduce to equations (4.49) with m = 1 and k = 2

 $a_1(\omega_1 C_1 - 3Q_1 a_1 a_2 \sin \mu_1) = 0 \qquad (8.33a)$

$$a_{1}[\omega_{1}(\sigma_{1} + \frac{1}{2}\rho)/3 + \frac{1}{8}(\gamma_{11}a_{1}^{2} + \gamma_{12}a_{2}^{2}) + 3Q_{1}a_{1}a_{2}\cos\mu_{1}] = 0$$
(8.33b)

$$\omega_2 C_2 a_2 + \frac{1}{2} f_{22} a_2 \sin\beta + Q_1 a_1^3 \sin\mu_1 = 0 \qquad (8.34a)$$

 $\frac{1}{2}\omega_2\rho a_2 + \frac{1}{2}(\gamma_{21}a_1^2 + \gamma_{22}a_2^2)a_2 - \frac{1}{2}f_{22}a_2\cos\beta + Q_1a_1^3\cos\mu_1 = 0$ (8.34b)

where

 $\beta = \rho T_1 - 2\alpha_2 \tag{8.35a}$

and as above

 $\mu_1 = \sigma_1 T_1 + \alpha_2 - 3\alpha_1 \tag{8.35b}$

From equations (8.33) and (8.34), it appears that there are two possibilities for a nontrivial solution: either a_1 is zero and a_2 is nonzero or both a_1 and a_2 are nonzero. Thus, the situation is the same as that in Case (1). When a_1 is zero, the results resemble those illustrated in Figures 1-3; consequently, they are not given here. The results when both a_1 and a_2 differ from zero are illustrated in Figures 6 and 7. The top and bottom branches go together, and the middle two branches go together. The important thing to note is that over a fairly wide range a_1 is much larger than a_2 . This means that the first mode dominates the deflection, though the second mode is the only one directly excited. Consequently, when both a_1 and a_2 differ from zero, there can be a significant transfer of energy from the second mode to the first.

Case (3), λ near $\omega_1 + \omega_2$

In this case we introduce a detuning parameter ρ defined as $\lambda = 4\omega_1 + \epsilon \rho = \omega_2 + \omega_1 + \epsilon (\rho - \sigma_1)$ (8.36) Also, we let

$$\mu_1 = \sigma_1 T_1 + \alpha_2 - 3\alpha_1 \tag{8.37a}$$

$$\mu_2 = (\rho - \sigma_1)T_1 - \alpha_1 - \alpha_2 \tag{8.37b}$$

For steady-state solutions $a'_n = \mu'_n = 0$, and equations (8.24) and (8.25) yield those of (4.22) with m = 1 and k = 2; that is

$$\omega_1 C_1 a_1 + \frac{f_{12} a_2}{2} \sin \mu_2 - 3Q_1 a_1^2 a_2 \sin \mu_1 = 0 \qquad (8.38a)$$

$$\omega_1 a_1 \frac{\rho}{4} + \frac{1}{8} (\gamma_{11} a_1^2 + \gamma_{12} a_2^2) a_1 - \frac{f_{12} a_2}{2} \cos \mu_2 + 3 Q_1 a_1^2 a_2 \cos \mu_1 = 0$$
(8.38b)

$$\omega_2 C_2 a_2 + \frac{f_{21} a_1}{2} \sin \mu_2 + Q_1 a_1^3 \sin \mu_1 = 0 \qquad (8.39a)$$

$$\omega_{2}a_{2}(\frac{3\rho}{4} - \sigma_{1}) + \frac{1}{8}(\gamma_{12}a_{1}^{2} + \gamma_{22}a_{2}^{2})a_{2} - \frac{f_{21}a_{1}}{2}\cos\mu_{2} + Q_{1}a_{1}^{3}\cos\mu_{1} = 0$$
(8.39b)



- stable, ---- unstable.



---- unstable.

Two possibilities exist: either a_1 and a_2 are zero, or neither is zero. Figures 8 and 9 show the variation of a_1 and a_2 with F and ρ . In both figures, the top curves for a_1 go with the bottom curves for a_2 , and the two middle curves go together. Again we note that over a wide range, a_1 is much larger than a_2 , and hence the first mode dominates the response.

8.4 Observations - Internal and Parametric Resonances.

a. The Effect of the Detuning Parameter ρ on the Response, $a_n \mbox{ vs F, } \lambda \mbox{ near } 2\omega_1.$

Some interesting observations may be noted concerning the behavior represented in Figure 4, which depicts the relationship between the response amplitudes, a_n , and the amplitude of the forcing function, F. The detuning parameter ρ measures the closeness of the frequency of the excitation to $2\omega_1$. In Figure 10, when ρ is small, the response is essentially indistinguishable from that of Figure 3, i.e. the case when no internal resonance is present. The amplitude of the second-mode is almost zero. As the frequency of the excitation increases, (i.e., ρ increases), the effect of the internal resonance begins to appear. The response may resemble that in Figure 10, or the second mode may dominate the first mode as shown in Figures 11 and 12. As the detuning of the parametric resonance approaches that of the internal detuning ($\rho \simeq \sigma$), we see in Figure 11, strong modal interaction and the possible dominance of a_2 over a_1 for a large range of the amplitude of the excitation, F.



---- unstable.






---- stable, ---- unstable.



---- unstable.



first mode, over a very limited range of the amplitude of the excitation. However, strong modal coupling is present, and the second mode amplitude becomes relatively significant at the expense of the first; hence we note energy being transferred from the excited mode to the one involved in the internal resonance. Figures 4, 10, 11, and 12 graphically demonstrate the transition from a single-mode response to a multimode response. The larger the detuning ρ is, the more the second mode enters the response.

b. The Effect of Damping and the Amplitudes of the Excitation on the Response, $a_n v s \rho$.

Using Figure 5 as a point of departure, we see that the effects of damping on the system can be made apparent. As a general rule, the smaller the damping coefficients C_n are, the more interaction one can expect. Increasing the damping coefficients tends to reduce the modal interaction to the point where the second-mode response is below that of the first mode, and there is no longer a crossover point between a_1 and a_2 . In addition, jump phenomena tend to disappear as damping increases. Figures 13 and 14 demonstrate both of these effects, As the damping increases, the response approaches that of Figure 2 as a limit, or the response aproaches the single-mode response without internal resonance. Increasing the amplitude of the excitation tends to spread the stable and unstable branches and straighten the curves, first the unstable branch, then the stable one. We note this behavior in comparing Figure 13 with Figure 14 and Figure 5 with Figure 15. As F increases beyond the zone where jumps can occur, (the central portion

98



Figure 13. The modal amplitudes as functions of the frequency of the excitation, $\rho(\epsilon \rho = \lambda - 2\omega_1)$, for constant amplitude of the excitation, F. —— stable, ---- unstable.

.





Figure 15. The modal amplitudes as functions of the frequency of the excitation, $\rho(\epsilon \rho = \lambda - 2\omega_1)$, for constant amplitude of the excitation, F. —— stable, ---- unstable.

of Figure 4), then a further increase in F tends to force a single-mode response, approaching that of Figure 2.

8.5 Summary

The method of solution described in Chapters 2, 3, and 4 is used to solve two numerical examples -- one without internal resonances (a hinged-hinged column) and one with internal resonances (a hinged-clamped column).

For hinged-hinged columns, there are in effect no internal resonances and no combination resonances. Thus, the parametric excitation can, in the first approximation, excite only one mode. Jump phenomena can be produced by varying either the frequency or the amplitude of the excitation. In this case the frequency of the response is always one-half. that of the excitation.

For hinged-clamped columns, there are both internal resonances and combination resonances. When λ is near $2\omega_1$ or $\omega_1 + \omega_2$, both the first and second modes are strongly excited (i.e., appear in the first approximation), but generally the first mode dominates the second. Qualitatively, the response in both cases closely resembles a single-mode (the first) response. However, in the first case (λ near $2\omega_1$) the frequency of the response is one-half that of the excitation, while in the second case it is one-fourth the frequency of the excitation.

When λ is near $2\omega_2$, two nontrivial solutions are possible -either the second mode only is strongly excited or the second and first modes are strongly excited. In the second case, the first mode dominates, and the response qualitatively resembles the response when λ is near $2\omega_1$ and when λ is near $\omega_1 + \omega_2$. Multiple jumps are possible. When only the second mode is excited, the frequency of the response is onehalf that of the excitation; on the other hand, when both modes are strongly excited, the frequency of the response is one-sixth that of the excitation.

Thus, for the cases considered, it is possible to produce responses having frequencies that are one-half, one-fourth, and one-sixth that of the excitation.

One excitation can produce rather dissimilar responses. On the other hand, several different excitations can produce responses which appear to be quite similar.

When λ is near $2\omega_1$, detuning the excitation allows the internal resonance to induce strong modal coupling and the second mode can be made to dominate the response. Increasing the damping and/or the amplitude of the excitation inhibits modal coupling. If the increase is large enough, it can result in essentially a single-mode response.

CHAPTER NINE

PANEL FLUTTER

The analysis contained in Chapters 5, 6, and 7. is applied to the motion of a plate in an airstream undergoing cylindrical bending (no spanwise bending). As in the last chapter, the governing equations are nondimensionalized, but here they are converted to the form discussed in Chapters 5, 6, and 7. The process is slightly more involved than for the beam.

9.1 General

Following Dowell (1966, 1975), we write the governing equation as follows

$$D \frac{\partial^{4} w}{\partial x^{4}} + \rho_{m} h \frac{\partial^{2} w}{\partial t^{2}} = [N_{x} + \frac{Eh}{2a} \int_{0}^{a} (\frac{\partial w}{\partial \xi})^{2} d\xi] \frac{\partial^{2} w}{\partial x^{2}}$$
$$- C \frac{\partial w}{\partial t} - \frac{\rho_{\infty} U_{\infty}^{2}}{(M^{2} - 1)^{1/2}} [\frac{\partial w}{\partial x} + (\frac{M^{2} - 2}{M^{2} - 1}) \frac{1}{U_{\infty}} \frac{\partial w}{\partial t}] \qquad (9.1)$$

where D is the flexural rigidity, $\rho_{\rm m}$ is the material density, h is the panel thickness, a is the panel length, ρ_{∞} is the density of the air in the freestream, U_{∞} is the freestream velocity, and M is the Mach number, and N_x refers to the inplane parametric loading and is harmonic in what follows. The edges of the plate are restrained so that stretching occurs, and the term which accounts for it has the same cubic form as that of a beam. Modal damping is again assumed. The final term accounts for the aerodynamic pressure due to the plate motion and it is in the form of the so-called "piston-theory" approximation.

It is convenient to introduce nondimensional variables (denoted by primes); thus we let

$$x = \frac{1}{2} ax', \qquad w = \frac{h^2}{a} w', \qquad t = \frac{a^2}{4} \frac{\rho_m n}{D} t'$$

$$\varepsilon^{\delta} F_{R_x} = \frac{N_x a^2}{4D}, \qquad \varepsilon^{\delta} N = \frac{3(1-v^2)h^2}{a^2}, \qquad 2\varepsilon^{\delta} C' = \frac{a^2}{4\sqrt{\rho_m hD}} C$$
(9.2)

The parameter which is a measure of the dynamic pressure is

$$\tilde{\lambda} = \frac{\rho_{\infty} U_{\infty}^2 a^3}{8(M^2 - 1)^{1/2} D}$$
(9.3)

The coefficient of the last aerodynamic term is

$$\frac{(M^2-2)}{(M^2-1)^{5/4}} = \frac{2\lambda\rho_{\infty}a}{\rho_{m}h}$$
(9.4)

and can be taken as small because the ratio of the density of air to the density of the material is small.

Substituting equations (9.2) and (9.3) into equation (9.1) and dropping the primes, we obtain

$$\frac{\partial^{4} w}{\partial x^{4}} + \frac{\partial^{2} w}{\partial t^{2}} + \tilde{\lambda} \quad \frac{\partial w}{\partial t} = \left[\varepsilon^{\delta} F_{R_{x}} + \varepsilon^{\delta} N \int_{0}^{2} \left(\frac{\partial w(\xi, t)}{\partial \xi} \right)^{2} d\xi \right] \frac{\partial^{2} w}{\partial x^{2}} \\ - 2\varepsilon^{\delta} C_{C} \quad \frac{\partial w}{\partial t}$$
(9.5)

where δ_0 , δ_F , δ_N , etc. conform with the ordering of Chapters 5-7. We express the deflection as an expansion in terms of the linear freeoscillation modes; that is

$$w(x,t) = \sum_{m=1}^{\infty} U_m(t;\varepsilon)\phi_m(x)$$
(9.6)

The $\boldsymbol{\varphi}_{m}$ are the orthonormal solutions of

 $\phi_{\rm m}^{\rm iv} - \omega_{\rm m}^2 \phi_{\rm m} = 0$

where

$$\phi_{\rm m} = 0 \quad \text{and} \quad \phi_{\rm m}' = 0 \tag{9.7a}$$

at clamped edges and

$$\phi_{\rm m} = 0 \text{ and } \phi_{\rm m}^{\rm u} = 0$$
 (9.7b)

at hinged edges. The ω_m are the natural frequencies and the φ_m are the linear free-oscillation modes.

Substituting equation (9.6) into equation (9.5), multiplying by ϕ_n , and integrating the result from x = 0 to x = 2, we obtain

$$\ddot{U}_{n} + \omega_{n}^{2}U_{n} + \tilde{\lambda} \sum_{m=1}^{\infty} \alpha_{nm}U_{m} = \left[-\varepsilon^{\circ F} \sum_{m=1}^{\infty} 2\cos\lambda tf_{nm}U_{m}\right]$$
$$- 2\varepsilon^{\circ C}C_{n}\dot{U}_{n} + \varepsilon^{\circ N} \sum_{m=1}^{\infty} \sum_{p=1}^{\infty} \sum_{q=1}^{\infty} \Gamma_{nmpq}U_{m}U_{p}U_{q}\right] (9.8a)$$

where

$$\alpha_{nm} = \int_{0}^{c} \phi_{n} \phi_{m}^{\dagger} dx \qquad (9.8b)$$

$$R_{\chi} = F \cos \lambda t$$
 (9.8c)

$$2f_{nm} = -F \int_{0}^{2} \phi_{n} \phi_{m}^{"} dx = F \int_{0}^{2} \phi_{n}^{'} \phi_{m}^{'} dx = FG_{nm}$$
 (9.8d)

$$\Gamma_{nmpq} = \int_{0}^{2} \phi_{n} \phi_{m}^{"} dx \int_{0}^{2} \phi_{p}^{'} \phi_{q}^{'} dx = -G_{nm}G_{pq} \qquad (9.8e)$$

At this stage with the beam, we had achieved the desired form of the governing equations, but this is not so for the flutter problem.

106

Equation (9.8a) may be written in matrix form as

$$\ddot{AU} + BU = \varepsilon^{\delta} F_{FU} + \varepsilon^{\delta} C_{CU} + \varepsilon^{\delta} N_{TU}$$
(9.9)

provided we use a finite number of terms in the summation. The matrices \ddot{U} and U are column matricies of length M. The remainder are square matrices of dimension M x M. Then we note that

$$[A] = [I]$$
 (9.10a)

The identity matrix

$$[B] = [\omega_n^2] + \lambda[\alpha_{nm}]$$
(9.10b)

where the frequency matrix is diagonal,

$$[F] = -2\cos\lambda t[f_{nm}] \qquad (9.10c)$$

$$[C] = -2 [C_{nm}]$$
 (9.10d)

and

$$[T] = - [U]^{T} [G_{nm}][U][G_{pq}]$$
(9.10e)

The system of equations which has been analyzed in Chapters 5 through 7 is of the form

$$IU' + JU = \varepsilon^{\delta} F F'U + \varepsilon^{\delta} C'U + \varepsilon^{\delta} \Lambda B'U + \varepsilon^{\delta} N T'U \qquad (9.11a)$$

where the first coefficient matrix is the identity matrix and

is of the Jordan canonical form, which, in this case is a diagonal matrix with one off-diagonal term that is non-zero; F', C' and T' contain the same type of terms as their counter parts in equation (9.9).

The matrix B' contains aerodynamic coefficients which appear when the dynamic pressure causes the stability parameter $\tilde{\lambda}$ to be near the critical value $\tilde{\lambda}_c$ which causes two natural frequencies to merge and flutter to ensue. A refinement of the matrix B' is required before an attempt can be made to transform equation (9.9) into equation (9.11a), the Jordan form. Equation (9.10b) may be written

$$[B] = [\omega_n^2] + \tilde{\lambda}_c[\alpha_{nm}] - (\tilde{\lambda}_c - \tilde{\lambda})[\alpha_{nm}] \qquad (9.12a)$$

$$= [B_c] + (\tilde{\lambda} - \tilde{\lambda}_c)[\alpha_{nm}]$$
(9.12b)

where $\tilde{\lambda} - \tilde{\lambda}_{c}$ is a measure of the nearness of $\tilde{\lambda}$ to the critical value $\tilde{\lambda}_{c}$, and can be expressed in terms of a detuning parameter as follows $\varepsilon^{\delta}\Lambda = \tilde{\lambda} - \tilde{\lambda}_{c}$ (9.12c)

Then it is actually B_c which must be converted to Jordan form when the value of the stability parameter reaches the critical value, $\tilde{\lambda}_c$, and the onset of flutter occurs. The two lowest eigenfrequencies of B_c become equal at this time. The larger eigenfrequencies remain very near the ω_n for $n \geq 3$.

It is now possible to introduce a similarity transformation which transforms the matrix B_c to a Jordan form. (A complete discussion of the choice of this matrix is contained in Appendix A). We thus choose a matrix Z such that

Then the similarity transformation that leaves the eigenvalues unchanged becomes

$$Z^{-1}B_{C}ZV = JV = \begin{bmatrix} \omega_{2} & 0 & 0 & 0 & . & . & 0 \\ 1 & \omega_{\omega}^{2} & 0 & 0 & . & . & 0 \\ 0 & 0 & \omega_{3}^{2} & . & . & 0 \\ . & . & . & . & . & . \\ 0 & . & . & . & . & . \\ 0 & . & . & . & . & . \\ 0 & . & . & . & . & . \\ 0 & . & . & . & . & . \\ 0 & . & . & . & . & . \\ W_{M} \end{bmatrix}$$
(9.13b)
$$Z^{-1}AZV = Z^{-1}IZV = IV$$
(9.13c)

$$Z^{-1}FZV = F'V$$
 (9.13d)

and

U

$$Z^{-1}TZV = T'V$$
 (9.13e)

This last transformation merits some detailed consideration. From equation (9.13a) we can write the transpose as

$$u^{T} = v^{T} Z^{T}$$
(9.13f)

Then using equations (9.10e) and (9.13e) we write

$$Z^{-1}TZV = Z^{-1}V^{T}Z^{T}G_{nm}ZVG_{pq}ZV$$
(9.13g)

Letting

$$G_{k} = Z^{T}G_{nm}Z$$
 (9.13h)

we note that

v^TG₁v (9.13i)

is a quadratic form and it is thus a scalar. Hence we may write equation (9.13g) as $V^{T}G_{1}VZ^{-1}G_{pq}ZV$ (9.13j) Letting

$$G_2 = Z^{-1}G_{pq}Z$$
 (9.13k)

we write

$$T'V = -V^{T}G_{1}VG_{2}V$$
 (9.131)

Equation (9.131) has the same cubic form as before, or

$$\sum_{m} \sum_{p} \sum_{q} \Gamma'_{nmpq} V_m V_p V_q \qquad (9.14a)$$

where

$$\Gamma'_{nmpq} = G_{1}_{nm}G_{2}_{pq} \tag{9.14b}$$

Finally, the second half of the aero-matrix of equation (9.12b) can be transformed into

$$B' = -Z^{-1}[\alpha]Z = \beta$$
 (9.15)

Hence

$$\ddot{\mathbf{IV}} + \mathbf{JV} = \varepsilon^{\delta} \mathbf{F} \mathbf{F} \cdot \mathbf{V} + \varepsilon^{\delta} \mathbf{C} \cdot \mathbf{V} + \varepsilon^{\delta} \mathbf{N} \mathbf{T} \cdot \mathbf{V} + \varepsilon^{\delta} \Lambda \beta \mathbf{V}$$
(9.16)

and the panel flutter problem has been reduced to the form of equations (5.1). We shall now discuss the response for the various cases of parametric and internal resonances examined in Chapters 5-7.

9.2 Simply Supported Plate.

For cylindrical bending of a simply supported plate we may write

$$\phi_n = \sin \sqrt{\omega_n} x \tag{9.17a}$$

and

$$\omega_{\rm n} = \frac{1}{4} \, {\rm n}^2 \pi^2 \tag{9.17b}$$

where the ϕ_n and ω_n are the free-oscillation modes and frequencies. It follows from equation (9.8) that

$$\alpha_{nm} = \begin{cases} \frac{2nm}{n^2 - m^2} & \text{if } m + n \text{ is odd} \\ 0 & \text{if } m + n \text{ is even} \end{cases}$$
(9.18a)

$$f_{nm} = \frac{1}{2} \omega_n F \delta_{nm}$$
(9.18b)

$$G_{nm} = \frac{1}{2} \omega_n \delta_{nm}$$
(9.18c)

$$\Gamma_{nmpq} = -G_{nm}G_{pq} \tag{9.18d}$$

The transformations of equations (9.13) are applied to equations (9.18) to provide

β12 =	= 99.2	(9.19a)

$$f_{12} = 404.9 \frac{F}{2}$$
 (9.19b)

$$f_{13} = -3917 \frac{F}{2}$$
 (9.19c)

$$f_{32} = .1017 \frac{F}{2}$$
 (9.19d)
 $f_{33} = 22.74 \frac{F}{2}$ (9.19e)

$$\Gamma_{1222} = -5729$$
 (9.19f)

and

$$\Gamma_{3222} = -1.438 \tag{9.19g}$$

The frequencies become

 $\omega_1 = \omega_2 = 8.1076$ (9.20a)

 $\omega_3 = 22.066$ (9.20b)

While the critical value of the dynamic-pressure parameter is

$$\tilde{\lambda}_{c} = 42.917$$
 (9.20c)

where the ω_n and λ_c satisfy

$$\phi^{i\nu} + \tilde{\lambda}_{c}\phi' - \omega_{n}^{2}\phi = 0 \qquad (9.21)$$

The various cases of Chapters 5-7 are discussed, beginning with the response in the presence of a parametric resonance. The cases considered are

- (1) λ near $2\omega_1$
- (2) λ near $\omega_3 \pm \omega_1$

9.3 Parametric Resonance

a. Case (1) $\lambda \approx 2\omega_1$

The governing equations for this case were developed in Section 6.1, and for convenience we rewrite equations (6.11) as

$$a_{2}^{2} = \left[\frac{4}{3\Gamma_{jkkk}} \left(-\Lambda\beta_{12} + \omega_{1}^{2}(4C_{1}C_{2} - \rho^{2}) \pm \sqrt{f_{12}^{2} - 4\omega_{1}^{4}\rho^{2}(C_{1} + C_{2})^{2}}\right)\right]$$
(9.22a)

$$a_1 = \omega_1 a_2 \sqrt{\rho^2 + 4C_2^2}$$
 (9.22b)

A general observation that can be made immediately concerns the relative sizes of a_1 and a_2 . If

$$\sqrt{\rho^2 + 4C_2^2} < \frac{1}{\omega_1}$$
 (9.23a)

Hence, for $a_2 > a_1$ the detuning must be very nearly zero and the damping extremely small, otherwise $a_1 > a_2$. We note that even though $a_1 > a_2$

the response is not necessarily dominated by the first mode. The modal amplitude a_1 multiplies a one (1) whereas a_2 multiplies $\epsilon^{-\delta_0}$ and hence its effect can be much greater. Figure 16 shows the variation of the modal amplitude with the amplitude of excitation and the response has essentially the same form as that in Figure 3. The response contains the same regions where trivial and nontrivial solutions exist and jump phenomena are possible. We note that as the amplitude of the parametric resonance term increases, $0 < F < F_1$, the only solution is the trivial one and it is stable. Then we enter the region where two nontrivial solutions exist, $F_1 \le F \le F_2$, one stable and one unstable. Finally when $F \ge F_2$ the trivial solution is unstable and only one nontrivial solution exists, which is stable, and a jump occurs. If we scan in the opposite direction we start with a nontrivial stable solution and remain on that upper branch until we reach the region where the only solution is the stable trivial solution and a jump occurs. One can refer to Figure 3 where the arrows indicate the jump phenomena and the regions where trivial and nontrivial solutions exist are shown. The form of the response for the second mode is the same as the first except the amplitude is so small that it appears to be zero by comparison.

The analog of Figure 2, modal amplitude vs detuning of the excitation frequency, is Figure 17. This figure shows the modal amplitude a_1 vs the detuning of the dynamic-pressure parameter, Λ . However, the form of the response is essentially the same and exhibits the same regions of stability and jump phenomena. As we scan from large negative Λ to large positive Λ , or alternatively, as we increase the dynamic pressure from below the flutter speed, we encounter first the region



---- stable, ---- unstable.



detuning, Λ , for constant frequency of excitation $\rho(\epsilon \rho = \lambda - 2\omega_1)$ and constant amplitude of excitation, F. ---- stable, ---- unstable. where the trivial solution is the only solution and it is stable, $\Lambda \leq \Lambda_1$. Between $\Lambda_1 \leq \Lambda \leq \Lambda_2$ the trivial solution is unstable and the only nontrivial solution is stable. Beyond $\Lambda > \Lambda_2$ there is again a stable trivial solution and two nontrivial solutions, only the larger one being stable. Again the form of the second mode is essentially the same except the magnitude appears to be zero by comparison.

In Figure 18 we plot the modal amplitude vs the detuning of the parametric excitation ρ . This new graph has a character of its own and has no analog with the case of distinct frequencies. However, it still contains regions where multiple solutions exist and points out where jumps can occur. We can scan the frequency axis and observe the following. For $\rho \leq \rho_1$ only the trivial solution exists and it is stable. Between $\rho_1 \leq \rho \leq \rho_2$ two nontrivial solutions exist, the larger being stable, the smaller unstable. In addition, the trivial solution is stable, and since we are scanning from the left, the response remains trivial. Between $\rho_2 \leq \rho \leq \rho_3$ only one nontrivial solution exists and it is stable. Concurrently, the trivial solution becomes unstable and an upward jump occurs. Arrows indicate the jumps and paths followed by a1. As the frequency increases between $\rho_3 \leq \rho \leq \rho_4$ the response remains nontrivial, since that solution is stable. A nontrivial, unstable solution also exists as well as a stable trivial solution. For $\rho > \rho_4$ we see that the only solution possible is the trivial one, which is stable, and a downward jump occurs where the nontrivial solution ceases to exist. Scanning from the right we note a mirror-image response. The solution is first trivial until $\rho = \rho_3$. It jumps to the nontrivial





solution and remains nontrivial until $\rho = \rho_1$ at which time it jumps back to the trivial solution for further decreases in ρ .

When the damping is small and the frequency of excitation is almost exactly tuned, or ρ is very near zero, the condition of equation (9.23a) can be met. The second mode is greater than the first mode and we note from Figures 19 and 20 that the characteristic shape of the curves is the same. Hence, the amount of detuning and damping will determine which mode amplitude is larger.

It should be noted that for the plate considered, Γ_{1222} is negative in equations (9.22a) and the existence of any solution other than the trivial one depends upon the conditions

 $\Lambda \beta_{12} - \omega_1^2 (4C_1C_2 - \rho^2) \pm \sqrt{f_{12}^2 - 4\omega_1\rho^2(C_1 + C_2)^2} > 0$ (9.24a)

and

 $f_{12}^2 > 4\omega_1^4 \rho^2 (C_1 + C_2)^2$ (9.24b)

If (9.24b) is not satisfied then there are no nontrivial solutions. The same holds for (9.24a). If both conditions are met then there are two nontrivial solutions. The stable one is associated with the positive value of the radical. When only one solution exists, it is also associated with the positive value of the radical. In addition, it is dependent on the magnitude of the amplitude, F, of the parameteric excitation, the relative sizes of the detuning and the damping, and the aerodynamic detuning parameter, Λ . We note that in this case, β_{12} is a positive constant and that a positive value of Λ refers to a dynamic pressure in excess of the critical flutter speed. The presence of large damping tends to necessitate large forcing amplitudes, large detunings of the parametric terms, and large values of dynamic pressures to produce



---- stable, ---- unstable.



detuning, Λ , for constant frequency of excitation, $\rho(\epsilon\rho = \lambda - 2\omega_1)$ and constant amplitude of excitation, F. ----- stable, ---- unstable. nontrivial solutions. However, this same condition make a_2 very small compared with a_1 . On the other hand, small damping permits strong interaction of the modes and even allows a_2 to be much greater than a_1 when detuning is small. The amplitude of the parametric excitation and the aerodynamic detuning merely influence the amplitude of the response but not the relative sizes of a_1 and a_2 .

A few special cases may also be examined. If no parametric resonance is present, (9.24a) has the form

$$\Lambda \beta_{12} - 4C_1 C_2 \omega_1^2 > 0 \tag{9.25}$$

and there is either the trivial solution or one nontrivial solution, depending on whether (9.25) is satisfied. Figure 21 illustrates the response for small damping. When large damping is present $a_1 > a_2$. As $C_2 \rightarrow 0$, $a_1 \rightarrow 0$ and the only mode excited is the second, in spite of the parametric force exciting the first mode. The response is as shown in Figure 21 without a_1 being present.

B. Case (2) λ near $\omega_3 \pm \omega_1$.

The governing equations for these cases were developed in Sections (6.2) and (6.3) and are

$$\omega_1 C_1 a_1 + \frac{f_{13}}{2} a_3 \sin\beta_1 + \left[\frac{3}{8} \Gamma_{1222} a_2^3 + \frac{\Lambda}{2} \beta_{12} a_2\right] \sin\gamma = 0$$
(9.26a)

$$\omega_1 C_2 a_2 + \frac{a_1}{2} \sin \gamma = 0 \tag{9.26b}$$

$$\omega_{3}C_{3}a_{3} + \frac{f_{32}}{2}a_{2}\sin\beta_{2} = 0 \qquad (9.26c)$$

$$\rho = \frac{a_1}{2\omega_1 a_1} \cos \gamma + \frac{f_{32}a_2}{2\omega_3 a_3} \cos \beta_2$$
 (9.26d)

 $a_{1}^{2}\cos\gamma = 2a_{2} \left\{ \frac{f_{13}}{2} a_{3}\cos\beta_{1} - \left[\frac{3}{8}\Gamma_{1222}a_{2}^{3} + \frac{\Lambda}{2}\beta_{12}a_{k}\right]\cos\gamma \right\}$ (9.26e)



— stable, ---- unstable.

where

$$\beta_{1} = \begin{cases} \rho T_{2} - \alpha_{r} - \alpha_{j} & \text{for } \omega_{3} + \omega_{1} \\ -\rho T_{2} + \alpha_{r} - \alpha_{j} & \text{for } \omega_{3} - \omega_{1} \end{cases}$$
(9.27a)

$$\gamma = \alpha_j - \alpha_k \tag{9.27b}$$

$$\beta_{2} = \begin{cases} \rho T_{2} - \alpha_{r} - \alpha_{k} & \text{for } \omega_{3} + \omega_{1} \\ -\rho T_{2} - \alpha_{r} + \alpha_{k} & \text{for } \omega_{3} - \omega_{1} \end{cases}$$
(9.27c)

Equations (9.26) do not readily admit a closed-form solution so a Newton-Raphson technique was used to produce Figures 22, 23, and 24 for the case of λ near $\omega_1 + \omega_3$. A similar response to that of the last section is evident. In the presence of large damping, $a_1 > a_2$ and a_3 . Regions of stability exist as well as jump phenomena. Upon examining the response, amplitude vs detuning, Figure 24, we note that there is a loss of symmetry from the case of λ near $2\omega_1$.

9.4 Internal Resonance, $\omega_3 \simeq 3\omega_1$

We now investigate the response in the presence of an internal resonance of the form, ω_3 near $3\omega_1$. For the simply supported case this is within about 10%. By a proper choice of the boundary conditions, for example some form of elastic supports, it would be possible to obtain a relationship within much less than 10%. We now investigate the response in the presence of an internal resonance for the case of parametric resonance:

(1) λ near $2\omega_1$

(2) no parametric resonance

123





Figure 22. First modal amplitude as a function of aerodynamic detuning, Λ , for constant frequency of excitation, $\rho(\epsilon \rho = \lambda - \omega_1 - \omega_3)$, and constant amplitude of excitation, F. —— stable, ---- unstable.



e 23. First modal amplitude as a function of amplitude of excitation, F, for constant frequency of excitation, $\rho(\epsilon\rho = \lambda - \omega_1 - \omega_3)$ and constant aerodynamic detuning, Λ . ----- stable, ---- unstable.





A. Case (1) Where $\lambda \simeq 2\omega_1$.

Equations (9.22) remain valid and the entire discussion, concerning the response of the first and second modes, of Section 9.3 is valid. The only change is that the third mode may now be drawn into the response through equations (7.6c, f) or

$$\omega_{3}C_{3}a_{3} - \frac{1}{8}\Gamma_{3222}a_{2}^{3}\sin\beta + \frac{f_{32}a_{2}}{2}\sin\gamma = 0 \qquad (9.28a)$$

$$\omega_{3}a_{3}(\frac{3\rho}{2} - \sigma) + \frac{1}{8}\Gamma_{3222}a_{2}^{3}\cos\beta - \frac{f_{32}}{2}a_{2}\cos\gamma = 0 \qquad (9.28b)$$

where for ε = .001

$$\sigma = \frac{22.066 - 3(8.1076)}{(.001)^{1/2}} = -71.36$$
(9.28c)

Using a Newton-Raphson technique we find that the response is identical to that of Section 9.3 for no internal resonance. The relative sizes of a_1 and a_2 are controlled by the damping and detuning of the parametric resonance. The amplitude a_3 is very small and the response appears as shown in Figures 16 - 21. To obtain a qualitive feel for the conditions under which a_3 is drawn into the response, we let the amplitude F of the forcing term be small. Then equations (9.28) can be combined to yield

$$a_{3} \approx \frac{\Gamma_{3222} a_{2}^{3}}{8\omega_{3} \sqrt{C_{3}^{2} + (\frac{3\rho}{2} - \sigma)^{2}}}$$
(9.29)

Thus, a_3 will be small as long as the radical of (9.29) is large. For small damping the radical will be small only if

$$\frac{3\rho}{2} - \sigma \stackrel{\sim}{\sim} 0 \tag{9.30}$$

Since the internal-resonance detuning, σ , is fixed by virtue of the form of the boundary conditions, then the parametric detuning must be very nearly $2\sigma/3$ for a_3 to be significant. However, in this case σ is large and thus ρ must also be large. But from Section 9.3 we note that a large detuning caused a_1 to be so large that a_2 was negligible by comparison. So that for large detuning of both internal and parametric resonances the result is essentially that of Figures 16, 17, and 18.

As previously discussed, the detuning of the internal resonance is governed by the boundary conditions. For small damping and small detuning the response is depicted essentially as described by Figures 19 and 20 with the addition of the third mode; see Figures 25 and 26. Thus any of the three modes can be made greater than the other two, depending on the type of boundary conditions, damping, and detuning of the parametric excitation. A very striking representation of the relative sizes of the modes is portrayed in Figures 27 and 28. We note here that the response of the third mode is the same as that for a linear oscillator with a single degree-of-freedom. The second mode amplitude a_2 merely acts as the external forcing function. A comparison with Figure 18 can be made and we see how the decrease in damping allows the second and third modes to enter the response. The second mode can be seen to be larger than the first when the detuning of the excitation is near zero. Detuning the internal resonance shifts the response of the third mode considerably.

B. The Case of No Parametric Resonance.

This is essentially a special case of the previous one. In the absence of a parametric resonance the response for the first two modes

128





detuning, Λ , for constant frequency of excitation, $\sigma(\varepsilon\sigma = \lambda - 2\omega_1)$, constant amplitude of excitation, F, and internal resonance $\sigma(\varepsilon\sigma = \omega_3 - 3\omega_1)$. ----- stable, ---- unstable.


excitation, $\rho(\epsilon \rho = \lambda - 2\omega_1)$, for constant amplitude of excitation, F, constant aerodynamic detuning, Λ and internal resonance $\sigma(\epsilon \sigma = \omega_3 - 3\omega_1)$. —— stable, ---- unstable.



excitation, $\rho(\varepsilon \rho = \lambda - 2\omega_1)$, for constant amplitude of excitation, F, constant aerodynamic detuning, Λ and internal resonance $\sigma(\varepsilon \sigma = \omega_3 - 3\omega_1)$. —— stable, ---- unstable.

remains the same, or

$$a_{2} = \sqrt{\frac{4\omega_{1}^{2}C_{1}C_{2} - \Lambda\beta_{12}}{3/4 \Gamma_{1222}}}$$
(9.35a)
$$a_{1} = 2\omega_{1}C_{2}a_{2}$$
(9.35b)

and the third mode response is

$$a_{3} = \frac{\frac{1}{8} \Gamma_{3222} a_{2}^{3}}{\sqrt{\omega_{3}^{2} (C_{3}^{2} + \sigma^{2})}}$$
(9.35c)

The response is as shown in Figure 29. Again any of the three modes may be the largest depending on the damping and detuning of the internal resonance. For large damping $a_1 > a_2$ or a_3 . For small damping a_3 or $a_2 > a_1$ depending on the detuning of the internal resonance. In addition, the larger the dynamic pressure is the more likely the third mode will be greatest.

9.5. Summary

The method of solution described in Chapters 5, 6, and 7 is used to solve numerical examples of a simply supported plate undergoing cylindrical bending. Various combinations of internal and parametric resonances have been examined.

When neither internal nor parametric resonances are present the response depends on the extent to which the dynamic pressure exceeds the critical value for inducing flutter. The amplitude of the response increases as the dynamic pressure increases. Both modes associated with the repeated frequency are excited. Only one stable solution exists and it is either trivial or nontrivial depending on the dynamic pressure.





When a parametric resonance of the form λ near $2\omega_1$ or $\omega_3 \pm \omega_1$ is present, then two nontrivial solutions can exist but only one is stable. The only modes excited are associated with the repeated frequency. Jump phenomona can be produced by varying the frequency or amplitude of the parametric excitation or the dynamic pressure.

When both internal and parametric resonances are present all three modes can be excited. The detunings of internal and parametric resonances have a great effect on the third and second modes, respectively. When the detuning of the internal resonance is large, as in the case of the simply supported plate considered, we see that the response is essentially identical to the case of no internal resonance. Jump phenomena exist regardless of which amplitude is largest.

REFERENCES

- Andronov, A. and M. A. Leontovich (1927). On the vibrations of systems with periodically varying parameters. <u>Zh. Russk</u>. <u>Fiz-Khim Obshch.</u> 59, 429-443.
- Andronov, A., A. Vitt and S. Khaikin (1966). <u>Theory of Oscillations</u>. Addison-Wesley, Reading.
- Atluri, S. (1973). Nonlinear vibrations of a hinged beam including nonlinear inertia effects. <u>J. Appl. Mech.</u> 40, 121-126.
- Bayles, D. J., R. L. Lowery and D. E. Boyd (1973) Nonlinear vibrations of rectangular plates. ASCE J. Struct. Div 99, 853-864.
- Beliaev, N. M. (1924). Stability of prismatic rods, subject to variable longitudinal forces. Collection of Papers: <u>Eng. Construct. Struct.</u> <u>Mech. Put</u>; Leningrad, 149-167.
- Bennett, J. A. (1971). Nonlinear vibration of simply supported angle ply laminated plates. AIAA J. 9, 1997-2003.
- Bennett, J. A. and J. G. Eisley (1970). A multiple-degree-of-freedom approach to nonlinear beam vibrations. <u>AIAA</u> J. 8, 734-739.
- Bennett, J. A. and R. L. Rinkel (1972). Ultraharmonic vibrations of nonlinear beams. <u>AIAA</u> J. 10, 715-716.
- Bert, C. W. and C. A. Fisher (1972). Stability analysis of a pinned-end beam undergoing nonlinear free vibration. <u>J. Sound Vib</u>. 22, 129-131.
- Bogoliubov, N. N. and Y. A. Mitropolsky (1961). <u>Asymptotic Methods in</u> the Theory of Nonlinear Oscillations. Gordon and Breach, New York.

Bolotin, V. V. (1963). <u>Nonconservative Problems Of the Theory of</u> Elastic Stability. Pergamon, New York.

Bolotin, V. V. (1964). <u>The Dynamic Stability of Elastic Systems</u>. Holden-Day, San Francisco.

Bondarenko, G. V. (1936). <u>The Hill Differential Equation and its Uses</u> <u>in Engineering Vibration Problems</u>. Akademiia Nauk SSSR, Moscow. Bulkeley, P. Z. (1963). An axisymmetric nonlinear vibration of circular

plates. <u>J. Appl. Mech.</u> 30, 630-631.

Burgreen, D. (1951). Free vibrations of a pin ended column with constant distance between ends. J. <u>Appl. Mech.</u> 18, 135-139.

Busby, H. R., Jr. and V. I. Weingarten (1972). Non-linear response of a beam to periodic loading. <u>Int. J. Non-Linear Mech.</u> 7, 289-303.

Cesari, L. (1971). <u>Asymptotic Behavior and Stability Problems in Ordinary</u> Differential Equations, 3rd ed., Springer-Verlag, New York.

Chandra, R. (1976). Large deflection vibration of cross-ply laminated

plates with certain edge conditions. <u>J</u>. <u>Sound</u> <u>Vib</u>., 47, 509-514.

- Chandra, R. and B. Basava Raju (1975a). Large amplitude flexural vibration of cross ply laminated composit plates. <u>Fibre. Sci.</u> <u>Tech.</u> 8, 243-263.
- Chandra, R. and B. Basava Raju (1975b). Large deflection of angle ply laminated plates. J. Sound Vib. 40, 393-408.
- Chelomei, V. N. (1939). <u>The Dynamic Stability of Elements of Aircraft</u> Structures. Aeroflot, Moscow.
- Coddington, E. A. and N. Levenson, (1955). <u>Theory of Ordinary Differential</u> Equations. McGraw-Hill, New York.

- Craw, J. and S. Atluri (1975). Nonlinear vibrations of a flat plate with initial stresses. <u>J. Sound Vib</u>. 43, 117-129.
- Cunningham, W. J. (1958). <u>Introduction to Nonlinear Analysis</u>. McGraw-Hill, New York.
- Den Hartog, J. P. (1974). <u>Mechanical Vibrations</u>. McGraw-Hill, New York.
- E. H. (1966). Nonlinear oscillations of a fluttering plate. <u>AIAA J</u>. 4, 1267-1275.
- Dowell, E. H. (1967). Nonlinear oscillations of a fluttering plate II. AIAA J. 5,1856-1862.
- Dowell, E. H. (1970). Noise or flutter or both? <u>J</u>. <u>Sound Vib</u>. 11, 159-180.
- Dowell, E. H. (1973). Theoretical vibration and flutter studies of point supported panels. J. Spacecraft 10, 389-395.
- Dowell, E. H. (1975). <u>Aeroelasticity of Plates</u> and <u>Shells</u>. Noordhoff, Leyden.
- Dugundji, J. (1966). Theoretical considerations of panel flutter at high supersonic Mach numbers. <u>AIAA J</u>. 4, 1257-1266; <u>Errata and</u> <u>Addenda</u> 7, 1663.
- Dzygadlo, A. (1970). Local analysis of nonlinear forced vibrations of a plate of finite length in plane supersonic flow. <u>Proc</u>: <u>Vib</u>. Prob.
- Eastep, F. E. and S. C. McIntosh (1971). Analysis of nonlinear panel flutter and resonnse under ramdom excitation of nonlinear aerodynamic loading. AIAA J. 9, 411-418.

- Eisley, J. G. (1964a). Large amplitude vibration of buckled beams and rectangular plates. <u>AIAA</u> J. 2, 2207-2209.
- Eisley, J. G. (1964b). Nonlinear vibration of beams and rectangular plates. ZAMP 15, 167-175.
- Eisley, J. G. and J. A. Bennett (1970). Stability of large amplitude forced motion of a simply supported beam. <u>Int. J. Nonlinear Mech.</u> 5, 645-657.
- Ewan-Iwanowski, R. M. (1976). <u>Resonance Oscillations in Mechanical</u> <u>Systems</u>. Elservier, New York.

Evenson, D. A. (1968). Nonlinear vibrations of beams with various boundary conditions. <u>AIAA J.</u> 6, 370-372.

- Faraday, M. (1831). On a peculiar class of acoustical figures and on certain forms assumed by a group of particles upon vibrating elastic surfaces. <u>Phil Trans. Roy. Soc.</u>, London, 299-318.
- Fu, F. C. L. and S. Nemat-Nasser (1972). Stability of solution of systems of linear differential equations with harmonic coefficients. <u>AIAA</u> J. 10, 30-36.
- Fu, F. C. L. and S. Nemat-Nasser (1975). Response and stability of linear dynamic systems with many degrees of freedom subjected to nonconservative and harmonic forces. <u>J. Appl. Mech.</u> 42, 458-463.
- Hayashi, C. (1953). <u>Forced Oscillations in Nonlinear Systems</u>. Nippon, Osaka, Japan.
- Hayashi, C. (1964). <u>Nonlinear Oscillations in Physical Systems</u>. McGraw-Hill, New York.
- Hsu, C. S. (1963). On the parametric excitation of a dynamic system having multiple degrees of freedom. <u>J. Appl. Mech.</u> 30, 367-372.

Hsu, C. S. (1965). Further results on parametric excitation of a dynamic system. J. Appl. Mech. 32, 373-377.

Huang, C. L. (1972a). Nonlinear axisymmetric flexural vibration equations of a cylindrically anistropic plate. AIAA J. 10, 1378-1379.

Huang, C. L. (1972b). Nonlinear responses for a circular plate subjected to a dynamic ring load. In <u>Develop</u>. <u>The. Appl. Mech.</u> 6, 489-514.

Huang, C. L. (1973). Finite amplitude vibrations of an orthotropic circular plate with an isotropic core. <u>Int. J. Non-Linear Mech.</u> 8, 445-457.

Huang, C. L. (1974). Nonlinear oscillations of an annulus with variable thickness. In <u>Develop</u>. <u>The. Appl. Mech.</u> 7, 271-284.

- Huang, C. L. and Be. E. Sandman (1971). Large amplitude vibrations of a rigidly clamped circular plate. <u>Int. J. Non-linear Mech</u>. 6, 451-468.
- Huang, C. L. and H. K. Woo (1973). Large oscillations of an orthotropic annulus. In <u>Develop Mech</u>. 7, 1027-1039.
- Huang, C. L., H. K. Woo, and H. S. Walker (1976). Nonlinear flexural oscillations of a partially tapered annular plate. <u>Int. J. Non-</u> <u>Linear Mech</u>. 11, 89-97.
- Iwatsubo, T., M. Saigo and Y. Sugiyama (1973). Parametric instability of clamped-clamped and clamped-simply supported columns under periodic axial load. <u>J. Sound Vib</u>. 30, 65-77.
- Iwatsubo, T., Y. Sugiyama and S. Ogino (1974). Simple and combination resonances of columns under periodic axial loads. <u>J. Sound Vib.</u> 33, 211-222.

- Kobayashi, S. (1962a). Flutter of simply-supported rectangular panels n a supersonic flow.. two dimensional panel flutter I...simplysupported panel II ...clamped panel. <u>Trans. Japan Soc. Aero. Space</u> Sci. 5,
- Kobayashi, S. (1962b). Two-dimensional panel flutter II. Clamped panel. <u>Trans. Japan Soc.Aero. Space Sci</u>. 5, 103-118.
- Kreyszig, E. (1972). Advanced Engineering Mathematics, John Wiley and Sons, Inc, New York.
- Krylov, N. M. and N. N. Bogoliubov (1935). Calculations of the vibration of frame construction with the consideration of normal forces with the help of the methods of nonlinear mechanics. <u>In-</u> vestigation of Vibration of Structures ONTI Kharkov, Kiev, 5-24.
- Kuo, C. C., L. Morino and J. Dugundji (1972). Perturbation and harmonic balance methods for nonlinear panel flutter. <u>AIAA J</u>. 10, 1479-1484.
- Kuo, C. C., L. Morino, and J. Dugundji (1973). Nonlinear interaction of panel flutter with harmonic forcing excitation. <u>AIAA J</u>. 11, 419-420.

Leissa, A. W. (1969). Vibration of Plates. NASA-Sp-160.

Lobitz, D. W., A. H. Nayfeh and D. T. Mook (1977). Nonlinear analysis of vibrations of irregular plates. J. Sound Vib. 49, 203-217.

Lou, C. L. and D. L. Sikarskie (1975). Nonlinear vibration of beams using a form-function approximation. <u>J. Appl. Mech.</u> 42, 209-214.
 Malkin, I. G. (1956). <u>Some Problems in the Theory of Nonlinear Os-</u>

cillations. GITTL, Moscow.

- Meirovitch, L. (1970). <u>Methods of Analytical Dynamics</u>. McGraw-Hill, New York.
- Melde, F. (1859). Yber Erregung stewphender Wellen eines fadenfomigen Korpers. Ann. Phys. Chemi. 109, 193-215.
- Min, G. B. and J. G. Eisley (1972). Nonlinear vibration of buckled beams. J. Eng. Ind. 94, 637-646.

Minorsky, N. (1947). <u>Non-Linear Mechanics</u>. J. W. Edwards, Ann Arbor.
 Minorsky, N. (1962). <u>Nonlinear Oscillations</u>. Van Nostrand, Princeton.
 Mook, D. T., L. R. Marshall and A. H. Nayfeh (1974). Subharmonic and superharmonic resonances in the pitch and roll modes of ship motions. J. Hydro. 8, 32-40.

- Mook, D. T., S. Sridhar and A. H. Nayfeh (1978). Nonlinear resonances in the forced responses of plates: Part II. <u>J</u>. <u>Sound Vib</u>. in[®] press.
- Morino, L. (1969). A perturbation method for treating nonlinear panel flutter problems. AIAA J. 7, 405-411.

Morris, N. F. (1965). The dynamic stability of beam columns with a fixed distance between supports. J. <u>Franklin Inst</u>. 280, 163-173.

- Murthy, D. N. S. and A. N. Sherbourne (1972). Free flexural vibrations of clamped plates. <u>J. Appl. Mech</u>. 39, 298-300.
- McLachlan, N. W. (1947). <u>Theory and Application of Mathieu Functions</u> Oxford University, New York.
- McLachlan, N. W. (1950). Ordinary Nonlinear Differential Equations in Engineering and Physical Sciences. Claredon Press, Oxford.

Nayfeh, A. H. (1973a). Nonlinear transverse vibrations of beams with properties that vary along the length. <u>J Acoust. Soc. Am.</u> 53, 766-770.

Nayfeh, A. H. (1973b). Perturbation Methods. Wiley, New York.

- Nayfeh, A. H. and D. T. Mook (1977). Parametric excitations of linear systems having many degrees of freedom. J. Acoust. Soc. Am. 62, 375-381.
- Nayfeh, A. H. and D. T. Mook (1979). <u>Nonlinear Oscillations</u>. Wiley, New York.
- Nayfeh, A. H., D. T. Mook and D. W. Lobitz (1974). A numerical-perturbation method for the nonlinear analysis of structural vibrations. <u>AIAA</u> J. 12, 1222-1228.

Nayfeh, A. H., D. T. Mook and L. R. Marshall (1973). Nonlinear coupling of pitch and roll modes in ship motions. J. Hydro. 7, 145-152.

- Nayfeh, A. H, D. T. Mook and L. R. Marshall (1974). Perturbation-energy approach for the development of the nonlinear equations of ship motion. J. <u>Hydro</u>. 8, 130-136.
- Nayfeh, A. H., D. T. Mook and S. Sridhar (1974). Nonlinear analysis of the forced response of structural elements. <u>J. Acoust. Soc. Am</u>. 55, 281-291.
- Nowinski, J. L. (1962). Non-linear transverse vibrations of circular elastic plates built-in at the boundary. <u>Proc. 4th U.S. Nat. Cong</u>. <u>Appl. Mech.</u> 325-334.
- Raju, I. S., G. Venkateswara Rao and K. Kanaka Raju (1976). Effect of longitudinal or inplane deformation and inertia on the large

amplitude vibrations of a slender beams and thin plates. <u>J</u>. <u>Sound</u> Vib. 49.

- Raman, C. V. (1912). Experimental investigations on the maintenance of vibrations. <u>Proc. Indian Assoc. Cultivation Sci. Bull</u> 6, 405.
- Ray, J. D. and C. W. Bert (1969). Nonlinear vibrations of a beam with pinned ends. J. Eng. Ind. 9a, 997-1104.
- Rehfield, L. W. (1973). Nonlinear free vibrations of elastic structures. <u>Int. J. Solids Struct</u>. 9, 581-590.
- Sandman, B. E. and C. L. Huang (1971). Finite amplitude oscillations of a thin elastic annulus. In <u>Develop</u>. <u>Mech</u>. 6, 921-934.
- Sathyamoorthy, M. and K. A. V. Pandalai (1970). Nonlinear flexural vibrations of orthotropic rectangular plates. <u>J. Aero. Soc. India</u> 4, 264-266.
- Smith, P. W., C. T. Malme and C. M. Gogos (1961). Nonlinear response of a simple clamped panel. J. Acoust. Soc. Am. 33, 1476-1480.
- Smith, L. and L. Morino (1976). Stability analysis of nonlinear differential autonomous systems with applications to flutter. <u>AIAA J</u>. 14, 333-341.
- Sridhar, S., A. H. Nayfeh and D. T. Mook, (1975). Nonlinear resonances in a class of multi-degree-of-freedom systems. <u>J. Acoust. Soc. Am</u>. 58, 113-123.
- Sridhar, S., D. T. Mook and A. H. Nayfeh (1975). Non-linear resonances in the forced responses of plates. Part I: Symmetric responses of circular plates. <u>J. Sound Vib.</u> 41, 359-373.

- Sridhar, S., D. T. Mook and A. H. Nayfeh (1978). Nonlinear resonances in the forced response of plates, Part II: Asymmetric responses of circular plates. J. Sound Vib. in press.
- Srinivasan, A. V. (1965). Large amplitude-free oscillations of beams and plates. AIAA J. 3, 1951-1953.
- Srinivasan, A. V. (1966a). Nonlinear vibrations of beams and plates. Int. Non-linear Mech. 1, 179-191.
- Srinivasan, A. V. (1967). Dynamic stability of beam columns. AIAA J. 5, 1685-1686.
- Stephenson, A. (1906). On a class of forced oscillations. Q. J. Math. 37, 353-360.
- Stephenson, A. (1908). On a new type of dynamic stability. <u>Mem. Proc</u>. <u>Manchr. Lit. Phil. Soc</u>. 52, 1907-1908.

Stoker, J. J. (1950). Nonlinear Vibrations. Wiley, New York.

- Strutt, J. W. (1887). On the maintenance of vibrations by forces of double frequency and on the propagation of waves through a medium endowed with a periodic structure. <u>Phil</u>. <u>Mag</u>. 24, 145-159.
- Sugiyama, Y., N. Fugiwara and T. Sekiya (1970). Studies on nonconservative problems of instability of columns by means of an analogue computer. In <u>Proc. 18th Japan Cong</u>. <u>Appl. Mech</u>. 1968, 113-126.
- Tseng, W. Y. and J. Dugundji (1970). Nonlinear vibrations of a beam under harmonic excitation. <u>J. Appl. Mech.</u> 37, 292-297.
- Tseng, W. Y. and J. Dugundji (1971). Nonlinear vibrations of a buckled beam under harmonic excitation. J. Appl. Mech. 38, 467-476.

Tso, W. L and K. G. Asmis (1974). Multiple parametric resonance in a nonlinear two degree of freedom system. <u>Int. J. Non-Linear Mach</u>. 9, 269-277.

Vendhan, C. P. (1975a). Model equations for the nonlinear flexural vibrations of plates. <u>AIAA J</u>. 13, 1092-1094

- Vendhan, C. P. (1975b). A study of Berger equations applied to nonlinear vibrations of elastic plates. <u>Int. J. Mech. Sci.</u> 17, 461-468.
- Vendhan, C. P. and Y. C. Das (1975). Application of Rayleigh-Ritz and Galerkin methods to non-linear vibration of plates. <u>J. Sound Vib</u>. 39, 147-157.
- Vendhan, C. P. and B. L. Dhoopar (1973). Nonlinear vibration of orthotropic triangular plates. <u>AIAA J</u>. 11, 704-709.
- Ventres, C. S. and E. H. Dowell (1970). Comparison of theory and experiment for nonlinear flutter of loaded plates. <u>AIAA J</u>. 8, 2022-2030.

Verma, M. K. and K. A. V. Murthy (1974). Non-linear vibrations of nonuniform beams with concentrated masses. <u>J. Sound Vib</u>. 33, 1-12.
Wagner, H. (1965). Large amplitude free vibrations of a beam. <u>J Appl</u>. Mech. 32, 887-892.

Woinowsky-Krieger, S. (1950). The effect of an axial force of the vibration of hinged bars, J. Appl. Mech. 17, 35-36.

Woodall, S. R. (1966). On the large amplitude oscillations of a thin elastic beam. Int. J. Non-linear Mech. 1, 127-238.

Yamaki, N. (1961). Influence of large amplitudes on flexural vibrations of elastic plates. <u>ZAMM</u> 41, 501-510.

Yamamota, T. and A. Saito (1970). On the vibrations of summed and differential types under parametric excitation. <u>Nagoya Memiors</u> <u>Faculty Eng.</u> 22, 54-123.

APPENDIX A

SIMILARITY TRANSFORMATIONS

A.1 General

Given a nonsingular matrix Z, then

 $[J] = [Z^{-1}][B][Z]$

(A1)

is a similarity transformation where the matrices [J] and [B] have the same eigenvalues. The choice of [Z] can determine the complexity of the form of [J], the simplist being the diagonal form when all the eigenvalues as distinct. If the eigenvalues are not all distinct [J] may still be diagonal, if the eigenvectors corresponding to the repeated roots are distinct. When the eigenvectors corresponding to repeated roots are identical, then the matrix can at best take on the Jordan form, where the eigenvalues appear as diagonal terms and a 1 (one) appears as an off-diagonal term adjacent to the repeated roots; see equation (9.11b). The number of off-diagonal terms depends on the number of repeated frequencies or more precisely the number of times a given root is repeated, and the corresponding number of independent eigenvectors. The latter depends on the rank of the matrix which is not the major topic of discussion of this appendix. However, a brief aside is in order. The reader must accept what follows in this example with faith or confirm if with rigor in a mathematics text such as Kreyszig (1972) (pp. 256-279).

Given an $n \times n$ matrix with one repeated frequency, then, if the rank of the matrix is r = n - 1, there is only one linearly independent

eigenvector associated with the repeated frequency. If the rank is r = n - 2 then there are two linearly independent eignevectors associated with the repeated frequency. The last case will allow generation of a diagonal matrix while the first will produce the Jordan form with one off diagonal term.

The problem is one of determining the matrix [Z] which will put [B] in Jordan form. Mathematics texts indicate [Z] exists and give examples to prove it or direct the student to verify a sample problem with mechanical computations. However, nowhere, does there appear to be a method to determine the matrix [Z] in the first place. That is precisely the purpose of this Appendix.

Finding the eigenvalues and eigenvectors of [B] is a relatively trivial problem with the advent of sophisticated library Computer programs. Using that as a starting point we proceed with a Newton-Raphson iterative technique to generate the matrix [Z] from the set of eigenvectors.

A.2 Determining the Form of [J].

We start with the matrix [B] and use any standard library computer program which will return eigenvalues and eigenvectors. An inspection of eigenvalues will indicate immediately if any are repeated. If they are too numerous, a simple sort routine can be employed to order them from highest to lowest to aid in identifying repeated roots. Once the repeated roots are identified, the corresponding eigenvectors are compared. The best way is to normalize them by dividing the elements, usually returned as a column vector, by the value of the first element. The vectors which remain are either identical or they are not. If they are identical then the Jordan form will evolve. There need not be any discussion of rank, linear independence or whatever.

A.3 Newton-Raphson Iteration Technique

The Newton-Raphson technique is especially useful in solving nonlinear problems. It is "initial guess" sensitive to some degree but converges rapidly if the system is fairly well behaved near the solution. The basic problem is one where we are required to find the values of a vector $\{x\}$ which will make

$$\{f(x)\} = 0$$
 (A2)

For a 2 x 2 example we expand in a Taylor series and obtain

$$f_{1}(x_{1}) = f_{1}(x_{10}) + \frac{\partial f_{1}}{\partial x_{1}} (x_{10}) \Delta x_{1} + \frac{\partial f_{1}}{\partial x_{2}} (x_{10}) \Delta x_{2} + \dots \quad (A3a)$$

$$f_{2}(x_{2}) = f_{2}(x_{20}) + \frac{\partial f_{2}}{\partial x_{1}} (x_{20}) \Delta x_{1} + \frac{\partial f_{2}}{\partial x_{2}} (x_{20}) \Delta x_{2} + \dots \quad (A3b)$$

where the x_{no} are the initial guesses. The Δx_n are the increments which will cause all the $f_i(x_n)$ to be zero simultaneously, so we write

$$\frac{\partial f_1}{\partial x_1} \Delta x_1 + \frac{\partial f_1}{\partial x_2} \Delta x_2 = -f_1(x_{10})$$
(A4a)

$$\frac{\partial f_2}{\partial x_1} \Delta x_1 + \frac{\partial f_2}{\partial x_2} \Delta x_2 = -f_2(x_{20})$$
(A4b)

and we have a linear system which can be solved for the Δx_n to add to the x_{n0} and proceed to another iteration until we reach some predetermined limit of convergence in either the f_i or Δx_i .

A.4 The [Z] Matrix With One Repeated Frequency

The [Z] matrix is constructed as follows. The eigenvectors are sorted so that the repeated vectors are in the first two columns and are normalized so the top element in each column is one. We seek a column vector to replace the first column, the repeated eigenvector. The vector function, f, represents the first column of the matrix [J] below the first element; that is

$$\{J_1\} = \begin{bmatrix} 1 \\ f_1(x_1) \\ f_2(x_2) \\ . \\ . \\ . \\ f_{n-1}(x_{n1}) \end{bmatrix}$$
(A5a)

We wish the final result to be

$$\{J_1\} = \begin{bmatrix} 1\\ 1\\ 0\\ .\\ 0 \end{bmatrix}$$
 (A5b)
the functional form $f(x_n)$ were known, we would have no problem

If the functional form $f(x_n)$ were known, we would have no problem in evaluating the partial derivatives

 $\frac{\partial f_i}{\partial x_j}$

(A6)

Here we must construct them numerically. We begin by assuming the first column of the [Z] matrix to be

$$\{Z_1\}_0 = \begin{bmatrix} 1 \\ 0 \\ 0 \\ . \\ . \\ 0 \end{bmatrix}$$

Then we compute

 $[J] = [Z^{-1}][B][Z]$ (A8)

The first column of [J] becomes

		Γ ₁]
	•	$f_1(x_0)$
$\{J_1\}_0$	=	$f_2(x_0)$
		•
		•
		$f_{n-1}(x_0)$

Then we set

$$\{Z_1\}_1 = \begin{bmatrix} 1 \\ 1 \\ 0 \\ . \\ . \\ 0 \end{bmatrix}$$

(A10)

(A9)

(A7)

and compute (A8) again so that

We may now compute the derivatives numerically as

$$\frac{\partial f_1}{\partial x_1} = \frac{f_1(x_1) - f_1(x_0)}{\Delta x_1} = f_1(x_1) - f_1(x_0) = f_{11} \quad (A12a)$$

$$\frac{\partial f_2}{\partial x_1} = \frac{f_2(x_1) - f_2(x_0)}{\Delta x_1} = f_2(x_1) - f_2(x_0) = f_{21} \quad (A12b)$$

Since $\Delta x_1 = 1$. We repeat for

 $\{Z_1\}_2 = \begin{bmatrix} 1\\0\\1\\0\\.\\.\\0 \end{bmatrix}$ (A13)

to obtain

$$\{J_1\}_2 = \begin{bmatrix} 1 \\ f_1(x_2) \\ f_2(x_2) \\ \vdots \\ \vdots \\ f_{n-1}(x_2) \end{bmatrix}$$
(A14)

so that the second column of derivatives may be computed according to

$$\frac{\partial f_1}{\partial x_2} = \frac{f_1(x_2) - f_1(x_0)}{\Delta x_2} = f_1(x_2) - f_1(x_0) = f_{12}$$

r = n - 1 $\begin{bmatrix} f_{11} & f_{12} & \cdots & f_{1_n} \\ f_{21} & \cdots & \cdots & \vdots \\ f_{31} & \cdots & \cdots & \vdots \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ f_{n^1} & \cdots & \cdots & f_{nn} \end{bmatrix} \begin{bmatrix} \Delta x_1 \\ \Delta x_2 \\ \vdots \\ \vdots \\ \Delta x_n \end{bmatrix} = \begin{bmatrix} 1 - f_1(x_1) \\ - f_2(x_2) \\ - f_3(x_3) \\ \vdots \\ \vdots \\ - f_n(x_n) \end{bmatrix}$ (A15)

where the right side was computed from

Г 7

$$\{\mathbf{Z}_{1}\}_{\mathsf{R}} = \begin{bmatrix} 1\\ 1\\ 1\\ 1\\ 1\\ 1\\ 1 \end{bmatrix}$$
(A16)

and the factor of one applied to the first element to satisfy (A5b). Equations (A15) may now be solved for the Δx_i and added to one to form a new guess for the x_i . Now the product of (A1) is computed to determine if the Jordan form has been achieved. Convergence is achieved by specifying the accuracy with which (A17) is satisfied:

$$[Z^{-1}][B][Z] - [J] \stackrel{\sim}{\sim} 0$$
 (A17)

In this problem one step was required to produce an accuracy of 10^{-10} in the element using double precision.

154

and so on, until we have a system of the form of (A4) and we write for

The author was born in Steelton, Pennsylvania, on 16 October 1940. He graduated from Central Darphin High School in Harrisburg, PA in 1958. He attended the U. S. Military Academy at West Point, NY and graduated in 1963 as a Distinguished Cadet (top 5% of the class) and was commissioned as a Second Lieutenant in the U. S. Army. Between 1963 and 1965 he attended the Engineer Officers Basic Course at Fort Belvoir, VA, Airborne Training at Fort Benning, GA, and served as Executive Officer and Company Commander of B. Co. 13th Engineer Battalion, 7th Infantry Division, Korea. In 1967 he received a Master of Science degree in Astrodynamics from the University of California at Los Angeles (UCLA). He then spent a year as an Engineer Construction Battalion Advisor in Vietnam and followed that with attendance at the Amphibious Warfare School in Quantico, VA. Between 1969 and 1972 he spent a year as an Instructor for Engineering Mechanics and two years as an Assistant Professor for Space Mechanics in the Mechanics Department at the U.S. Military Academy. This was followed by a second Vietnam tour where he served as the Plans Officer for the U.S. Army Engineer Group. He is currently the Group Director for solid mechanics courses for the Department of Mechanics at the U. S. Military Academy.

Eldsyck

VITA

PARAMETRICALLY EXCITED NONLINEAR MULTI-DEGREE-OF-FREEDOM SYSTEMS

by

Edward G. Tezak

(ABSTRACT)

An analysis of parametrically excited nonlinear multi-degree-offreedom systems is presented. The nonlinearity considered is cubic and small so that the system of equations is weakly nonlinear. Modal damping is included and the parametric excitation is harmonic. The systems examined include those with distinct natural frequencies as well as those with a single repeated frequency. The significant role played by the existence of an internal resonance is explored in depth. The derivative-expansion version of the method of multiple scales, a perturbation technique, is used to develop solvability conditions for the various combinations of internal and parametric resonances considered.

Regions where trivial and nontrivial solutions exist are defined and the stability of the solutions within each region is discussed. Nontrivial, unstable solutions have been shown to exist in regions where nontrivial stable solutions are known. Numerical solutions do not hint at the existence of these solutions.

The role of internal resonance in parametrically excited systems is explored. Strong modal interaction is demonstrated as a consequence of the presence of the cubic nonlinearity and the internal resonance. Because of this modal coupling, modes other than the one excited can dominate the response. A multiplicity of jumps is shown to exist.

Parametric excitation in the nonlinear flutter problem has been examined in detail for the first time. The effect of the parametric excitation can raise the flutter speed if the amplitude is small and lower it if the amplitude is large.

Limit-cycle behavior in the flutter problem is developed in a unique analytic way. The condition which predicts the onset of flutter is developed by using a linear analysis. The solution grows without bound. Interestingly, the same condition in the nonlinear analysis predicts the existence of a real nontrivial solution with a finite amplitude, the so-called limit-cycle.